

Assessment of physical activity in adolescents: protocols, compliance and participants' perceptions

Joseph John Scott

B. Teaching (secondary) / B. Health and Physical Education Priority Research Centre for Physical Activity and Nutrition School of Education Faculty of Education and Arts University of Newcastle

Thesis submitted in fulfilment of the requirements for the award of the degree of:

Doctor of Philosophy

University of Newcastle

August, 2017

Statement of Originality

This thesis contains no material which has been accepted for the award of any other degree of diploma in any university or other tertiary institution. To the authors' knowledge it contains no material previously published or written by any other person, except where due reference has been made within the text. I give consent to the final version of my thesis being made available worldwide when deposited in the University's Digital Repository, subject to the provisions of the Copyright Act 1968.

Signed:

Name: Joseph J. Scott

Date: August, 2017

Statement of Authorship

I hereby certify that this thesis is submitted in the form of a series of published papers of which I am a joint author. I have included as part of the thesis a written statement from each co-author; and endorsed by the Faculty Assistant Dean (Research Training), attesting to my contribution to the joint publications. I warrant that I have obtained, where necessary, permission from the copyright owners to use any third-party copyright material reproduced in the thesis (e.g. questionnaires, artwork, unpublished letters), or to use any of my own published work (e.g. journal articles) in which the copyright is held by another party (e.g. publisher, co-author)

Signed:

Name: Joseph J. Scott

Date: August, 2017

Supervisors

Supervisors

Principal supervisor

Professor David R. Lubans Priority Research Centre for Physical Activity &Nutrition School of Education Faculty of Education & Arts University of Newcastle, Australia

Co-supervisors

Professor Philip J. Morgan Priority Research Centre for Physical Activity & Nutrition School of Education Faculty of Education & Arts University of Newcastle, Australia

Professor Ronald C. Plotnikoff Priority Research Centre for Physical Activity & Nutrition School of Education Faculty of Education & Arts University of Newcastle, Australia

Acknowledgments

I would personally like to acknowledge the incredibly important people who have been involved in, and have significantly contributed to the completion of this thesis.

First, I would like to thank my principal supervisor Professor David Lubans. You have been a true mentor to me. Your success in your career and field is attributed to your deep knowledge, expertise and enthusiasm to promote health and physical activity in education. I have learnt so much from you, and continue to do so. Thank you for always getting up early or staying at work late so that we could meet outside work hours in your own time. You have always made my candidature a priority, no matter what was happening in your busy schedule. I can't thank you enough for what you have done for me. Thank you for your support both with my studies and personal life throughout my PhD journey. I am forever grateful for your support and guidance and I look forward to working with you in the future as colleagues.

Second, my co-supervisors Professor Ron Plotnikoff and Professor Phillip Morgan who have always made time to provide me with invaluable advice. Your wisdom and esteemed knowledge have provided me with many skills that I will use long into my career as an academic. Your dedication to promoting health is truly reflected in your incredibly successful careers. I was so lucky to have you both as supervisors and am so grateful for the guidance you both provided.

Third, to the University of Newcastle: The School of Education: Faculty of Education and Arts, for its support in the form of research training scheme (RTS) funding. In particular, I would like to thank Professor John Fischetti and Professor Max Smith. Both of you were key in my motivation to continue to pursue the completion of my thesis. With your assistance, I was able to devise strategies to overcome the barriers that I faced as a part-time student. You both went over and above your roles to support me as a part-time PhD candidate. Your guidance and advice is one of the reasons the completion of this thesis has been possible.

Fourth, I would like to thank the University of Newcastle, Priority Research Centre: Physical Activity and Nutrition (PRCPAN) for its support in the form of seed funding. More importantly,

Acknowledgments

all of the seasoned academics and fellow Masters and PhD students that provided me guidance and support.

Fifth, I would also like to thank the involved New South Wales schools, administrators, teachers, parents, students that made this research possible.

Sixth, to my friends, both inside and outside work. The laughs and good times are the only reasons that I was able to remain so positive about the completion of my candidature.

Seventh, I would like to thank my family for the love and support they have provided me in both the good, and more difficult times. To my parents, Stephen and Brigitte, your support, resilience and courage each day has given me the motivation to continue through this PhD journey.

Finally, to my wife Al, thank you for your infinite and enduring support throughout my candidature. Over the years, you have continually provided me with love, wisdom and optimism throughout many of life's challenges. Your achievements in your own PhD candidature and career is truly inspirational. Without you, I would never have possessed the motivation and determination to complete this thesis, nor be where I am today, and for that I am truly grateful.

Publications arising from this thesis

This thesis presents a series of five papers. I am the lead author in four papers and I am the coauthor of one paper. At the time of submission, five of these papers were published. I have presented results arising from this thesis at the University of Newcastle (Faculty of Education and the PRCPAN conferences) and also at two international conferences (detailed below).

Manuscripts published in peer-reviewed journals:

- Lubans, D. R., Plotnikoff, R. C., Miller, A., Scott, J. J., Thompson, D., & Tudor-Locke, C. (2014). Using pedometers for measuring and increasing physical activity in children and adolescents: the next step. *American Journal of Lifestyle Medicine*. doi: 10.1177/1559827614537774
- 2. Scott, J.J., Morgan, P. J., Plotnikoff, R. C., & Lubans, D. R., (2015). Reliability and validity of a single-item physical activity measure for adolescents. *Journal of Paediatric Child Health*, 2015. **51**(8): p. 787-93.
- Scott, J. J., Morgan, P. J., Plotnikoff, R. C., Trost, S. G., & Lubans, D. R. (2013). Adolescent pedometer protocols: examining reactivity, tampering and participants' perceptions. *Journal of Sports Sciences*, 32(2), 183-190
- Scott, J. J., Rowlands, A. V., Morgan, P. J., Plotnikoff, R. C., & Lubans, D. R (2017). Comparability and feasibility of wrist- and hip-worn accelerometers in free-living adolescents, *Journal of Science and Medicine in Sport*, doi: 10.1016/j.jsams.2017.04.017
- Scott, J. J., Hansen, V. Morgan, P. J., Plotnikoff, R. C., & Lubans, D. R (2017). Young Peoples' Perceptions of the Objective Physical Activity Monitoring Process: A Qualitative Exploration, *Health Education Journal*, doi: 10.1177/0017896917734576

Peer-reviewed conference presentations:

- Scott, J. J., Morgan, P. J., Plotnikoff, R. C., Trost, S. G., & Lubans, D. R. (2012). Pedometer protocols for measuring physical activity: An examination of reactivity, tampering and perceptions among adolescents. *International Congress of Health and Physical Activity*. Sydney, 31 Oct - 03 Nov. 2012. Oral paper presentation.
- Scott, J. J., Hansen, V. Morgan, P. J., Plotnikoff, R. C., & Lubans, D. R (2015). Adolescents' Perceptions of the Objective Physical Activity Monitoring Process: A Qualitative Exploration, *International Congress of Diet and Physical Activity Measurement*. Brisbane, 1 - 3 Sept, 2015. Symposium presentation.

Statement of contribution of others

I attest that Research Higher Degree candidate Mr Joseph J Scott contributed to the publication

for which I am a co-author. For all publications, where applicable, Joseph has:

- Contributed to the development of research questions
- Contributed to research design and methodology
- Contributed to the development and modification of data collection tools
- Contributed to study design
- Managed data collection procedures
- Cleaned the data
- Led all data analysis
- Led the writing of each manuscript

Full name of co- author	Chapters co-authored	Signature of co-author	Date
David R. Lubans	2, 3, 4, 5, 6		24 June 2017
Philip J. Morgan	2, 3, 4, 5, 6		24 June 2017
Ronald C. Plotnikoff	2, 3, 4, 5, 6		24 June 2017

Full name of co- author	Chapters co-authored	Signature of co-author	Date
Andrew Miller	2		13 Feb 2017
Debbe Thompson	2		11 Feb 2017
Catrine Tudor- Locke	2		13 Feb 2017
Stewart Trost	4		12 Feb 2017
Vibeke Hansen	5		12 Feb 2017
Dylan Cliff	6		13 Feb 2017
Alex V Rowlands	6		11 Feb 2017

Signature of PhD candidate:

Joseph Scott

Date: 14 August 2017

Signature of Assistant Dean Research Training (ADRT):

Rosalind Smith

Date: 14 August 2017

Table of contents

Sutement of offshulty.	i
Statement of authorship	ii
Supervisorsi	iii
Acknowledgements	iv
Publications arising from this thesis	vi
Statement of contribution of author	<i>ii</i>
Table of contents	ix
List of tables	v
List of figuresx	vi
Thesis Abstractxv	'ii
Contribution statementxxi	iii
Study designxxi	iii
Data collection managementxxi	iii
Reporting and presentation of resultsxx	iv
Operational definitionsxx	v
Thesis structurexx	vi
Thesis structure	vi .1
Thesis structure	vi .1
Thesis structure	vi .1 .1
Thesis structure	vi .1 .4
Thesis structure	vi .1 .4 .4
Thesis structure	vi .1 .4 .4 .6
Thesis structure	vi .1 .4 .6 .6
 Thesis structure	vi .1 .4 .6 .6 .7
 Thesis structure	vi .1 .4 .6 .6 .7 .8 .8
Thesis structure .xx CHAPTER 1: INTRODUCTION 1.1. Background 1.2. Physical activity: benefits and consequences 1.2.1. Current physical activity guidelines for adolescents 1.2.1. Current physical activity guidelines for adolescents 1.2.2. Physical activity prevalence 1.2.3. National and global costs associated with physical inactivity 1.2.4. Patterns of physical activity among adolescents 1.3 Measurement of physical activity 1.3.1. Assessing physical activity 1.3.2. Subjective measures of physical activity 1	vi .1 .4 .6 .6 .7 .8 .8

1.3.3. Self-report measures
1.3.4. Strengths and limitations of self-report measures10
1.3.5. Objective measures of physical activity13
1.3.6. Direct observation
1.3.6.1. Strengths and limitations of direct observation14
1.3.7. Pedometers
1.3.7.1. Strengths and limitations of pedometers15
1.3.8. Accelerometers
1.3.8.1. Strengths and limitations of accelerometers
1.4. Summary of physical activity assessment methods
1.5. Specific objective physical activity measurement issues in adolescent populations22
1.5.1. Introduction
1.5.2. Objective measurement: Technical issues
1.5.3. Selection of device and comparability
1.5.4. Use of epochs and cut-points
1.5.5. Treatment of missing data and data imputation24
1.5.6. Behavioural issues27
1.5.7. Reasons for non-compliance with protocols27
1.5.8. Device placement and compliance
1.5.9. Reactivity
1.5.10. Tampering
1.6. Chapter 1 summary
1.7. Research aim and study objectives
1.7.1. Overall aim
1.7.2. Study objectives

CHAPTER 2: USING PEDOMETERS FOR MEASURING AND INCREASING
PHYSICAL ACTIVITY IN CHILDREN AND ADOLESCENTS: THE NEXT STEP33

2.1. Abstract	
2.2. Introduction	34
2.3. Pedometer monitoring protocols	
2.4. A step-defined sedentary lifestyle index for children and adolescents	
2.5. Using pedometers to increase physical activity in young people	
2.6. Using pedometers in schools to promote physical activity in young people	44
2.7. Conclusions and future research	46
CHAPTER 3: RELIABILITY AND VALIDITY OF A SINGLE-ITEM PHYSICAI	L
ACTIVITY MEASURE FOR ADOLESCENTS	49
3.1. Abstract	50
3.2. Background	50
3.3. Methods	52
3.3.1. Study design	52
3.3.2. Participants	52
3.3.3. Procedures	53
3.3.4. Research questions	53
3.3.5. Accelerometers	54
3.3.6. Single-item physical activity measure	54
3.3.7. Oxford physical activity questionnaire	54
3.3.8. Treatment of missing data	54
3.3.9. Analysis	55
3.4. Results	56
3.4.1. Reliability	56
3.4.2. Validity	57
3.5 Discussion	57

3.6 Conclusions	0
CHAPTER 4: ADOLESCENT PEDOMETER PROTOCOLS: EXAMINING	
REACTIVITY, TAMPERING AND PARTICIPANTS' PERCEPTIONS' 64	4
4.1. Abstract6	5
4.2. Introduction	5
4.3. Methods	7
4.3.1. Study design	7
4.3.2. Measures	7
4.3.3. Pedometers	7
4.3.4. Accelerometers	7
4.3.5. Pedometry behaviour questionnaire	8
4.3.6. Procedures	8
4.3.7. Treatment of data	C
4.4. Results7	1
4.4.1. Reactivity	1
4.4.2. Tampering	1
4.4.3. Self-reported behaviours and perceptions	1
4.5. Discussion	2
4.6. Conclusions and recommendations	5
CHAPTER 5: YOUNG PEOPLES' PERCEPTIONS OF THE OBJECTIVE PHYSICAL	
ACTIVITY MONITORING PROCESS: A QUALITATIVE EXPLORATION	9
5.1. Abstract	0
5.2. Background	1
5.3. Methods	3
5.3.1. Study design and procedures	3
5.3.2. Measures	3
5.3.3. Pedometry/accelerometry	3

5.3.4. Focus groups	
5.3.5. Analysis	85
5.4. Results	86
5.4.1. Attitudes relating to pedometers and their usefulness	86
5.4.2. General perception of wearing pedometers	87
5.4.3. Perceived need to impress others	87
5.4.4. Perceived dishonesty amongst others	
5.4.5. Perceived usefulness of assessing physical activity with pedometers	
5.4.6. Behaviours relating to pedometer use	90
5.4.7. Reasons for removing pedometers	90
5.4.8. Tampering with pedometers	90
5.4.9. Changes in physical activity levels while wearing pedometers	91
5.5. Discussion	92
5.6. Conclusion	95
CHAPTER 6: COMPARABILITY AND FEASIBILITY OF WRIST- AND H	IP-WORN
CHAPTER 6: COMPARABILITY AND FEASIBILITY OF WRIST- AND H ACCELEROMETERS IN FREE-LIVING ADOLESCENTS	IP-WORN
CHAPTER 6: COMPARABILITY AND FEASIBILITY OF WRIST- AND H ACCELEROMETERS IN FREE-LIVING ADOLESCENTS 6.1. Abstract	IP-WORN 98 99
CHAPTER 6: COMPARABILITY AND FEASIBILITY OF WRIST- AND HE ACCELEROMETERS IN FREE-LIVING ADOLESCENTS	IP-WORN
CHAPTER 6: COMPARABILITY AND FEASIBILITY OF WRIST- AND HE ACCELEROMETERS IN FREE-LIVING ADOLESCENTS	IP-WORN
CHAPTER 6: COMPARABILITY AND FEASIBILITY OF WRIST- AND HE ACCELEROMETERS IN FREE-LIVING ADOLESCENTS	IP-WORN
CHAPTER 6: COMPARABILITY AND FEASIBILITY OF WRIST- AND HE ACCELEROMETERS IN FREE-LIVING ADOLESCENTS	IP-WORN
CHAPTER 6: COMPARABILITY AND FEASIBILITY OF WRIST- AND HE ACCELEROMETERS IN FREE-LIVING ADOLESCENTS	IP-WORN
CHAPTER 6: COMPARABILITY AND FEASIBILITY OF WRIST- AND HE ACCELEROMETERS IN FREE-LIVING ADOLESCENTS	IP-WORN
CHAPTER 6: COMPARABILITY AND FEASIBILITY OF WRIST- AND HE ACCELEROMETERS IN FREE-LIVING ADOLESCENTS	IP-WORN
CHAPTER 6: COMPARABILITY AND FEASIBILITY OF WRIST- AND HE ACCELEROMETERS IN FREE-LIVING ADOLESCENTS	IP-WORN
CHAPTER 6: COMPARABILITY AND FEASIBILITY OF WRIST- AND H ACCELEROMETERS IN FREE-LIVING ADOLESCENTS	IP-WORN

7.3.2. Strengths and limitations
7.3.3. Implications for practice and recommendations for future research122
7.4. Pedometer monitoring protocols in adolescents: compliance, reactivity and tampering
7.4.1. Overview of findings
7.4.2. Strengths and limitations
7.4.3. Implications for practice and recommendations for future research126
7.5 A qualitative investigation of adolescents' perceptions of the pedometer monitoring
process
7.5.1. Overview of findings
7.5.2. Strengths and limitations
7.5.3. Implications for practice and recommendations for future research130
7.6. Comparability and feasibility of wrist- and hip-worn accelerometers in adolescents132
7.6.1. Overview of findings
7.6.2. Strengths and limitations134
7.6.3. Implications for practice and recommendations for future research135
7.7. Concluding remarks137
BIBLIOGRAPHY
APPENDICIES

List of tables

Table 1.1. International physical activity guidelines for young people (ages 5 – 17years)
Table 1.2. Summary of key attributes for current methods to measure physical activity in
children and adolescents; adapted from Trost and colleagues21
Table 3.1. Characteristics of participants 61
Table 4.1. Characteristics of sample across three groups
Table 4.2. Changes in daily step counts and reactivity by pedometer group
Table 4.3. Relationship between mean steps/day and accelerometer counts per minute
Table 5.1. Characteristics of larger and focus group sample 96
Table 5.2. Examples of focus group questions
Table 6.1. Relationship between the wrist- and hip-worn accelerometer output
Table 6.2. Paired sample t-tests to explore individual differences in MVPA
Table 6.3. Comparison of non-valid (missing) days 115

List of Figures

Figure 1.1. Schematic diagram of chapter one	.3
Figure 1.2. Categorisation of physical activity measurement	.9
Figure 3.1. Bland-Altman Plot for the OPAQ with 95% confidence limit comparing mean self-	-
reported MVPA and inter-trial difference (trial 1 and trail 2)	52
Figure 3.2. Bland-Altman Plot for the single-item measure with 95% confidence limits	
comparing mean self –reported MVPA and inter-trial difference (trial 1 and trail 2)	53
Figure 6.1. Scatter plot showing linear fit of acceleration measured by hip-worn GT3X+ (VM))
against the wrist-worn GENEActiv ENMO (mg)11	6
Figure 6.2. Scatter plot showing linear fit of MVPA (minutes) measured by hip-worn GT3X+	
against the wrist-worn GENEActiv11	7

Thesis abstract

Background

The identification of appropriate and accurate tools for measuring physical activity is an international research priority. There is now a plethora of existing physical activity questionnaires that have been designed to assess activity patterns; few, however, are both valid and reliable. Existing questionnaires vary extensively in length and complexity with some requiring high level cognitive recall, rendering them unsuitable for some population groups. In particular, adolescents represent a problematic and complex sub-population in which to measure physical activity, due to incomplete or inaccurate reporting. There is thus a need for simplified and validated physical activity questionnaires designed specifically for use with adolescents.

Objective monitoring devices such as pedometers and accelerometers have acceptable feasibility, validity and reliability. In addition, as both motion sensors share similar technical and adherence issues. There is, however, a lack consensus regarding the optimal monitoring protocols. Poor adherence to objective monitoring protocols is a common reason why assessing activity patterns of adolescents remains complex. This may explain why adolescents are one of the least studied sub-population groups in physical activity research. As physical activity levels decline through adolescence, it is important that we improve our understanding of physical activity assessment in this target population.

Pedometers are now commonly used to measure physical activity in different age groups (that is, children, adolescents, adults and older adults); however, little is known about the influence of different pedometer protocols on the accuracy of physical activity measurement. Previous studies in children and adults have investigated reactivity and tampering with pedometer monitoring, however the research in the adolescent domain remains relatively sparse. Furthermore, no studies have explored adolescents' perceptions of the pedometer monitoring process to gain insight into how and why they do not comply with monitoring protocols.

xvii.

To date, accelerometers have predominantly been worn on the hip. However, in an attempt to increase participant compliance, researchers have more recently trialled the use of wrist-worn accelerometers in various populations. Currently, no studies have investigated the comparability and feasibility of hip- and wrist-worn accelerometers in adolescents. Given the importance of accurate physical activity measurement, there is a clear need for standardised physical activity monitoring protocols.

Aims

This thesis-by-publication presents a series of studies that were conducted to address the gap in the evidence base surrounding physical activity assessment in adolescents. The overarching aim of this thesis was to investigate current physical activity measurement protocols to assess physical activity levels in this sub-population. This thesis presents a series of studies that investigated six key study objectives, which are described briefly below. As these studies provide important context for the overall aim, the thesis is presented in the following order: *Study objective 1: assess the test–retest reliability of a single-item physical activity questionnaire for adolescents.*

Study objective 2: determine the concurrent validity of a single-item physical activity measure for adolescents compared to accelerometry.

Considering the large number of existing physical activity measures available, the wide-ranging variability in their format; length and complexity and their limitations in accuracy, there is a need for more direct comparisons of questionnaires to improve our understanding of the most appropriate physical activity measures. The aim of this study was to investigate the test–retest reliability and validity of a single-item questionnaire in free-living adolescents (n = 123) by testing it against a more extensive existing physical activity questionnaire (The Oxford Physical Activity Questionnaire [OPAQ]) and accelerometer output. The single-item (intra-class correlation coefficient [ICC] = 0.75, 95% confidence interval [CI] = 0.64 - 0.83, p < 0.001) and the OPAQ (ICC = 0.79, 95% CI = 0.69 - 0.86, p < 0.001) were both found to have moderate-to-strong reliability. Correlations between self-reported and objectively-measured moderate-to-

vigorous physical activity (MVPA) were similar for the single-item measure (r = 0.44, 95% CI = 0.24 - 0.63, p < 0.001) and the OPAQ (r = 0.50, 95% CI = 0.30 - 0.65, p < 0.001). The single-item questionnaire compared well to both the OPAQ and accelerometer derived activity counts. Study findings suggest the single-item measure can provide a reliable and valid assessment of adolescent physical activity. The single-item measure is easy to administer and may have utility for screening purposes and for use in population surveys.

Study objective 3: explore the impact of different pedometer monitoring protocols on compliance, reactivity and tampering in a sample of adolescents.

The aim of this study was to investigate adolescents' potential reactivity and tampering while wearing pedometers. In this study, adolescents were randomised to one of three pedometer monitoring protocols: (i) daily sealed pedometer group, (ii) unsealed pedometer group or (iii) weekly sealed pedometer group. Participants wore pedometers (Yamax Digi-Walker CW700, Yamax Corporation, Kumamoto City, Japan) and accelerometers (ActiGraph GT3X+, Pensacola, USA) simultaneously for seven days and completed a pedometry behaviour questionnaire. Repeated measures analysis of variance was used to examine potential reactivity. Bivariate correlations between step counts and accelerometer output were calculated to explore potential tampering. The correlation between accelerometer output and pedometer steps/day was strongest among participants in the weekly sealed group ($r = 0.82, p \le 0.001$), compared with the unsealed (r = 0.63, $p \le 0.001$) and daily sealed (r = 0.16, p > 0.05) groups. The daily sealed $(p \le 0.001)$ and unsealed $(p \le 0.001)$ groups, but not the weekly sealed (p = 0.886) group, showed evidence of reactivity. Responses from the questionnaire indicated that almost half the participants reported shaking their pedometers to increase their step count, 40% reported that they did not like wearing pedometers, 81% found wearing a pedometer uncomfortable, and 69% reported that they found wearing a pedometer 'embarrassing at times'. Contrary to previous research, the study findings suggest that the protocol selected for pedometer monitoring impacts behaviour and compliance. A seven-day monitoring protocol using sealed pedometers capable of storing at least seven days of step count data in their internal memory is recommended to

limit reactivity and tampering in adolescents.

Study objective 4: explore adolescents' perceptions of wearing pedometers and investigate behaviours exhibited while wearing pedometers.

The aim of this study was to explore adolescents' perceptions of pedometer monitoring and investigate the physical activity behaviours exhibited whilst wearing the devices. Six focus groups (3 boys' groups and 3 girls' groups), each involving four participants, were completed. The focus group participants were selected from the larger group of students participating in the monitoring protocol study; that is, from the 123 participants who wore pedometers (Yamax CW700) and accelerometers (ActiGraph GT3X+) simultaneously for seven days. Students were then grouped based on their daily accelerometer-determined MVPA level: i) low active: < 30 minutes/day in MVPA, ii) medium active 30 - 60 minutes/day in MVPA and iii) high active \geq 60 minutes/day in MVPA over their monitoring period).

Participants were questioned on the behaviours exhibited while they were wearing activity monitors and their perceptions of the monitoring process. A large proportion of the participants (approximately 60%) reported purposely changing their levels of physical activity during the monitoring process, and 21 of the 24 focus group participants reported shaking their pedometers to increase their step counts. More participants in the medium and high active groups reported changing their activity patterns than in the low active groups. The study findings suggest that the reasons for non-adherence to pedometer protocol were not related to sex or physical activity level. The high amount of reported reactivity indicates that pedometers are still a useful tool to promote physical activity in adolescents, but are less useful, however, when attempting to obtain habitual activity patterns.

Study objective 5: test the comparability and feasibility of the wrist- and hip-worn accelerometers in the free-living adolescent population.

Study objective 6: compare wear-time, missing data and participant perceptions regarding wrist- and hip-worn accelerometers.

The aim of this study was to determine the comparability and feasibility of wrist- and hip-worn accelerometers among free-living adolescents. In addition, the study focused on participants' perceptions of wrist- and hip-worn accelerometers to assist in determining the superior site placement for accelerometers when assessing adolescents in free-living conditions. The sample included 89 adolescents (age = 13 - 14 years) from eight secondary schools in New South Wales (NSW), Australia. Participants wore wrist-worn GENEActiv and hip-worn ActiGraph (GT3X+) accelerometers simultaneously for seven days and completed an accelerometry behaviour questionnaire. Bivariate correlations between the wrist- and hip-worn output were used to determine concurrent validity. Paired samples t-test were used to compare minutes per day in MVPA. Group means and paired sample t-tests were used to analyse participants' perceptions of the wrist- and hip-worn monitoring protocols to determine feasibility.

Wrist-worn accelerometers compared favourably with hip-worn in average activity (r = 0.88, p < 0.001) and MVPA ($r = 0.84 \ p < 0.001$, mean difference = 3.54 mins/day, SD = 12.37). The wrist-worn accelerometer had 50% fewer non-valid days (75 days, 12%) than the hip-worn accelerometer (n = 152, 24.4%). Participants reported they preferred to wear the device on the wrist (p < 0.001), and that it was less uncomfortable (p = 0.023) and less embarrassing to wear on the wrist (p < 0.001). Furthermore, they reported they would be more willing to wear the device on the device on the wrist than on the hip (p < 0.001). Study findings revealed a strong linear relationship between wrist- and hip-worn accelerometer output among adolescents in free-living conditions. Compliance was significantly higher for wrist placement, with participants reporting that it was more comfortable and less embarrassing to wear and, importantly, they would be more willing to wear it again on the wrist than on the hip.

Summary

This thesis reviews the existing literature surrounding existing pedometer protocols among adolescents to investigate gaps in the research. The single item physical activity questionnaire designed specifically for adolescents compared well to an existing physical activity

xxi.

questionnaire. As such, this measure may utility for determining whether or not adolescents are meeting physical activity guidelines. In addition, this thesis presents findings that support that reactivity and tampering are inherent risks to validity when assessing adolescents' physical activity with pedometers. Strategies to limit reactivity and tampering should be considered by researchers during studies when attempting to accurately assess adolescents' physical activity patterns. In an attempt to increase compliance with monitoring protocols in the adolescent population, it is recommended that future research utilise wrist-worn accelerometers. The original contribution of this thesis is the collection of data using both quantitative and qualitative methods to improve our understanding of adolescents' compliance to physical activity monitoring protocols. The evidence collected in this thesis may assist in the development of pedometer and accelerometer measurement protocols designed for adolescents.

Contribution statement

- This thesis includes five peer-reviewed journal publications.
- I was the sole PhD student for this study and was involved in all aspects of the study design, implementation, data collection and interpretation, reporting writing and manuscript preparation through to submission. A summary of the contributions that I made to this study is provided below:

Study design

- With the assistance of my supervisors, I was involved in study conception and design. This involved a number of joint decisions that would allow us to test various existing physical activity measures, including self-report measures, pedometer and accelerometer protocols.
- With the assistance of my supervisors, I led the completion of the ethics applications for both university and other educational (secondary school) institutions.
- With the assistance of my principal supervisor, I completed seed funding grant funding application.

Data collection and management

- In collaboration with my supervisors, I led the development of the measurement protocol resources, consent letters, physical activity questionnaire and focus group questions.
- I was personally responsible for recruiting the study principals, schools, and students.
- I ran workshops for the teachers involved in study 1 at the three different schools to assist with data collection. I then coordinated data collection timeframes with principals, head teachers and involved teachers.
- I was responsible for the distribution and collection of physical activity questionnaires, pedometers and accelerometers. I conducted the focus groups with students and handled recording and arranged transcription of data.

- I was responsible for the self-report questionnaire data entry. I also pre-set and initialised pedometers and accelerometers prior to monitoring, and, on completion of monitoring, downloaded, entered and cleaned all of the data.
- The data used in Chapter 6 was collected research assistants as part of the existing Switch-off 4 Healthy Minds (S4HM) cluster randomised controlled trial [1]. I designed the accelerometer monitoring protocol, trained the involved research assistants prior to data collection and developed the accelerometer behaviour questionnaire. I was responsible for the accelerometer data entry, cleaning, interpretation and analysis of the data so that it could later be reported.
- With the assistance of my supervisors, I completed the statistical analysis on four of the five papers presented in this thesis.

Reporting and presentation of results

- With the assistance of my supervisors, I was responsible for reporting the findings of the project and drafting manuscripts for submission to peer-reviewed journals.
- Throughout my candidature, I presented two peer-reviewed paper presentations at international conferences.

Definitions

Operational definitions

Child: Individuals 5 - 12 years of age [2].

Youth: Individuals 12 - 24 years of age [3].

Adolescent: Individuals 10 - 19 years of age [4].

Adult: Individuals ≥ 18 years of age [5].

MET: Metabolic equivalent count is the ratio of a person's working metabolic rate relative to their resting metabolic rate. Metabolic equivalents are used to express activity intensity [6].

Physical activity: Any bodily movement produced by skeletal muscles that requires energy expenditure [5].

Moderate-to-vigorous physical activity (MVPA): Physical activity with a MET count of > three [7].

Pedometer: A small device, most commonly worn on the hip or the wrist, used to measure step counts [8].

Accelerometer: A small electromechanical device, commonly worn on the hip or the wrist,

used to measure acceleration forces, which can be converted into activity estimates [8].

Reactivity: Defined as a change in normal activity pattern when participants are aware that they are being monitored [9].

Tampering: The term used for shaking a pedometer to inflate/increase an individual's step counts [10].

Thesis structure

This thesis presents a series of five papers, four of which are published, and one is under peerreview. The thesis structure is presented below.

Chapter 1: Introduction

The first chapter presents an overview of physical activity benefits, recommendations and trends. It also reviews the literature in relation to physical activity measurement and highlights current issues regarding the assessment of physical activity in adolescents.

Chapter 2: Using pedometers for measuring and increasing physical activity in children and adolescents: the next step

Chapter 2 examines the current pedometer protocols for children and adolescents within the literature and provides future recommendations for objective monitoring protocols. (*Overall aim*)

Previously published as:

Lubans, D. R., Plotnikoff, R. C., Miller, A., **Scott, J. J.**, Thompson, D., & Tudor-Locke, C. (2014). Using pedometers for measuring and increasing physical activity in children and adolescents: the next step. *American Journal of Lifestyle Medicine*. doi:

10.1177/1559827614537774

Chapter 3: Reliability and validity of a single-item physical activity measure for adolescents Chapter 3 reports the test-retest reliability and concurrent validity of a single-item measure designed for adolescents. (*Overall aim and study objectives 1 and 2*)

Previously published as:

Scott, J. J, Morgan, P. J., Plotnikoff, R. C., & Lubans, D. R. (2015). Reliability and validity of a single item measure for adolescents. *Journal of Paediatrics and Child Health*, *51*(8), 787-793. doi: 10.1111/jpc.12836

Chapter 4: Adolescent pedometer protocols: examining reactivity, tampering and participants' perceptions

Chapter 4 investigates the effect of pedometer protocols on level of reactivity and tampering in a sample of adolescents. (*Overall aim and study objectives 3 and 4*)

Previously published as:

Scott, J. J., Morgan, P. J., Plotnikoff, R. C., Trost, S. G., & Lubans, D. R. (2014). Adolescent pedometer protocols: examining reactivity, tampering and participants' perceptions. *Journal of Sports Sciences*, *32*(2), 183-190. doi: 10.1080/02640414.2013.815361

Chapter 5: Young Peoples' perceptions of the objective physical activity monitoring process: a qualitative exploration

Chapter 5 uses focus group data to explore participants' perceptions of pedometers and reports reasons why adolescent may change their habitual activity pattern during objective monitoring. (*Overall aim and study objective 4*)

Previously published as:

Scott, J. J., Hansen, V. Morgan, P. J., Plotnikoff, R. C., & Lubans, D. R (2017). Young Peoples' perceptions of the objective physical activity monitoring process: A qualitative exploration, *Health Education Journal*, doi: 10.1177/0017896917734576

Chapter 6: Comparability and feasibility of wrist- and hip-worn accelerometers in free-living adolescents

Chapter 6 examines the comparability and feasibility of the wrist- and hip-worn accelerometers in a free-living adolescent population. (*Overall aim and study objectives 5 and 6*)

Previously published as:

Scott, J. J., Rowlands, A. V., Morgan, P. J., Plotnikoff, R. C., & Lubans, D. R (2016).
Comparability and feasibility of wrist- and hip-worn accelerometers in free-living adolescents, *Journal of Science and Medicine in Sport*, doi: 10.1016/j.jsams.2017.04.017 (Article in press, accepted 16 April 2017.)

Chapter 7: Discussion and conclusion

Chapter 7 includes a synthesis of study findings, chapter highlights, strengths, weaknesses and recommendation for future research.

1 CHAPTER 1: INTRODUCTION

1.1 Background

The accurate assessment of physical activity is essential for a number of reasons including: i) determining population physical activity levels; ii) identifying mechanisms of physical activity behaviour change; iii) determining dose-response relationships between physical activity and health outcomes; iv) identifying individuals at risk of physical inactivity and v) evaluating the effectiveness of physical activity interventions [11].

Identification of appropriate and accurate assessment tools is therefore a key priority in physical activity research [12]. Although self-report measures have been commonly used in physical activity studies to date [13-15], over the last ten years there has been a proliferation in the use of objective monitoring devices such as pedometers and accelerometers [11, 14, 16, 17]. Due to the considerable variability in objective physical activity monitoring protocols, and in the treatment and analysis of data, however, it is difficult to compare findings across studies [16]. There is a clear need to improve our understanding of objective monitoring protocols by examining both existing technical problems (e.g., monitor selection, site placement, minimum number or required days and data reduction techniques) and issues related to participant behaviour (e.g., non-compliance with monitoring protocols and reactivity) while being measured.

The literature review in this chapter provides a summary of the following:

- i) benefits of physical activity
- ii) current physical activity recommendations and guidelines for youth
- iii) physical activity prevalence and patterns among adolescents
- iv) existing measurement protocols for adolescents

1.

v) current issues surrounding physical activity measurement in adolescents, both technical and behavioural.



Figure 1.1. Schematic diagram of Chapter 1

1.2 Physical activity: benefits, recommendations and trends

Extensive research has clearly demonstrated the benefits of physical activity for health [5]. Compelling findings from a large number of studies have demonstrated the relationship between higher levels of physical activity and lower morbidity and mortality rates [18-20]. More specifically, it has been shown that regular physical activity can reduce the risk of cardiovascular disease, osteoporosis, diabetes and obesity, and promote psychological wellbeing [18]. A 2010, systematic review of 86 studies focusing on the health benefits of physical activity in children and adolescents concluded that increased physical activity is positively associated with better respiratory and cardiovascular health, musculoskeletal improvements and higher bone density, as well as reduced adiposity and improved mental health [21, 22]. In addition, a more recent review of the benefits of physical activity examined 15 longitudinal studies with at least a 5-year follow up and a total of 288,724 subjects (aged 18-85). The review concluded that there is a longitudinal relationship between higher levels of physical activity and lower incidence of non-communicable disease and health problems [23].

1.2.1 Current physical activity guidelines for adolescents

The World Health Organisation [24] outlined the following physical activity recommendations for young people:

- Children and youth aged 5 17 years should accumulate at least 60 minutes of moderate-to vigorous-intensity physical activity (MVPA) daily.
- Amounts of physical activity greater than 60 minutes provide additional health benefits.
- Most of the daily physical activity should be aerobic. Vigorous-intensity activities should be incorporated, including those that strengthen muscle and bone, at least three times per week.

Table 1.1 shows variations in the minimum international physical activity guidelines compared with World Health Organisation recommendations.

Country	Quantity and Intensity		Type of Activity	
	Total PA	Vigorous PA	Aerobic PA	MBSPA
Australia [25]	≥ 60 mins/day MVPA (up to several hours)	Include daily	Include variety daily	\geq 3 days/week
USA [26]	\geq 60 mins/day MVPA	\geq 3 days/week	Most of total activity should be MVPA	≥ 3 days/week (as part of total MVPA)
Canada [27]	\geq 60 mins/day MVPA	\geq 3 days/week	-	\geq 3 days/week
UK [28]	≥ 60 mins/day MVPA (up to several hours)	\geq 3 days/week	-	Include as part of vigorous PA
WHO [24]	\geq 60 mins/day MVPA	\geq 3 days/week	Most of total activity should be MVPA	Include as part of vigorous PA

 Table 1.1. International physical activity guidelines for young people (5 - 17 years)

Abbreviations: MBSPA: Muscle and bone strengthening physical activity; MVPA: moderate-to-vigorous physical activity; PA: physical activity; UK: United Kingdom; USA: United States of America; WHO: World Health Organisation.

1.1.2. Physical activity prevalence

In 2009, physical inactivity was identified as the fourth leading risk factor for noncommunicable diseases and was estimated to have caused more than three million preventable deaths globally [22]. Physical 'inactivity' is defined to be an activity level insufficient to meet current recommendations or guidelines [29]. In 2012, *The Lancet* published a series of papers that reviewed global inactivity, physical activity trends and evidence-based strategies to improve physical activity. The review concluded that physical activity was an important modifiable risk factor for chronic disease and highlighted the need for further global intervention [22, 30-33]. In 2016, The Lancet published a second series of papers, reporting the largest harmonised meta-analysis on the joint health effects of sedentary behaviour and physical activity. This series highlighted the need for policy change at both national and international levels to address the physical inactivity pandemic [34-37].

In 2012, the data collected from up to 122 countries revealed that, worldwide, 31.1% of adults are physically inactive [22]. This shows an increase in inactivity globally; prior research by the World Health Organisation in 2010 estimated that 23% of adults are inactive [19]. In Australia, an estimated 62% of adults fail to meet minimum physical activity guidelines [38]. Despite the health benefits associated with an active lifestyle, many children and adolescents also fail to meet the minimum physical activity guidelines [39]. The 2012 series of articles in *The Lancet* also provided data from 105 countries on adolescents aged 13-15 and concluded that 80.1% failed to meet the minimum international physical activity guidelines of 60 minutes of MVPA per day.

1.2.3 National and global costs associated with physical inactivity

The 2012 *Lancet* series on physical activity indicated that analysis of the global economic burden of inactivity was required to properly understand the pandemic of physical inactivity [33]. Consequently, the 2016 *Lancet* series completed an in-depth analysis of the burden of physical inactivity, and concluded that the estimated cost to health-care systems globally was \$53.8 billion in 2013. This consisted of \$31.2 billion paid by the public sector, \$12.9 billion paid by the private sector and \$9.7 billion paid by households [37]. Moreover, decades of epidemiological research and recent health reports have illustrated the worldwide negative public health impact and global financial burden of physical inactivity [5, 18, 19, 37, 40]. The associated health-care system burden in Australia is exponentially growing from year to year, and was estimated at \$1.5 billion in 2008 [41]. Both nationally and globally, these findings clearly highlight the need to prioritise resources to implement strategies that promote physical activity, to reduce the economic burden associated with physical inactivity.

1.2.4 Patterns of physical activity among adolescents

Existing cross-sectional and longitudinal studies have reported that physical activity steeply declines through adolescence [42-44]. For example, one recent longitudinal study using Norwegian adolescents (1,945 9-year-olds and 1,759 15-year-olds) reported that, from ages 9-15, both girls and boys reduced light physical activity, by an average of \geq 106.7 min/day (p <0.001), and MVPA, by \geq 20.8 min/day (p < 0.001) [43]. These findings were supported by a further longitudinal descriptive analyses study that used 1032 subjects in the 1991-2007 National Institute of Child Health and Human Development Study of Early Child Care and Youth Development birth cohort found that between the ages of 9 and 15, weekday MVPA decreased by 38 minutes, and weekend day MVPA by 41minutes [45]. In addition, National longitudinal and cross-sectional studies have revealed that as few as 15 - 19.7% of Australian adolescents achieve the recommended amount of daily physical activity [46, 47]. A recent systematic review focusing on the impact of physical activity interventions in youth examined 10 studies and concluded that further research is required to clarify the relationship between interventions and long-term impact on physical activity level, self-efficacy and weight management [48]. Further research is warranted to determine if physical activity interventions can ameliorate the decline of activity typically observed during adolescence. Consequently, adolescents are now an important target population for physical activity interventions.

7.
1.3 Measurement of physical activity

1.3.1 Assessing physical activity

There is a plethora of existing assessment instruments used to measure physical activity, and researchers are commonly faced with the issue of selecting the most appropriate measurement approach [49]. Measure appropriateness is dependent on purpose of study (e.g. intervention, clinical research, descriptive assessment/cohort studies or population surveys); sample size; age group(s); respondent burden; method/delivery mode; assessment timeframe; type of physical activity information required; data management; budget and other study limitations [50]. In addition, selection may be dependent on whether the researcher is attempting to obtain physiological response to activity (e.g. energy expenditure (EE)) or movement response to activity (e.g. mode, frequency, intensity) [51]. Physical activity measurement instruments are typically categorised into two sub-groups: 'subjective' and 'objective' [52].

Subjective measures include physical activity questionnaires, diaries, and log books. Questionnaires are typically used for large-scale population survey or studies as they have the ability to collect large amounts of data, are relatively inexpensive and are less burdensome for the participant than other measures [53]. Logbooks and diaries require participants to record their activity patterns over a period of time, typically one week [13, 15]. Objective measures include direct observation, heart rate monitors, pedometers and accelerometers [50]. The flowchart in Figure 1.2, which is modified from a review of physical activity measurement approaches [50], provides an overview of how physical activity measurement instruments are commonly categorised.



Figure 1.2 Categorisation of physical activity measurement

Abbreviations: HR: heart rate; EE: energy expenditure; AEE: activity energy expenditure; MVPA: moderate-to-vigorous-intensity physical activity; EMA: ecological momentary assessment.

1.3.2 Subjective measures of physical activity

1.3.3 Self-report measures

Physical activity self-report measures require participants to recall and report their participation in physical activity over a specified period of time. The aim of questionnaires in general is to obtain an estimate of an individual's frequency, duration and distribution of physical activity [14]. This information can be used to assess prevalence of population groups meeting physical activity guidelines; to make cross-sectional comparisons and to evaluate interventions [54]. As the most commonly used subjective method for physical activity assessment in adolescents is self-report questionnaires [15], this thesis only investigated this form of subjective measure. There are four categories of physical activity subjective measures generally used [55]:

- i) Self-report (self-completed) questionnaires
- ii) Diaries and logs (recording all activities)
- iii) Interviewer administered (e.g., face-to-face, telephone and focus groups)
- iv) Proxy reports of physical activity completed by parents, carers or teachers.

1.3.4 Strengths and limitations of self-report measures

Self-report measures are still used extensively as they are non-invasive and relatively easy to administer [12, 13]. Questionnaires are perhaps the most feasible method of assessing physical activity in large-scale epidemiological and surveillance studies as they are logistically easier, have low participant burdens and are relatively inexpensive [56-58].

Self-report data can be collected in a variety of ways. Diaries and logs are simple tools that allow participants to record their activity at regular intervals, thus reducing extended recall times and increasing the likelihood of accurate recall [59]. Individual and focus group interviews provide rich qualitative data in relation to the type, timing, frequency and intensity of physical activity, which allows researchers to gain a deeper understanding of habitual activity patterns [60, 61]. To collect data on younger children (usually < 10 years of age), parent/carer proxy reporting is commonly used in physical activity intervention studies and large-scale

Chapter 1: Background and rationale

population studies [62, 63]. Due to their low cost, low participant burden and general acceptability, physical activity self-report measures have been, and continue to be, designed to gain insight into target population level and type of activity [64].

The major limitation of self-report measures is that they are characterised by respondent bias [65]. Recalling physical activity is a complex cognitive task, and children and adolescents are less likely to provide accurate recall than adults [15]. Prince and colleagues systematically reviewed 83 physical activity measurement studies, and found that 72% of children and youth significantly over reported their physical activity levels [64]. More active youth were more likely to 'under report' their physical activity level, whereas the least active youth were more likely to 'over report' [64]. A large study of 2,761 adolescents, which compared self-reporting measures to accelerometer-determined physical activity levels, found that 65.4% of participants over reported their time spent in MVPA by at least 5 minutes per day, 20% under reported their MVPA, and only 14.6% accurately reported their MVPA level [66]. It was reported that this was a result of skewed perception of time and intensity and issues inherent with recall [67].

In contrast to children and adolescents, the literature suggests that adults are more accurate in self-reporting their physical activity; perhaps due to their greater ability to think abstractly and perform more detailed recall [14]. Although not surprising, a further study that focused on young people found that adolescents were more accurate in recalling their physical activity than were children [66]. The literature is inconsistent regarding sex differences in misreporting in youth [66, 68]. One study indicated that adolescent boys over report activity levels more than adolescent girls [69], while another found the opposite to be true [70]. The research available suggests that race and age are not associated with over reporting in adolescents [64, 66].

Consistent with previous reviews [13, 71], a recent systematic review of 96 physical activity questionnaires designed for youth and adults reported that very few questionnaires demonstrated acceptable reliability and validity [14]. The modified version of the International Physical Activity Questionnaire designed specifically for adolescents (IPAQ-A) was designed

to capture health-enhancing activity levels globally, and is still commonly used in this population. However, when validated against accelerometer output, the IPAQ-A was shown to have only acceptable validity in older adolescents aged 15–17 years, and poor validity in younger adolescents (< 14 years) [72].

The increasing interest in physical activity research has led to the development of a plethora of questionnaires designed to assess physical activity behaviours and patterns. Given the vast differences in the types of instruments available, comparing results across studies is difficult [73]. For this reason, it is challenging for researchers to identify the most appropriate instrument to address their study aims [13, 15]. A 2008, systematic review of 89 youth physical activity measures identified the Physical Activity Questionnaire for Adolescents (PAQ-A) [74], the Youth Risk Behaviour Surveillance Survey (YRBS) [75], and the Teen Health Survey [76] as the most suitable and useful tools for population surveillance, based on their validity and reliability [73]. The World Health Organisation also designed the Health Behaviours in School Children (HSBC) cross-national survey that was administered in 43 countries with 11-15year olds revealed the magnitudes of international differences in physical activity [77]. Some of the physical activity items in the HSBC survey were later adapted for use in the Australian Health Survey (2011-2013) [78].

A recent systematic review of adolescent physical activity self-report measures [15] concluded that the Oxford Physical Activity Questionnaire (OPAQ) was one of the most reliable instruments assessing adolescent physical activity (ICC = 0.89 for boys and 0.78 for girls) although it was considered to hold only acceptable validity [79]. The OPAQ is a time-based seven-day recall questionnaire for which participants are asked to report their time spent in MVPA [15]. This questionnaire was developed for adolescents, and respondents are required to report the last seven days of physical activity, including type, frequency and duration [79]., Further research is required to determine the strengths and limitations of the OPAQ in terms of its validity, however. Further research into the specificity of items and the effect this has on

Chapter 1: Background and rationale

reliability and validity will inform future work in the field of valid and reliable physical activity measures.

To reduce participant burden, researchers have experimented with shorter and less detailed questionnaires [13]. It has been suggested that shorter questionnaires may result in fewer outliers, less participant fatigue, fewer inaccuracies and less missing data, particularly among adolescents who have shorter attention spans [80, 81]. A recent study evaluated the validity and reliability of a single-item measure of physical activity for adults, and concluded that it compared well with other existing physical activity questionnaires [12]. Currently, no such instrument exists for adolescents. As inaccuracies with self-reporting continue to be an inherent threat to the validity in adolescents, researchers have been encouraged to further refine self-report measures and validate questionnaires against objective measures, such as accelerometers [14, 15].

In summary, considering the large number of existing physical activity measures available, the wide-ranging variability in their format, length and complexity, and their limitations in accuracy [13, 15], there is a need for more direct comparisons of questionnaires to improve our understanding of the most appropriate physical activity measures. The in-depth detail of most existing physical activity questionnaires and high participant burden [14, 15] highlights the need for shorter, more simplified physical activity questionnaires tailored to adolescents. There appears to be no existing single-item physical activity measures designed specifically for use in free-living adolescents. Furthermore, no previous study has examined the validity and reliability of a single-item measure by comparing it with both an existing physical activity questionnaire and accelerometer output.

1.3.5 Objective measures of physical activity

This section discusses the strengths and limitations of the following objective measures: direct observation, pedometers and accelerometers. There is some interchange in the discussion of these pedometers and accelerometers devices as they are compared and contrasted below.

1.3.6 Direct observation

Systematic (direct) observation of physical activity is a data collection procedure that involves observing an individual (or group of individuals) using an instrument to code the activity, which may result in identification of the type of activity, the intensity of activity and the place it occurs [82]. Activities are coded on time-sampled intervals; normally between five seconds and one minute [65]. Researchers then collate the data to give a representation of activity patterns based on what has been observed.

1.3.6.1 Strengths and limitations of direct observation

Direct observation has typically been used to measure activity within a specific context (such as a school or park setting) [83]. It can provide contextually-rich data on the environment, such as where the activity is being completed; who the activity is completed with (social interaction); type of activity and so on [82]. This has become particularly important when exploring the environmental and psychosocial factors that influence physical activity levels. Direct observation has been considered an appropriate criterion measure for physical activity in adolescents [84]. The major strength of direct observation is it can provide a contextually rich and comprehensive assessment of an individual's physical activity. It is particularly useful when researchers are interested in the influence that setting (e.g., parks and schools) has on the individual or group level of activity The flexible nature of the direct observation procedures also allows researchers to collect data on other factors that might influence physical activity levels, such as environmental conditions, availability of equipment, access to facilities and family or peer involvement [65].

However, direct observation can be burdensome and expensive due to the time taken to train research staff, the length of observation time, and the time spent completing coding and data imputation. It may be less suitable, therefore, for studies that are concerned with assessing habitual physical activity in large samples [65, 85]. As direct observation involves actually witnessing the activity and also recording it, it is therefore less suitable for longer-term or daily or weekly data collection. Furthermore, direct observation is not suitable for assessing physical

activity in 'free-living' participants unless they are in a confined space such as a school, gym facility or recreational area. Therefore, data collection timeframes are normally short and only provide a 'snapshot' of an individual activity pattern within a certain context [49].

In summary, direct observation has been considered a criterion measure for physical activity in youth [52]. Although somewhat limited in terms of its broader use, it remains an important tool to assist researchers to better understand the social and environmental factors that may influence adolescents' physical activity levels. [49]. Greater understanding of these environmental and social factors will assist researchers in designing and implementing physical activity behaviour change interventions.

1.3.7 Pedometers

Pedometers provide an accurate and feasible method for measuring adolescents' physical activity levels [15, 58, 86-88]. Studies have also shown that pedometer step counts are moderately associated with doubly labelled water, heart rate and VO₂, and strongly associated with accelerometer output [89-92]. There is now consensus that measurement of cumulative steps over the course of a 24-hour period is a suitable and effective measure of physical activity level in adolescents [65]. While pedometers can be worn on various parts of the body (upper arm, wrist, thigh, ankle etc.), they have are most commonly worn at the hip [93]. Pedometers do not measure acceleration, but work by detecting vertical motion. When there is a change in vertical motion, a horizontal arm bounces up and down inside the unit and records the step. More recently, piezoelectric pedometers have been developed. In these pedometers, the internal mechanism is a suspended beam and piezoelectric crystal that measure horizontal movement past the in-built threshold. The movement is displayed on a digital screen and stored into the device's internal memory [94]. The step counts can then be recorded by the individual or by the researcher periodically during data collection, or once the data collection period has been completed, if the device has the ability to store data over a series of days.

1.3.7.1 Strengths and limitations of pedometers

Pedometers provide a valid, reliable [95, 96] and cost-effective way to collect physical activity in free-living youth [94, 97]. They are easy to use and normally do not require initialisation prior to use or downloading of data after use. They offer an easily-understood measure in a standard metric (most commonly steps/day) [96]. Some pedometers, such as the

Yamax Digi-Walker CW700 (Yamax Corporation, Kumamoto City, Japan), possess the ability to store data over a seven-day period and therefore do not require daily resets. This is an important advantage when measuring youth as it removes the cognitive requirement for participants to self-report/log their step count, and reduces the chance of over/under estimating and risk of accidental resets.

While pedometers are commonly used as an objective way to assess habitual activity in young people, there are some limitations that should be noted. It has been reported that: i) wearing them can be considered invasive [98]; ii) they are insensitive to non-ambulatory movements and are normally not waterproof; hence they have to be removed for water activities [99]; iii) they are prone to data loss due to accidental resets, iv) they are subject to potential participant tampering (i.e., shaking) [58, 99]; v) they do not provide intensity or any contextual information of the activity completed [15, 93]; and vi) they normally require either the participant, teacher or parent to log the daily step count, which can lead to inaccuracies with reporting [88]. Finally, as the participant wearing the device may see the step count, there is the potential for them to increase their physical activity in response to the feedback provided by the pedometer [100].

In summary, many studies have now used pedometers to successfully collect physical activity in free-living adolescent populations [93, 101, 102]. Pedometers provide a widely and easily understood step count, which allows for simple analysis of data and comparison across studies [103]. There are, however, no standardised protocols for the use of pedometers in adolescent populations; in addition, there is limited research investigating how these protocols may influence participant behaviour while wearing the pedometers.

1.3.8 Accelerometers

Accelerometers are small devices most commonly worn at the hip and, more recently, at the wrist [104]. The monitor measures the magnitude or acceleration of movements, which can be categorised into different intensities using cut-points [49]. The majority of accelerometers use a piezoelectric sensor to detect accelerations on one to three planes (anteroposterior, mediolateral, and vertical). The accelerations are stored in the internal memory of the device and later downloaded to a computer software program for analysis [105].

Some of the accelerometers used in physical activity research such as the activPALTM (Pal Technologies Ltd, Scotland, United Kingdom), ActiGraph GT3X+ (ActiGraph, Pensacola, United States of America (USA)) and StepWatch[™] (Modus Health IIc Washington, USA) include an inclinometer feature. By assessing the tilt angle across two or more planes (forwards/backwards as well as sideways), the inclinometer provides output to estimate anatomical position in space (i.e., not wearing the monitor, lying supine, sitting, standing) [106, 107]. A recent study, using the inclinometer feature in the commonly used ActiGraph and StepWatch accelerometers, found that the inclinometer correctly identified anatomical position two thirds of the time during sedentary behaviour, and was even more accurate (71.8-85.1%)during light intensity activity; however, it was highlighted that the device had limitations when discriminating between different intensities [108]. While inclinometers have merit for assessing general behaviour pattern, their lack of sensitivity in identifying activity intensity is a noted weakness [108]. It has been hypothesised that multi-unit accelerometers that contain features such as inclinometers and physiological sensors may be the future of physical activity measurement [109]. These devices remain relatively under researched in free-living populations; thus, further research is warranted.

There is now an extensive range of commercial accelerometers designed for measuring physical activity (e.g., Fitbit[®] (Fitbit Inc, San Francisco, USA), Misfit[®] (Misfit, Burlingame, USA), Archos Activity Tracker[®] (Archos S.A, Igny, France), Vivofit[®] (Garmin Ltd, Schaffhausen, Switzerland), Jawbone[®] (Jawbone Co., San Francisco, USA). As each

accelerometer brand varies slightly, comparability between accelerometer brands remains problematic and has introduced new challenges in interpretation of data [110]. A recent systematic review of consumer-wearable activity trackers analysed 22 studies and found that Fitbit monitors showed high inter-device reliability for steps, distance, energy expenditure and sleep. However, the authors concluded that further research is needed examining the measurement properties of each device as they are released into the market [17]. While many of the devices have novel features, research into comparability across brands is sparse [111]. In addition, there is limited research into the reliability and validity of the monitors [112], clearly highlighting the need for further research into consumer-wearable monitoring devices and their use in data collection for physical activity research.

1.3.8.1 Strengths and limitations of accelerometers

Reviews of physical activity assessment methods have concluded that accelerometers are an accurate, reliable and practical objective measure of physical activity among adolescents [49, 95, 113, 114]. Survey and intervention research has shown that they are currently one of the most effective ways to assess free-living adolescents' physical activity levels [115, 116]. The biggest advantage of accelerometers is that they can capture duration, timing and intensity of physical activity [15]. Consequently, we have seen an increase in the use of accelerometers in adolescent studies [114].

Accelerometers are suitable for well-funded large population studies, and provide an accurate way to collect objective measures of physical activity. They have been used for wide-scale surveillance studies (for example, National Health and Nutrition Examination Survey [NHANES] [115]), validation studies and adolescent physical activity interventions [50]. A further benefit is that they capture activity in 'real-time', allowing categorisation of activity over different periods of the day/monitoring period [50]. As adolescents remain a difficult population to measure, due to non-adherence to monitoring protocols (discussed in more detail in Section 1.5.7), the robust waterproof nature, large storage capacity and widely-recognised reliability of

Chapter 1: Background and rationale

accelerometers make them appealing to researchers attempting to improve adherence levels and accurately quantify activity in adolescents [115].

Some recognised limitations of accelerometers have been noted in the literature. For example: i) they require pre-set initialisation/time stamping, application of filters and charging prior to use [117]; ii) they can be burdensome on the participant [61]; iii) they may be subject to reactivity [118] and iv) require downloading after use; and iv) the output can vary, requiring time-consuming complex cleaning and analysis of data [49, 65]. In addition, accelerometers are typically more expensive than heart-rate monitors and pedometers [50].

In summary, accelerometers are a valid and reliable way to objectively assess activity in adolescent populations [114]. However, the inconsistencies with treatment of missing data, application of cut-points, choice of output, highlight the need for standardised accelerometer protocols in adolescents [114, 119, 120].

1.4 Summary of physical activity assessment methods

Quantifying physical activity in adolescents is a challenging endeavour [65]. Surveys, logs and self-report questionnaires can provide units of time spent in activity (in minutes/hours), contextual information, estimated intensity level (sedentary, low, moderate, high), and activity type (which can be converted to a metabolic equivalent [MET] count). As they are subject to respondent bias, however, the accuracy in the use of self-report measures varies regardless of the output chosen [15]. The assessment of physiological markers (e.g., heart rate monitoring, doubly labelled water and direct calorimetry) offers a potential advantage over self-reporting measures in reducing bias, though it provides complex output and requires tedious analysis by trained researchers. Objective measures such as pedometers and accelerometers have emerged as a feasible and accurate way to assess physical activity in free-living adolescents [114].

The validity and reliability of various physical activity measures have been studied extensively; however, very little is known about adolescents' perceptions of the monitoring process and why compliance is consistently low in this population group [60]. Previous studies have reported that adolescents' views of the objective monitoring process are mixed (both positive and negative) and dependent on the design of the protocols and the incentives that are provided [60, 61, 100]. Further refinement of objective monitoring protocols and in-depth qualitative research is warranted to advance the field. This will lead to a better understanding of participants' perceptions of the monitoring process and assist with implementation of strategies to maximise retention and adherence to monitoring protocols. Table 1.2 shows the strengths and limitations for existing measures used in both child and adolescent physical activity research, some of which are discussed above (Section 1.3). The table illustrates the importance of selecting the right measurement tool for the study aims, budget, number and age of participants.

Table 1.2. Summary of key attributes for current methods used to measure physical activity in children and adolescents; adapted from Trost and colleagues [65].

Method	Valid	Affordability	Objective	Ease of administration	Ease to complete	Measures patterns of activity	Non- reactive	Feasibility in large studies	Suitable for age <10yrs	Suitable for age >10yrs
Questionnaires	~	√ √ √	×	~ ~ ~	~~~	~~	v v v	√√√	×	~~
Diaries/logs	~	~~~	×	<i>√ √</i>	×	~~~	~	√	×	~
Interviews	√ √	√	×	~~	~~	~~	~~~	~~	×	√ √
Proxy reports	~	√ √ √	×	$\checkmark \checkmark \checkmark$	~	~~	~~~	~ ~ ~ ~	<i>√√√</i>	~
Observation	√ √ √	×	√ √	\checkmark	√ √ √	~~	×	×	v v v	√ √
Pedometers	√ √	√ √ √	√√√	$\checkmark\checkmark$	~~	×	×	~~	VV	VVV
Inclinometer	~~	√ √	V V V	~	~~	~~	~~~	~~	~~~	~ ~ ~
Accelerometers	~~	√	$\checkmark \checkmark \checkmark$	$\checkmark\checkmark$	~~	~~	√ √ √	~~	~~~	$\checkmark\checkmark\checkmark$
 ★ Poor or inappropriate; ✓ Acceptable; ✓ Good; ✓ ✓ Excellent * Does not induce changes in physical activity behaviour as a result of being measured. 										

1.5 Specific objective physical activity measurement issues in adolescent populations

1.5.1 Introduction

This section focuses on the challenges researchers face with objective physical activity assessment in adolescent populations. As both pedometers and accelerometers are types of motions sensors they share some common issues related to adherence and analysis, there is some interchange in the following discussion. The current technical challenges for researchers include: i) device selection and comparability; ii) outputs application filters and cut-points; iii) data reduction techniques; and iv) treatment of missing data. Adolescents are a particularly difficult population to objectively measure [115], due to behavioural challenges such as noncompliance with monitoring protocols, reactivity to activity monitors and device tampering.

1.5.2 Objective measurement: technical issues

1.5.3 Selection of device and comparability

There is a wide array of existing movement detection instruments that provide varying types of output, making it difficult for researchers to select the most suitable instrument [94]. Over the past two decades, there has been a public health shift from 'exercise to develop fitness' to a focus on moderate intensity lifestyle physical activity intended to improve overall health [93]. This has led to the development of numerous studies focusing on intensities of activity and relationship to health outcomes.

A study that investigated interchangeability of pedometer brands found that there was good accuracy across brands with walking speed; however, as speed increased > 3.2kms/hour the degree of error also increased [121]. Further inconsistencies have been shown with distance travelled, as the estimates are dependent on participants' stride length and frequency [122, 123]. As a result, researchers have recommended that pedometer output should be expressed as steps/day without further estimation of distance or energy expenditure, as the level of inconsistencies may be unacceptably high for comparative purposes [95]. Studies have shown strong correlations between step counts derived from pedometers and accelerometers [124]. Accelerometers quantify a further dimension of movement by capturing velocity in relation to time; that is, acceleration. In many studies, acceleration is expressed as an 'activity count', which is an arbitrary value and not normally comparable across brands [113]. Each brand of device has its own proprietary algorithm used to process, filter and scale the raw data [104]. This leads to further issues when attempting to translate various types of output into intensity levels, highlighting the need for consistent thresholds or cut-points [95]. The lack of equivalency of devices has resulted in low comparability of measurement approaches across studies. More recently, there has been the development of tri-axial accelerometers such as the ActiGraph GT3X+/GTX9 and GENEActiv. These have the ability to collect raw unfiltered accelerations, which can be subjected to data processing procedures to allow accurate comparison of output from different devices [110].

1.5.4 Use of epochs and cut-points

Accelerometers possess a time sampling mechanism that allow researchers to quantify the intensity, frequency and duration of activity [125]. This interval is normally termed an 'epoch', which summarises activity into a count that is stored, to maximise battery life. Although 60-second epochs are considered appropriate for adults [126], children and adolescents are involved in more sporadic bouts of physical activity, and shorter epoch lengths are recommended [127, 128]. Indeed, researchers have found young people's activity is highly intermittent, with the majority of activity bouts lasting 3 - 22 seconds [129, 130]. The application of epoch length can therefore be important when attempting to accurately quantify intensity of daily physical activity levels and adherence to physical activity guidelines. One recent study containing 268 participants aged 7 - 11 years, examined effect that epoch length (1-, 5-, 10-, 15-, 30- and 60-seconds) has on estimation of daily activity and concluded that the application of epoch length can substantially affect the estimation of daily activity and time spent in various intensities of activity [131].

A study using 534 youth aged 7-16 years, tested the effect of epoch length on physical activity intensity and recommended that a 30 - 60 second epoch be used when assessing adolescents

activity [132]. In addition, a recent review of 183 studies reported that epochs ranged from 5 - 60 seconds in adolescents (63.2% of these studies used a 60 second epoch) [114]. Best practice recommendations for children, adolescents and adults are to use shorter time sampling intervals (epoch lengths) for example 10-, 15- or 30 seconds rather than longer epochs to capture more sporadic movements [133]. In the last decade, calibration studies have attempted to find the most accurate way of converting accelerometer output (i.e., counts per epoch) into time spent in different physical activity intensities, by creating and applying cut-point thresholds [134]. There are now a range of age-specific cut-points that have been introduced for various monitors [134]. A recent systematic review, of existing cut-points specifically designed for youth, concluded that there are no widely agreed upon cut-points to classify MVPA in adolescents [134]. Further studies have shown that moderate-to-vigorous thresholds derived for young people range from 615 - 3200 counts/minute even with the same accelerometer model [135-138].

A review of 11 calibration studies concluded that there was no consensus on 'the best' cut-point to use in adolescents for MVPA, and recommended further research in the area [134]. Trost and colleagues evaluated five different ActiGraph cut-points, and recommended that the 'Evenson cut-points' should be applied when estimating sedentary, light, moderate and vigorous activity in children and adolescents [139]. However, a more recent review on accelerometer protocols highlighted the need for calibration studies to focus on a wider age-range and various epoch lengths and cut-points to determine a definitive protocol [114].

Due to the complexity of issues surrounding epoch length, activity 'counts', and cutpoints, there has been an increase in the use of unfiltered raw acceleration to compare different monitor brands (i.e., for validation studies) [111]. This has negated the need for epochs and cutpoints, although currently only a limited number of devices have the ability to capture unfiltered raw accelerations.

1.5.5 Treatment of missing data and data imputation

Poor adherence to objective monitoring protocols [10, 100, 115] results in large amounts of missing data, making it difficult to obtain an estimate of normal activity patterns [140]. It has

Chapter 1: Background and rationale

been suggested that the minimum number of days required to gain a valid physical activity representation is dependent on population size and age of participants [87]. Ozdoba and colleagues found that, in children, four days of measurement was enough to get an ICC of 0.90 [58]. Further research on children found that three to four days was sufficient to obtain an ICC value of (0.70) and five days was sufficient for a value of (0.80) [119, 141]. Trost and colleagues concluded that seven days of monitoring is needed to provide a reliable estimate of usual physical activity in adolescents and account for the differences in weekday and weekend expenditure [119]. To have a single reliable timeframe for this age group would require continual large data sets containing multiple days of pedometer monitoring [87].

Procedures for managing missing data is another important issue to consider when assessing physical activity objectively [142]. Some research suggests that participants should be excluded from the analysis if there is an incomplete day of data collection; this is most commonly due to accidental resets [58] or extended periods of non–wear time [16]. As some pedometers are not waterproof, they are removed for showering and water activities such as swimming. Non-ambulatory activities such as cycling and swimming are not accurately recorded by pedometers and accelerometers, as there is little vertical movement at the hip [99]. Therefore, when completing non-ambulatory activities, participants must estimate their step count or time spent being physically active via self-report, which, as previously reported, can result in bias.

The relative contribution of ambulatory and non-ambulatory movement to adolescents' physical activity has not been tested extensively. The most basic way to estimate non-ambulatory movement (and convert into step counts) is called the Simple Conversion Method (SCM). With this approach, every minute of non-ambulatory movement is converted to 100 pedometer-based steps. The same research team developed the 'intermediate' and 'complex' conversion methods (ICM and CCM). More complex conversions are used as they can gain activity counts per minute, instead of a simple step count [143]. These approaches use more

Chapter 1: Background and rationale

complex mathematical calculations and may not be suitable for all studies, as they require more expert researchers to complete the conversions.

Rowe and colleagues proposed that, to be considered a valid estimate of activity pattern, daily step counts must fall between 1,000 and 30,000/day; they suggested that values < 1,000 and > 30,000 are implausible and should be removed [144]. It is not yet clear whether this approach is appropriate [87], nor is there a set of common protocols to follow in the treatment of pedometer-based data. Physical activity studies show much variation in the way missing values and outliers are addressed, and the way data is transformed and analysed [88]. Standardised pedometer protocols and data reduction techniques are required to facilitate accurate physical activity data collection [16].

Recent accelerometer studies have also confirmed that non-adherence to accelerometer monitoring protocols is common in adolescent populations [115, 140]. In 2005, Ward and colleagues reviewed the existing accelerometers protocols and made future recommendations for researchers to address gaps in the literature. The researchers highlighted the need for consistent methodological approaches in monitor use protocols, analysis of accelerometer data, monitor selection and integration with other data sources [145]. A decade later, there remain several unresolved methodological issues, particularly regarding treatment of missing data [146].

To address the issue of non-adherence and missing data, researchers have proposed an analytical approach, whereby collected data (recorded while monitor was worn) is used to predict segments of data where the monitor was removed; known as 'data imputation' [147]. It has been proposed that the use of data imputation software provides more accurate and less biased assessments of activity patterns [147]. Reviews of the literature suggest, however, that most studies discard incomplete or invalid days from the analysis [146, 148]. Although there is growing recognition for the use of invalid days and data imputation methods to increase the accuracy of objectively assessing activity patterns, no standardised method exists for the adolescent population. Non-wear time is classified by pre-programming the accelerometer to

detect that there is no movement (i.e., the accelerometer has been removed); this can be set in consecutive strings of zeros for an amount of time (e.g., 20 minutes, 30 minutes, 40 minutes, and so on).

Currently there is no standard recommendation for the classification of non-wear time for accelerometers [148] with existing studies in youth ranging from 10 - 180 minutes [114, 149-151]. In addition, there are no widespread recommendations for minimum wear-time per day, and days per week to collect a valid estimation of normal activity patterns. As a result of widespread low adherence to accelerometer protocols [140], more flexible inclusion criteria have emerged for adolescent populations. A recent review of accelerometer methods in youth found that the most commonly used wear-time inclusion criteria were a minimum of 8 - 10 hours of wear-time/day, worn on at least 3 - 4 days, with a non-wear time classification of 20 minutes non-wear [114].

1.5.6 Objective measures: behavioural issues

1.5.7 Reasons for non-compliance with protocols

Few studies have investigated why adherence to objective monitoring is so poor in adolescent populations [100]. A small number of qualitative studies have examined the complex issues that might influence adolescents' perceptions of the objective monitoring process [61, 100]. Kirby and colleagues used five focus groups to explore the views of young people aged 7 - 18 on objective monitoring [61]. The reasons for poor compliance reported included size and comfort, unwanted attention and increased risk of being bullied, and feelings of embarrassment. This differed from adults, who reported adult-specific issues such as occupational factors; for example, discomfort when driving [61, 152]. An additional study conducted with young people found that girls were more concerned about the look of the device than boys and recommended that compliance could be increased with use of a two-part reward (before and after monitoring) [60]. In order to understand this complex phenomenon and increase adolescent compliance with monitoring protocols, further qualitative research is warranted.

1.5.8 Device placement and compliance

Researchers have experimented with device placement in an attempt to improve compliance with objective monitoring protocols. Pedometers are most commonly worn on the hip [93], but have also been placed on the wrist and lower back. For example, Oliver and colleagues studied children in free play wearing pedometers on their backs, and concluded that pedometers worn on the back were less accurate than those worn at the left or right hip [11]. Researchers now acknowledge that waistband tilt due to torso body fat may also diminish the accuracy of pedometers [153]. One study revealed that the effect of pedometer tilt can be minimised by placing the device level laterally at the participant's hip, rather than on the anterior or posterior of the waist [154].

Similar to pedometers, accelerometers have been trialled in different placement sites on the body [155]. The ActiGraph accelerometer is the most common objective monitoring device used in the adolescent domain. The ActiGraph has traditionally been worn at the hip and secured firmly with an elastic waist belt; however, there is growing support for its comparative accuracy when worn on the wrist [115]. In addition, there has been the development of tri-axial accelerometers that can be worn on other parts of the body, such as the GENEActiv (ActivInsights, Cambridgeshire, UK). Studies in adults and children have shown higher accelerometer compliance when the monitor was worn on the wrist than when it was worn at the hip [156, 157]. It is not currently known whether accelerometer monitoring compliance can be improved in adolescent populations if monitors are worn on the wrist instead of at the hip.

1.5.9 Reactivity

Reactivity is a potential threat to the accurate assessment of physical activity, and can be defined as a change in normal activity pattern when participants are aware that they are being monitored [141]. Although many studies have explored participant reactivity to wearing pedometers, the findings have been mixed [11, 58, 86, 141, 158-160]. If reactivity exists, it is expected that participants will exhibit an increase in activity at the start of the monitoring period and then return to a more stable pattern once they become accustomed to wearing the devices

[86]. A common method for minimising reactivity is by 'sealing' the pedometer using 'zip ties' or 'stickers' [141, 160].

Some studies have proposed that sealing of pedometers can sufficiently reduce reactivity [141]. For this reason, researchers have investigated reactivity in children and adults with using sealed and unsealed pedometers in their studies [11, 141, 159, 161]. A study focusing on children concluded that, if the pedometers are sealed (thus ruling out feedback), reactivity does not occur [141]. Ozdoba and colleagues used sealed and unsealed pedometers on 45 children over a 14-day period and determined that reactivity was not present. In addition, they concluded that the sealing of pedometers also greatly reduced accidental resets, which is a common problem in pedometer studies [58].

The evidence for reactivity for pedometer monitoring in adults is mixed. Clemes and colleagues examined reactivity in adults. They concluded that the reactivity was greatest when unsealed pedometers were used [161], and also that reactivity may only last a few days [159]. A further study exploring reactivity with young adults found no evidence of change in daily activity, nor evidence of a feedback effect [86].

Most accelerometers used for research purposes do not provide real-time feedback to participants; therefore, there is less risk of reactivity. However, it has also been noted in the literature that reactivity may be caused by a participant reacting to the attention from the researchers and the excitement and novelty of being measured, which might influence their behaviour, regardless of whether or not they receive feedback from the device [118]. Behrens and Dinger studied reactivity to accelerometers in adults and found that it was not present [86]. A more recent study involving adolescents found evidence of reactivity, noting that it was more likely to occur on the first day of monitoring [118]. As the literature in this area is limited, further research into the extent of reactivity with accelerometers and the effect it has on validity is needed.

1.5.10 Tampering

Device tampering is an additional threat to the accuracy of physical activity assessment. It involves the participant purposely attempting to increase their activity counts by manually shaking the device and/or putting the device on someone or something else (e.g., pets, cars or machines) to increase activity counts [100]. Although tampering may occur with both accelerometers and pedometers, previous research has focused on tampering with pedometers.

One study that reviewed pedometer use in children and adolescents reported that, even if researchers implement safeguards (such as sealing), some participants will still shake their pedometers [162]. A later study examining adolescent adherence to a pedometer protocol found that 30 of the 43 participants self-reported tampering with their pedometers [100]. Surprisingly, no other studies have explored pedometer tampering among adolescents and the potential threat to validity.

As noted previously, adolescents are a difficult group to measure, due to their poor compliance with monitoring protocols [115]. Further understanding of their behaviours while being monitored and their perceptions of the monitoring process is needed to improve existing objective monitoring protocols.

1.6 Chapter summary

Large-scale physical activity research has demonstrated that many adolescents fail to meet the minimum recommended daily guidelines. This chapter highlighted the current behavioural and technical issues that researchers face when attempting to measure physical activity in adolescents. Furthermore, it reinforced the need for valid, accurate and reliable measurement protocols for adolescents. To date, no previous study has examined the validity and reliability of a single-item measure in adolescents by comparing it to both an existing physical activity questionnaire and accelerometer output. Currently there are no globally accepted protocols for the objective measurement of habitual activity in adolescents. This thesis addresses this gap in research by testing the existing research tools for collecting physical activity levels including self-report measures, pedometers and accelerometers.

1.7 Research aim and study objectives

1.7.1 Primary aim

The overall aim of this thesis was to investigate current physical activity measurement protocols to assess physical activity levels in adolescents.

1.7.2 Study objectives

The study objectives of this thesis were to:

- 1. Assess the test-retest reliability of a single-item physical activity questionnaire for adolescents;
- 2. Determine the concurrent validity of a single-item physical activity measure for adolescents, by testing it against accelerometer output;
- 3. Explore the impact of different pedometer monitoring protocols on compliance, reactivity and tampering in a sample of adolescents;
- 4. Explore adolescents' perceptions of wearing pedometers and investigate behaviours exhibited by participants while wearing pedometers;
- 5. Test the comparability of hip- and wrist-worn accelerometers in the free-living adolescent population; and
- 6. Compare wear-time, missing data and participant perceptions of hip- and wrist-worn accelerometers in the adolescent domain.

CHAPTER 2: USING PEDOMETERS FOR MEASURING AND INCREASING PHYSICAL ACTIVITY IN CHILDREN AND ADOLESCENTS: THE NEXT STEP

Preface: This chapter presents an overview of the current pedometer protocols for measuring physical activity in children and adolescents. The objective of this review was to evaluate the current evidence and identify future research directions.

Citation:

Lubans, D. R., Plotnikoff, R. C., Miller, A., Scott, J. J., Thompson, D., & Tudor-Locke, C. (2014). Using pedometers for measuring and increasing physical activity in children and adolescents: the next step. *American Journal of Lifestyle Medicine*. doi: 10.1177/1559827614537774

Published in the America Journal of Lifestyle Medicine (Accepted, 10 July 2014).

2 Using pedometers for measuring and increasing physical activity in children and adolescents: the next step

2.1 Abstract

The science and practice of step counting in children (typically 6 to 11 years) and adolescents (typically 12 to 19 years) has evolved rapidly over a relatively brief period of time with the commercial availability of research-grade pedometers and accelerometers. Recent reviews have summarized considerations for assessing physical activity using pedometers in young people (both children and adolescents), but three areas have received little attention: pedometer monitoring protocols, minimal (as opposed to optimal) step counts necessary for maintaining basal levels of health, and appropriate pedometer-based interventions for young people. Therefore, the objective of this review was to evaluate the current evidence and identify future research directions in these areas. The challenges of objective monitoring of physical activity in children and adolescents reinforce the importance of using protocols that minimize participant burden and the potential for tampering/reactivity. Evidence for a sedentary lifestyle cut-point is limited, researchers are therefore encouraged to investigate several cut-points (i.e., < 5000, < 6000, < 7000 steps/day) in children and adolescents to identify the health consequences of very low levels of ambulatory activity. Personalized messages may be necessary for health behaviour change in pedometer-based interventions, but there is a need for more high-quality studies.

2.2 Introduction

The science and practice of step counting in children (typically 6 to 11 years) and adolescents (typically 12 to 19 years) has evolved rapidly over a relatively brief period of time with the ever-increasing commercial availability of research-grade pedometers and accelerometers. Unlike accelerometers, which are more expensive and generally require specialized software to interpret data, pedometers provide a cost-effective and feasible approach for measuring ambulatory physical activity in young people (both children and adolescents) [102]. In 1997, Rowlands, Eston, and Ingledew [163] wrote a seminal article presenting the potential for using

pedometry to study children's free-living physical activity and subsequently followed up with the first publication of expected values for steps/day in 8 to 10 year old children [164].

Today, a simple PubMed search (3rd December 2013) using the keywords "children" and "pedomet*" yields over 300 articles. Among these include two methods-based papers, [88, 98] a systematic review of pedometer-based intervention in young people (i.e., children and adolescents), [93] and a review article [102] compiling expected values for children's and adolescents' steps/day on weekdays vs. weekend days, and steps accumulated during school, recess, physical education (PE) classes, and after school. An international effort has produced a researchers' consensus statement addressing the question of "how many steps/day are enough?" in terms of children's and adolescents' health [165]. Steps/day are also now routinely collected as an outcome of interest in large accelerometer-based studies, [166] and recently the protocol of accelerometer-determined peak cadence (steps/min) indicators (a measure of the best daily effort) has been applied to children and adolescent data [167].

Recent reviews have summarized considerations for assessing physical activity using pedometers in children and adolescents [10, 88, 98]. These reviews have provided recommendations regarding pedometer monitoring periods, wear time, data treatment, reporting and choice of pedometer. However, there are a number of issues relating to pedometer use in young people that have received little attention. First, systematic research comparing the quality of data obtained from different protocols is sparse [8, 10]. Specifically, few studies have explored the factors contributing to reactivity and tampering, especially in adolescent populations. Second, although Tudor-Locke and colleagues have provided evidence for a sedentary lifestyle index for adults (i.e., < 5,000 steps/day), [168] researchers have failed to yet identify an equivalent value for children and adolescents. Physical inactivity has serious health consequences for young people, [21, 169] but is there a minimum number of steps necessary to prevent ill-health in young people. Finally, pedometers and step counting devices have been used extensively in interventions to promote physical activity in adults [170, 171], but less is known regarding the utility of pedometers for increasing physical activity in young people. As technology evolves and proliferates, so does the potential for using pedometers in behaviour

change interventions, raising the question, "How can pedometer-based interventions be implemented to optimally increase young people's physical activity?"

Therefore, the aims of this narrative review are threefold: 1) to discuss pedometer monitoring protocols for young people and explore issues of reactivity and tampering: 2) to evaluate the evidence for establishing a step-defined sedentary lifestyle index, perhaps separately for children and adolescents: and, 3). to present pedometer-based interventions undertaken to date for young people and identify research directions focused on optimizing their positive effects on physical activity in children and adolescents.

2.3 Pedometer Monitoring Protocols

Although there has been a proliferation of studies using pedometers to measure physical activity in children and adolescents [10, 88, 98, 144], little research has focused on comparing different monitoring protocols (in terms of maximizing best quality data) and young people's reactions to the assessment process. Reactivity (i.e., a change in normal activity pattern as a result of being monitored) is considered an inherent threat to the accuracy of pedometer data collection [172]. Although some studies have revealed little evidence of reactivity in children and adolescents [11, 52, 58, 144], others have shown reactivity is present in young people [100, 172]. Sealing pedometers (e.g., using cable ties or adhesive stickers/tape) limits access to feedback and the potential for peer competitiveness [10, 100]. Monitoring for extended periods of time may also diminish reactive behaviour attributed to device novelty [58, 141, 172], but the increased burden on participants may lead to lower levels of compliance. Daily un-sealing/sealing of pedometers by study staff recording data in schools can be an administrative burden for teachers in large-scale studies [144]. Pedometers that have on-board memory functions and thus the ability to record for multiple days are useful for addressing this logistical inconvenience, reducing the effect of visual feedback to the wearer, and eliminating any need for participants to record their own data. Such pedometers may also address the challenges of

collecting step counts on weekend days when children are not as easily trackable as when they are gathered together on school days [173].

Pedometer tampering ('shaking' or 'rattling' the pedometer to increase step count) may result in further inaccuracies in pedometer data collection [95, 100, 144, 174], however, there is little systematic research to illuminate the magnitude of this potential threat to validity. Almost half of 123 adolescents reported tampering with their pedometer in a recent study [172]. Similarly, 69% of 43 children admitted to shaking the device in another study [100]. Frequency of reported tampering alone does not quantify the potential magnitude of the threat. A few shaken steps produced as a result of a curious child's interest in a pedometer's mechanism may be a trivial issue relative to a day's worth of actual ambulatory steps. Characteristics of individuals given to tampering/reactivity are unknown. At odds with concerns for reactivity of measurement (which would theoretically produce inflated estimates) is the more pressing concern that children and adolescents actually accumulate fewer steps/day than expected, considering their age. Nevertheless, further study of pedometer tampering in child and adolescent populations is needed and strategies to understand, quantify and ultimately reduce/eliminate or tolerate/accommodate such behaviours to improve interpretation of pedometer monitoring studies.

Pedometers provide a feasible (e.g., practical, cost-effective etc.) way to collect objective physical activity data from large groups, but strategies to overcome potential reactivity and tampering, or at least interpret data cognizant of this possibility, require consideration. Further testing of pedometer monitoring protocols and innovative experiments, such as covert monitoring (when participants are unaware that their activity levels are being monitored) [98], may be necessary to identify the optimal measurement protocols for assessing physical activity in young people.

2.4 A Step-Defined Sedentary Lifestyle Index for Children and Adolescents

Lower levels of self-reported and accelerometer-determined physical activity have been associated with increased risk of detrimental health outcomes in children and adolescents, including higher cholesterol and blood lipid profiles [175], higher blood pressure [176], increased incidence of metabolic syndrome [175, 177], and increased incidence of obesity [178, 179]. Of the direct associations between low step/day and health outcomes in children and adolescents, less desirable body composition [180-182] and lower fitness levels [183-185] have been reported. In addition, Barreira and colleagues recently demonstrated that children and adolescents with higher peak cadence (i.e., steps/minute) had fewer cardiovascular disease risk factors [167]. Despite this emerging evidence, the question of "How many steps/day are too few for young people?" has not been answered.

Recently, Tudor-Locke et al [168] proposed a step-defined sedentary index of < 5000 steps/day in adults, that includes consideration of population distribution, socio-demographic characteristics, contextual factors, health risks associated with taking < 5000 steps/day, and the health effects associated with increasing steps/day above 5000. Low step counts may indicate that an individual has spent more time engaged in sedentary behaviour [i.e., described as activities that involve minimal energy expenditure (1 to 1.5 metabolic equivalent multiples of rest), typically performed while sitting or lying down] [186]. Tudor-Locke and colleagues [168] have argued that estimating time spent in sedentary behaviour from lack of steps is consistent with the approach of using low accelerometer counts (e.g., < 100 counts per minute) [187]. Using data from cross-sectional [187] and experimental studies [188], Tudor-Locke et al [168] provide evidence to support the use of low step counts to indicate a sedentary lifestyle (i.e., one characterized with by more sedentary behaviour and less ambulatory behaviour) in adults.

As yet, there is limited evidence to support the creation of a sedentary lifestyle index for children and adolescents. Population distribution data among Canadian young people [189] indicate the lowest 15^{th} percentile performing < 8448 and < 7761 step/day in boys and girls, respectively (5 to 13 years). Utilizing the 15^{th} percentile cut-point in US data [166] highlights the population specific nature of distribution data, with < 6040 and < 4855 steps/day in boys and girls, respectively. Whilst this normative information is valuable, it does not provide evidence of the health-related consequences of low step counts for children and adolescents.

To date, BMI referenced cut-points for normal weight and overweight/obese children have been used as a health-related index for steps/day. Using children's step count data from the US, Australia and Sweden, Tudor-Locke et al [182] identified step counts of 12000 for girls and 15000 for boys as criterion-referenced cut-points. These analyses get at "how many steps/day are enough?" and interpreted dichotomously (yes/no), suggest that those not achieving these cut points are not achieving "enough". A proposed graduated index [190] based loosely on these BMI determined cut-points (the posited values serve as anchors), includes multiple levels, including a sedentary lifestyle index for boys and girls of < 7000 steps/day. Using a recent suggestion of a non-sex-specific step-defined sedentary index for young people of < 7000 steps/day [168], approximately one-quarter of Canadian boys and one-third of girls accumulated < 7000 steps/day (5 to 19 years) [191].

Importantly, the question of "How many steps per day are enough?" is not the same as "How many steps per day are too few?" The former focuses on an optimal level and the latter on a minimal level to be interpreted as a "red flag" for intervention purposes. To clarify further, a sedentary lifestyle index for young people may enable the identification of individuals at the greatest risk of serious health consequences due to low ambulatory lifestyle behaviours. While the existing evidence base is limited, researchers are encouraged to investigate several cutpoints (i.e., < 5000, < 6000, < 7000 steps/day) to identify the health consequences of falling below this threshold for children and adolescents.

2.5 Using Pedometers to Increase Physical Activity in Young People

Behavioural interventions incorporating pedometers have been used to increase physical activity in child, adolescent and adult populations [93, 170, 192]. The principle underlying the use of pedometers to increase physical activity is that the 'real time' step count feedback increases an individual's awareness of how their personal behavioural choices influences their physical activity [93]. Pedometer-based programs provide individuals with up-to-the-minute information and encourage them to self-monitor and set step goals using tailored (i.e., based on specified baseline values), standardized (e.g., percentage-based increments) or pre-determined (e.g., an increase of 2,000 steps/day each month) step targets [183, 193-195]. A range of new and innovative ways to use pedometers and deliver pedometer-based interventions is emerging in the literature.

A previous systematic review of pedometer-based interventions targeting children and adolescents identified three major pedometer-based strategies for increasing physical activity [93]: (i) self-monitoring and goal setting interventions based on personalized or standardized step targets [183, 193-195], (ii) open-loop feedback interventions which involve making access to desirable sedentary activities such as television watching contingent on achieving step targets [196, 197] and, (iii) physical activity integration interventions that involve using pedometers as educational tools to increase physical activity throughout the school day [11]. All three strategies were found to contribute to increased physical activity, but due to the small number of studies, the high risk of bias and lack of low term follow-up in published studies at that time, the authors of the review were unable to provide optimal guidelines for pedometer-based interventions for young people [93].

One of the limitations identified in the review was the lack of theory in guiding the development of pedometer-based interventions for young people. In adults, theory-based interventions appear to be more effective in changing behaviour than atheoretical approaches [198, 199] and are hypothesized to impact upon relevant cognitions, which in turn influence behaviour [200]. Despite evidence for the importance of theory, few pedometer-based interventions for young people have aligned their behaviour change strategies with a health behaviour theory. Notable exceptions were the Learning to Enjoy Activity with Friends [183, 201] and Program X [193, 202] interventions, which were guided by social cognitive theory (SCT) and designed to target hypothesized mediators of behaviour change (e.g., self-efficacy, outcome expectations, social support) [201]. While Zizzi and colleagues [203] did not explicitly cite their theoretical framework, their intervention appeared to be guided by SCT and they measured potential mediators from SCT. However, none of the three interventions [183, 193, 203] found evidence for the

mediating effect of any measured SCT variables on changes in steps/day. The failure to identify the mechanisms of behaviour change is likely due to the poor measurement of theoretical mediators. Recent reviews examining the effects of physical activity interventions on mediating variables have noted the lack of significant findings and the challenges of accurately measuring constructs in young people [204-206].

Pedometer-based interventions designed to increase obese adolescents' steps/day have involved cognitive behavioural therapy [207] and coping skills training [208], also based on SCT. Similar to other pedometer-based interventions targeting healthy weight adolescents, participants in these studies [207, 208], were provided with pedometers and log books and encouraged to self-monitor their step counts. Participants were also encouraged to identify barriers to physical activity and formulate strategies to increase their steps and maintain positive health behaviour change. These studies however did not provide detailed descriptions of the self-monitoring procedures and therefore it is difficult to evaluate the strategies and recommendations. Goldfield and colleagues [209] employed an alternate approach to promote physical activity in obese children and demonstrated that making access to desirable sedentary activities (e.g., TV watching) contingent on physical activity can increase step counts [196, 210]. While this approach appears to have some utility, there is concern that treating sedentary activities as rewards may undermine children's autonomous motivation for physical activity and project an unhealthy message about the perceived value of sedentary behaviours.

One possible explanation for the failure of existing health behaviour models to adequately explain physical activity behaviour change in pedometer-based interventions for young people is that such models were originally designed for 'at-risk' adult populations. Motivation for physical activity changes over the lifespan [211], and while adults may engage in physical activity to reduce their risk of lifestyle diseases, such outcomes are unlikely to be important to young people. Furthermore, theoretical models that fail to address the social, cognitive and biological changes that occur during the transition from childhood to adolescence [212], are unlikely to provide a better foundation for behaviour change [213]. The importance of

integrating health behaviour theories [214] and adopting socio-ecological models [215] has been noted in the literature, yet such frameworks have not been adopted in pedometer-based interventions for children and adolescents. Health behaviour models such as the Youth Physical Activity Promotion Model [216] and Competence Motivation Theory [217] may have utility for guiding pedometer-based interventions for young people, but they are yet to be tested in experimental studies.

Technological advancements and in particular, the proliferation of social media, exergaming, and smartphone technologies have provided researchers and health professionals with exciting opportunities to combine pedometers with eHealth technology (e.g., internet and smartphone applications) to promote physical activity in young people. Such approaches are appealing as public health initiatives due to their potential for cost effectiveness and their considerable reach [218]. Young people's access to technology is increasing at a rapid rate and in developing nations there appears to be little evidence of a 'digital divide'. For example, smartphone ownership among young people has accelerated rapidly in recent years [219] and does not appear to be moderated by socio-economic status [220], creating an ideal opportunity for equitable health promotion. Although smartphones have in-built accelerometers with step counting features and global positioning systems, their size and design may prevent them from being worn during certain types of physical activity (e.g., games and sports). Furthermore, the validity and reliability of such features are only starting to emerge in the literature [221] and due to the broad range of technologies and brands available, it will be difficult to standardize results across studies.

One of the first studies to incorporate eHealth technology into a pedometer-based intervention for adolescents was Program X [193, 202]. Participants in the Program X intervention attended interactive seminars on goal setting and self-monitoring and were provided with pedometers and sent personalized email messages encouraging them to achieve their step count goals derived from baseline step counts [193]. The intervention resulted in a significant increase in step counts for boys (approx. 1000 steps/day) and girls

(approx. 2000 steps/day), but the strategy for generating personalized feedback was labour intensive for the research team and not feasible for large population groups. In contrast, the Nutrition and Enjoyable Activity for Teen Girls (NEAT Girls) intervention [222, 223], used bulk SMS messaging to reinforce health behaviour change. Participants in the NEAT Girls intervention were provided with pedometers and sent weekly generic SMS messages during the intervention period. However, there was no intervention effect on accelerometer-determined physical activity at the 12-month post-test [224]. Bulk SMS messaging was considered to have good reach, as messages were sent to 91% of girls, but the SMS messages were not rated highly by all participants, some of whom described the messaging as 'intrusive'.

It appears that pedometer-based interventions incorporating eHealth technologies may require a tailored component to engage adolescents. The multi-component Active Teen Leaders Avoiding Screen-time (ATLAS) program, included a purposely built smartphone application (app) to promote physical activity and reduce sedentary behaviour in adolescent boys [225]. A unique aspect of the ATLAS app was that it included tailored physical activity messages, based on information reported by participants and once the app was downloaded, participants received biweekly messages sent via 'push notifications' through the app. Although research findings are yet to be published, feasibility data suggests that the app was rated highly by participants and may have utility for physical activity promotion in young people. Similarly, Thompson and colleagues [226] recently evaluated a 12-week pedometer-based intervention guided by Self-Determination Theory (SDT) for adolescents. Participants were not given a daily step goal; rather, consistent with SDT, they were told daily step goals that experts recommend for teenagers [182], along with their personal average step counts (extracted from 7 days of accelerometry at baseline). Preliminary evidence suggests modest increases in moderate-to-vigorous physical activity (MVPA) occurred in the expected directions.

Although the number and quality of pedometer-based interventions designed to increase physical activity in young people are increasing at a rapid rate, there are barriers to
their successful implementation, evaluation and interpretation. First, there is little evidence to suggest that behavioural changes resulting from pedometer-based interventions are sustainable. It is possible that participants become fatigued with wearing their pedometers and regress to their pre-intervention physical activity levels. The majority of studies have been evaluated over short periods of time (i.e., < 6 months) and longer term studies are clearly needed. Second, pedometer-based interventions incorporating eHealth technologies must manage the congestion and competition from other commercial and social medial efforts competing for space using the same media. This may contribute to information overload, thus reducing the efficacy of such approaches for behaviour change in young people. Finally, as new measurement devices emerge (e.g., Fitbit[®] and Jawbone[®] monitors) that can synchronize with a user's smartphone and provide instant feedback regarding step counts, estimated energy expenditure and time spent in physical activity of various intensities, the basic pedometer may become obsolete. However, it is unlikely that pedometers will disappear any time soon. Pedometer-based interventions remain a feasible and effective strategy for increasing physical activity in people of all ages, in part due to their accessibility (i.e., pedometers can be cost-effectively distributed to a large group) and easy-to-interpret feedback. There is clearly a need for research comparing the effects and cost effectiveness of more simplistic pedometer-based interventions to those using new measurement devices (e.g., Fitbit[®] and Jawbone[®] monitors), both supported by similar eHealth methods.

2.6 Using Pedometers in Schools to Promote Physical Activity in Young People

Schools provide an ideal setting for physical activity promotion among children and adolescents as they have access to most of the population and have the necessary facilities, equipment and personnel to achieve this outcome [227]. Physical education (PE), school sport, physical activity integration into key learning areas (e.g., mathematics and science), active transportation, after school and break times represent opportunities for physical activity promotion in and around the school setting. Using pedometers to promote and monitor activity levels in primary and secondary schools is appealing because these devices offer an affordable and accessible technology that provides output in a simplistic format that is easy to understand (i.e., steps/day) [171].

PE is commonly recognized as the major vehicle for the physical activity promotion in young people and lessons that involve high levels of MVPA can make an important contribution to young people's overall physical activity levels and their health [228, 229]. Existing US guidelines suggest that students should be engaged in MVPA for 50% of PE lesson time [40]. Scruggs has demonstrated that pedometer steps/min intervals of 82 - 88 for Yamax SW651 and SW701 pedometers [230] and 76 - 80 for Walk4Life W4L LS2505 and DUO pedometers [231] are equivalent to the 50% MVPA recommendation for PE in middle schools. These step rates can be used by teachers to evaluate their PE lessons, by asking students to wear pedometers during class and then dividing students' total step counts for the lesson by the duration of the lesson. There are also commercial pedometers available that can track time above selected steps/min cut points, automating this practice if desired.

Integrating movement into key learning areas, such as mathematics, geography and science represents another opportunity for physical activity promotion in the school setting. In one of the earliest studies of its kind, Oliver and colleagues [11] evaluated the feasibility of implementing an intervention that used pedometry to integrate physical activity (subjects included English, social studies, mathematics, statistics and PE) throughout an elementary school curriculum. Although the overall intervention effect was not statistically significant, increases of approximately 2000 steps/day were observed among children with initially low activity levels. More recently, Riley and colleagues [232] evaluated the Encouraging Activity to Stimulate Young Minds (EASY Minds) physical activity integration program for elementary school students. Similar to the Take 10! program and other physical activity integration interventions that require teachers to provide bouts of MVPA related to curriculum outcomes [233, 234], EASY Minds aimed to improve student activity levels, engagement and attainment in numeracy through the use of cross-curricula teaching strategies. For example, pedometers were used as learning tools to reinforce key concepts

regarding measurement, distance and speed. Preliminary findings suggest that the EASY Minds intervention significantly improved students' MVPA and reduced their sedentary time measured using accelerometers during the school day. The findings from these studies highlight the potential of pedometers for promoting physical activity within the school day by using their user-friendly and quickly accessible output for a range of learning outcomes across key learning areas.

Recess and lunch breaks, as well a time before and after school while children aggregate on campus, represent important school-based opportunities for young people to be physically active [235, 236]. However, the potential contribution of these key time segments is dependent upon the availability of school facilities and existing policies that support or inhibit student activity levels [237-240]. For example, schools might have high quality indoor gymnasiums and well-manicured fields, but only allow students access to facilities during scheduled PE lessons. Interestingly, a review of studies designed to evaluate the impact of school-based policies and built environment changes on energy expenditure found that mandatory PE, classroom activity breaks, and active commuting to and from school produced the largest effects [237]. Pedometers provide a feasible means for assessing such physical activity policy initiatives in schools [241] and their immediately available and interpretable feedback make them appealing to researchers and teachers.

2.7 Conclusions and Future Research

Pedometers are now used regularly in population surveillance studies to determine youth activity levels and in health behaviour interventions to promote physical activity. This review focused on three areas related to pedometer monitoring in young people, which have received limited attention in the research literature: pedometer monitoring protocols, minimal step counts necessary for maintaining basal levels of health, and appropriate pedometer-based interventions for young people. The challenges of objective monitoring of physical activity in children and adolescents reinforce the importance of using pedometer protocols that minimize participant burden and the potential for tampering and reactivity.

There is little evidence for a sedentary lifestyle cut-point in young people and the health consequences of very low ambulatory activity have not been established. Personalized messages and feedback may be necessary for health behaviour change in pedometer-based interventions for children and adolescents, but few long-term studies have been conducted. As a final note, we offer the following suggestions for future research:

- 1) There is a need to further explore the utility of different pedometer measurement protocols and identify optimal strategies for improving compliance in population monitoring studies, particularly in adolescent populations. While pedometers remain a valid and reliable method for establishing physical activity levels in youth, researchers should avoid using the same instrument to both measure and motivate physical activity in the same study sample.
- 2) Further study of the physiological and psychological health-related outcomes of excessively low ambulatory behaviours is required to develop a step-defined sedentary lifestyle index for young people. Researchers are encouraged to investigate several counts points (i.e., < 5000, < 6000, < 7000 steps/day) in child and adolescent populations to identify the health consequences of falling below this threshold.
- 3) Further research is warranted to identify the ideal theory or combination of theories to guide pedometer-based physical activity interventions for children and adolescents. Additional questions remain unanswered that were not explicitly covered here in details: What combination of procedures and components is most effective in the promotion of physical activity in pedometer-based studies with young people? What are the long-term effects of pedometer-based interventions? Can pedometer-based interventions be regularly repeated in the same population? What is the ideal program length for a pedometer-based intervention and is it necessary to include "booster" sessions or other forms of contact once the intervention period has ended to maintain step counts? Are tailored interventions (i.e., those including personalized feedback on performance) more effective at increasing physical activity in young people and if yes, what are the ideal characteristics on which to tailor pedometer-based interventions?

Chapter 2: Using pedometers to measure and increase physical activity

How can pedometers be integrated into the school environment to induce sustainable

behaviour change?

CHAPTER 3: RELIABILITY AND VALIDITY OF A SINGLE-ITEM PHYSICAL ACTIVITY MEASURE FOR ADOLESCENTS

Preface: This chapter examines self-report measures and in particular, a single-item physical activity measure that was designed for the purposes of this study. The single item physical activity is tested against a previously validated physical activity questionnaire (The Oxford Physical Activity Questionnaire) and accelerometer output in a free-living adolescent population.

Citation:

Scott, J.J., Morgan, P. J., Plotnikoff, R. C., & Lubans, D. R., (2015), Reliability and validity of a single-item physical activity measure for adolescents. *Journal of Paediatrics and Child Health*, 51(8): p. 787-93. doi: 10.1111/jpc.12836

Published in the Journal of Paediatrics and Child Health (Accepted 23 December 2014).

3 Reliability and validity of a single-item physical activity measure for adolescents

3.1 Abstract

Aims: The aim of this study was to examine the test-retest reliability and concurrent validity of an adolescent single-item physical activity measure by comparing it to the existing Oxford Physical Activity Questionnaire (OPAQ) and accelerometer output.

Methods: Participants were 123 adolescents $(14.7 \pm 0.5 \text{ years})$ from three secondary schools in NSW, Australia. To determine reliability, participants completed both questionnaires on two occasions separated by two weeks. To assess validity, participants wore ActiGraph GT3X+ accelerometers for a seven-day monitoring period and completed both physical activity questionnaires. Bivariate correlations between self-reported moderate-to-vigorous physical activity (MVPA) and accelerometer MVPA minutes/day were calculated.

Results: The single-item [Intraclass correlation coefficient (ICC) = 0.75, 95% CI = 0.64 - 0.83, p < 0.001)] and the OPAQ (ICC = 0.79, 95% CI = 0.69 - 0.86, p < 0.001) were both found to have moderate-to-strong reliability. Correlations between self-reported and objectively measured MVPA were similar for the single-item measure (r = 0.44, 95% CI = 0.24 - 0.63, p < 0.001) and the OPAQ (r = 0.50, 95% CI = 0.30 - 0.65, p < 0.001).

Conclusions: These findings suggest the single-item measure can provide a reliable and valid assessment of youth physical activity.

3.2 Background

The accurate assessment of physical activity is an on-going challenge for researchers [242] and in recent years there has been an escalation in the use of objective measures [243]. Whilst accelerometers provide an objective estimate of physical activity, there are limitations to the devices. They are expensive, don't provide activity type and there is still uncertainty as to the most effective data management and reduction protocols [125, 139, 244]. Alternatively, selfreport measures are easy to administer and cost effective for large samples. However, they are subject to respondent bias, over and under-reporting and may require complex cognitive recall [13, 15, 245, 246]. Physical activity measurement is used for determining prevalence/trends, identifying determinants and to evaluate interventions; hence the method of physical assessment (objective or subjective) is tailored to study aims/feasibility [247] (refer to figure 1; Hardy et. al, 2013).

The growth of physical activity measurement research has led to the development of a plethora of physical activity questionnaires making it challenging for researchers to identify the most appropriate instrument to address their study aims [13, 15]. A recent systematic review of adolescent physical activity self-report measures concluded that the Oxford Physical Activity Questionnaire (OPAQ) was one of the most reliable instruments assessing adolescent physical activity (ICC = 0.89 and 0.78 for males and females, respectively) although was only considered to hold acceptable validity [15, 79]. As more detailed questionnaires, such as the OPAQ, require higher level cognitions and take longer for participants to complete; simpler, less burdensome questionnaires have been developed [12, 248-250]. These shorter questionnaires are easy to administer and can be used for screening purposes and use in population surveys. Research has revealed that single-item or short physical activity measures compare favourably in terms of validity and reliability with more detailed questionnaires in adult populations [250, 251]. Milton and colleagues (2010) found satisfactory concurrent validity of a single-item measure against the commonly used Global Physical Activity Questionnaire (GPAQ) and Active People Survey (APS) in a sample of 480 adults. They also demonstrated test-retest reliability of the single-item measure in both weekly (Cohen's kappa = 0.63; 95% CI: 0.54 - 0.72) and monthly (Cohen's kappa = 0.76; 95% CI: 0.69 - 0.82) recall comparable with more complex physical activity questionnaires [12]. However, while single-item physical activity measures have been developed and tested in adult populations [12, 250], little is known regarding their validity and reliability in adolescent populations.

Considering the large number of existing subjective physical activity measures available [13, 15], more direct comparisons of questionnaires are required to improve our understanding of the most appropriate physical activity measures. To the authors' knowledge [14, 15], no

single-item physical activity measures exist for the adolescent population; furthermore no previous study has examined the validity and reliability of a single-item measure by comparing it to both an existing physical activity questionnaire and accelerometer output. Milton and colleagues recently published a study focusing on an equivalent single item measure for use with adults and concluded that the single item questionnaire had potential for screening participants for physical activity interventions [12]. Based on this finding, we believe that there will be interest a single item measure for both clinicians and for researchers who are examining multiple health behaviours and determinants in population level surveys, where costs of items are at a premium. In addition, adolescents have a noted lack of concentration further illustrating the need for shorter questionnaires to capture physical activity accurately. Therefore, the aim of this study was to examine the test-retest reliability and concurrent validity of an adolescent single-item physical activity measure by comparing it to the existing Oxford Physical Activity Questionnaire (OPAQ) and accelerometer-based physical activity levels.

3.3 Methods

3.3.1 Study design

This study recruited six physical education (PE) classes from three secondary schools in the Hunter Region, NSW, Australia. To determine test-retest reliability, participants completed the OPAQ and the single item measure on two occasions, separated by two weeks. To assess concurrent validity, participants wore accelerometers (ActiGraph GT3X+, Pensacola, USA) for a seven-day monitoring period and completed both questionnaires on the final day of monitoring. The study was approved by the University of Newcastle review board (human research ethics committee) and parents and participants provided written informed consent and assent respectively.

3.3.2 Participants

For analysis, the sample consisted of adolescents (n = 123) (Mean age = 14.7 ± 0.5 years). Oneway analysis of variance and chi-square analyses indicated there were no significant differences in cultural background or country of birth between the three different school groups (p > 0.05). Of the 160 forms that were distributed, 123 participants returned signed consent letters (i.e., 76.9% response rate), although only 96 provided valid accelerometer data to be included in the analysis; 107 provided valid single item data and 104 participants provided valid OPAQ data.

3.3.3 Research questions:

1) What is the test-retest reliability of a single-item physical activity measure for adolescents?

2) What is the test-retest reliability of the OPAQ?

3) What is the validity of a single-item physical activity measure for adolescents against accelerometer-based physical activity levels? (480min wear time/day and 600min wear time/day)

4) What is the validity of the OPAQ against accelerometer determined accelerometer-based physical activity levels? (480min wear time/day and 600min wear time/day)

3.3.4 Procedures

Data were collected in the fourth school term of 2011 (i.e. October to December). Participants were blinded to the study aims and were told that the purpose of the study was to provide an estimate of their 'usual' physical activity [252]. Prior to the first day of monitoring, accelerometers were pre-set to record activity in 15 second epochs for the seven-day monitoring period with computer software Actilife 5, 2011, version 5.7.4 (ActiGraph, Pensacola, USA). On day 1, participants completed the OPAQ and the single-item measure recalling the last 7 days (Trial 1). Participants completed both questionnaires a second time 2 weeks later (Trial 2). Participants then wore accelerometers for 7 days and completed the questionnaires for a third time (Trial 3). Trials 1 and 2 were used for reliability and Trial 3 was used to establish concurrent validity. The single item questionnaire was a very short questionnaire that only took 1 - 2 minutes to complete, reported simple information and to limit any ordering effect, was always completed first. The longer measure (OPAQ) then followed and took approximately 10minutes to complete and questioned more detailed recall. Trained research assistants

conducted all assessments, which were completed at the study schools using the same instruments at each time point.

3.3.5 Accelerometers

In adolescents, accelerometers provide good validity and reliability and are often used as the criterion measure for assessing the validity of physical activity questionnaires [66, 243, 253, 254]. ActiGraph accelerometers were used to measure the amount of time participants were involved in moderate to vigorous physical activity (MVPA). MVPA data were derived from the vertical axis only. Accelerometers were dispersed on an adjustable elastic waist belt so that it could be tightly fastened to reduce inaccurate measures [255]. The valid objective accelerometer data were compared to seven days of self-reported data (OPAQ and single-item measure data).

3.3.6 Single-item physical activity measure

A single-item measure that has been found to be reliable and valid in an adult population [12] was modified to match the population studied. It read as follows '*In the past week, on how many days have you done a total of 60 minutes or more of physical activity, which was enough to raise your breathing rate? This may include sport, exercise and brisk walking or cycling for recreation or to get to and from places*'. Participants then ticked the appropriate box ranging from zero to seven days.

3.3.7 Oxford Physical Activity Questionnaire

The OPAQ was selected as a 7 day recall measure as it has shown promise in the adolescent population [15, 79]. Participants are required to complete a table, indicating the time they spent in MVPA for each day of the week and activity type.

3.3.8 Treatment of data

The accelerometer data were downloaded using Actilife 5. An Evenson 15-second cut-point was applied to the data [256]. Missing data were defined as \geq 20 minutes of consecutive zero counts [257]. Data were then cleaned and imported for analysis using IBM SPSS, 2010, version 22 (IBM Company SPSS. Inc.). Currently, there is existing disagreement regarding the required

number of monitoring days required in youth studies [65, 119, 244, 258-260]. A recent review of physical activity studies using accelerometers found that 70% of studies in adolescents used a three to four day wear time criteria to provide an estimate of usual activity [114]. Therefore, to maximize sample size, a three-day inclusion criteria with 480min wear time/day (n = 96) and three days with a 600min wear time/day was applied (n = 72) [114, 261, 262]. The OPAQ data was reported by the participants in MVPA (mins/day). If the activity type listed was a MET count > 3 it was included as self-reported MVPA [7, 263-265]. Daily MVPA (mins) were collaborated to provide a weekly total of time spent in MVPA (mins). Total reported MVPA data (7 days) from the OPAQ and ordinal data from the single-item questionnaire reported in number of days that participants achieved 60mins of MVPA was entered into IBM SPSS for analysis [7, 263].

3.3.9 Analysis

Data were checked for normality based on a visual inspection of histograms and satisfied the criteria. Bivariate Pearson correlations between self-reported MVPA and accelerometer output (time spent involved in MVPA) were used to determine the concurrent validity of the two questionnaires. Reliability was analysed in three ways: 1. Intra-class coefficient (ICC) was used to provide an estimate of repeatability. For ICC calculations, we used consistency agreement, two-way mixed model and reported the single measures estimates; 2. Paired sample t-tests were used to determine change in mean values with between Trial 1 (single-item: mean = 4.29, SD = 1.63; OPAQ mean = 70.61, SD = 52.37) and Trial 2 (single-item: mean = 4.64, SD = 1.64; OPAQ: mean = 74.20, SD = 45.75); 3. Bivariate Pearson correlations between inter-trial difference (T2 - T1) and the mean of the trials [(T2 - T1)/2] were used to identify proportional bias. Bland-Altman plot analysis (Figure 1 and 2) were employed to determine potential bias and limits of agreement by the assessing the total magnitude of time spent in MVPA and the inter-trial difference for each questionnaire. Trial 3 was used for validity (single-item: mean = 4.31, SD = 1.77; OPAQ: mean = 61.44, SD = 46.99). Bivariate Pearson correlations between self-reported MVPA and accelerometer output (time spent involved in MVPA) were also used

to determine the validity of the two questionnaires. Independent sample *t*-tests were also used to reveal if participants excluded from the analysis due not meeting the accelerometer inclusion criteria differed in terms of their self-reported physical activity.

3.4 Results

3.4.1 Reliability

Analysis of ICC revealed the single-item measure (n = 107) demonstrated similar overall reliability (ICC = 0.75, 95% CI: 0.64 - 0.83, p < 0.001) to the OPAQ (n = 104; ICC = 0.79, 95% CI = 0.69 - 0.86, p < 0.001). Paired sample t-tests revealed there was no significant group difference between T1 and T2 for the OPAQ (p = 0.80) or the single-item measure (p = 0.169). Bivariate Pearson correlations between inter-trial difference (T1 - T2) and the mean of the trials [(T2 - T1)/2] revealed the OPAQ had slight proportional bias (r = -0.17, 95% CI = -0.43 - 0.10, p = 0.139) (Figure 1). The results suggested there was no relationship between participants' MVPA inter-trial difference for the single item measure (r = 0.08, 95% CI = -0.12 - 0.26, p =0.465) (Figure 2). Independent samples *t*-tests revealed that there was no significant differences between participants that met the accelerometer inclusion criteria (Single item measure 600MVPAmins/day: mean = 3.96, SD = 0.20 and OPAO 600MVPAmins/day: mean = 65.06, SD = 49.46) and participants that did not meet the accelerometer inclusion criteria (p > 0.05) (single item measure 600MVPAmins/day: mean = 4.80, SD = 0.22 and OPAQ 600MVPAmins/day: mean = 80.63, SD = 56.88). When the accelerometer inclusion criteria was reduced to 480MVPAmins/day, similar trends were found. Although there were no outliers for the single item due to the response format (i.e., 0 to 7), two outliers were removed from the OPAQ data. In terms of missing data, the OPAQ had more missing data at all three time points (6.1%, 17.5%, 45.7% respectively) in comparison to the single-item questionnaire (5.2%, 13.1%, 33.3% respectively). However, it is important to note that the majority of missing data for both measures was due to students being unavailable on assessment days.

3.4.2 Validity

Correlations between accelerometer MVPA and self-reported MVPA were of moderate strength. For the shorter wear-time of 480MVPAmins/day (n = 96), the OPAQ and the single-item both demonstrated similar validity (r = 0.43, 95% CI = 0.23 - 0.62, p < 0.001) and (r = 0.46, 95% CI = 0.24 - 0.63, p < 0.001) respectively. When the accelerometry wear time was increased to 600MVPAmins/day (n =72), the OPAQ (r = 0.50, 95% CI = 0.30 - 0.65, p < 0.001, n = 96) again showed slightly higher validity than the single-item measure (r = 0.44, 95% CI = 0.24 - 0.63, p < 0.001).

3.5 Discussion

The primary aim of this study was to investigate the test-retest reliability and concurrent validity of simple physical activity measures in a sample of adolescents. The three main measures of test-retest reliability, correlation (e.g., ICC and Pearson bivariate correlation), systematic change in the mean (e.g., paired samples t-test) and within-subject random variation (e.g., typical error and limits of agreement), provide evidence for the different types of reliability. For example, paired samples t-tests can be used to determine if there was a significant mean difference between trials (i.e., Trial 1 and Trial 2) However, this measure is based on group means and does not account for individual variability (i.e., there might be no statistical difference between trial means, but there may have been considerable variability among individuals' two scores). Alternatively, within-subject random variation is concerned with random variability of a single individual's values on repeated testing. It has been suggested that within-subject variation is perhaps the most important type of reliability for intervention research because it will influence the precision of change score estimates [266].

Our results revealed that the single-item measure compares favourably with an existing comprehensive physical activity measure (i.e., OPAQ) in both test-retest reliability and concurrent validity against accelerometers. The single-item measure showed high repeatability (ICC = 0.75) and moderate validity (r = 0.44) which is promising, given a recent systematic review of 31 youth physical activity questionnaires found that no questionnaires showed

acceptable both reliability and validity [15]. Previous research into existing adolescent physical activity questionnaires revealed that reliability is often poorly assessed with most studies using Spearman's or Pearson's correlations which are not recommended over ICC [57]. Furthermore, ICC provides evidence of consistency between trials, but does not address the other types of reliability (i.e., typical error or proportional bias) which were addressed in the current study [13]. For reliability, an ICC > 0.70 is considered acceptable (Pearson's correlation or Spearman's rank of 0.80 was also considered to be acceptable) [15]. There is no consensus regarding what is considered to be an acceptable correlation coefficient for physical activity questionnaire validity in young people, but a recent systematic review reported that validity correlation coefficients were in the range of 0.30 - 40 [71]. The single item measure showed equivalent validity to that of existing adolescent physical activity questionnaires.

The Bland-Altman plots and Bivariate Pearson correlations revealed evidence of proportional bias for the OPAQ (r = -0.17). Participants, who reported higher levels of MVPA, were also found to have larger inter-trial differences. This finding may reflect limitations of the OPAQ, but also the large weekly variability in young people's activity patterns, particularly at higher intensities. There was no relationship between participants' MVPA and the inter-trial difference for the single-item measure. Although both the OPAQ and the single-item measure showed high reliability coefficients, the Bland-Altman analysis revealed relatively wide limits of agreement for both measures, indicating that they are more reliable for group physical activity estimates rather than individual estimates.

Accelerometers have been used extensively to validate larger, more detailed questionnaires [243, 261, 262], but evidence for the validity of single-item physical activity measures in youth is minimal [12, 249-251]. In the current study, accelerometer output (time spent involved in MVPA) was used as the reference measure [15]. As this study used two questionnaires of differing lengths, our examination of the single-item measure was strengthened as we could compare it to, not only the accelerometry, but also the OPAQ; a proven valuable tool in adolescent physical activity measurement [15]. The single-item measure was found to be equally valid and reliable as an in-depth questionnaire with regard to the

amount of days that adolescents are involved in 60mins of MVPA, but obviously lacks further detail regarding type, exact duration and intensity of activity. It is possible that shorter questionnaires may result in fewer outliers; inaccuracies and less missing data; particularly among adolescents who have shorter attention spans [80, 81, 267].

The bivariate correlations with accelerometry for 480MVPAmins/day of wear time, demonstrated that both the OPAQ and the single-item measure had similar validity (r = 0.43), and (r = 0.46) respectively. When the accelerometry wear time was increased to 600MVPAmins/day, the OPAQ (r = 0.50) again showed slightly higher validity than the singleitem measure (r = 0.44). Systematic reviews of physical activity questionnaires have reported that validity correlation coefficients were in the range of 0.25 - 0.41 [14] and 0.30 - 40 [71]. In terms of validity, the single-item measure and the OPAQ therefore compare favourably with existing adolescent physical activity questionnaires.

The study differed from the existing literature as it attempted to test the validity and reliability of a single-item measure in an adolescent population. Additional study strengths include the use of accelerometers to validate the measure and the assessment of three types of reliability (i.e., repeatability, change in mean and proportional bias). However, compliance to the accelerometer protocols was lower than desired and the inclusion criterion was subsequently amended to maximize the sample used in the analysis. Secondly, participants' were required to remove the accelerometers for water activities resulting in loss of data [268, 269]. Thirdly, socio-demographic variables may limit the generalizability of findings and the study sample size was relatively homogenous and similar in age. It is possible that younger students may have more difficulty interpreting the OPAQ instructions, which may limit the generalizability of our findings. Fourthly, individual inter-trial differences were not analysed. Fifthly, our failure to randomise the ordering of questionnaire completion is a potential study limitation. However, the single item questionnaire was a very short and was always completed first. The longer measure (OPAQ) then followed that took approximately 10minutes to complete and questioned more detailed recall. Finally, feasibility limitations prevented reliability and validity to be analysed on two separate sample groups.

3.6 Conclusions

The single-item measure was found to have comparable validity and reliability to an existing physical activity questionnaire (OPAQ) that has shown merit in an adolescent population. It also had a higher response rate and less missing data than the OPAQ. The single-item measure is easy to administer and may have utility for screening purposes and for use in population surveys. This measure however, does not provide the detail of more comprehensive questionnaires (i.e. activity type, duration, intensity, time of day/week, daily activity patterns etc.) and may lack the necessary sensitivity to detect change in habitual physical activity. We encourage researchers to compare the utility of different physical activity questionnaires in various populations and against objective physical activity measures. Further research into the appropriate length and detail of physical activity questionnaires will lead to more efficient and accurate data collection.

Characteristics	Males	Females	Total Group	
	n (%) ¹	n (%) ¹	n (%) ¹	
Born in Australia	74 (97.4)	46 (97.9)	120 (97.6)	
English spoken at home	72 (94.7)	46 (97.9)	118 (95.9)	
Ethnicity				
Australian	61 (82.4)	38 (80.9)	99 (81.8)	
Aboriginal/Torres-strait Islander	1 (1.4)	1 (2.1)	2 (1.7)	
European	9 (11.8)	5 (10.6)	14 (11.6)	
Other	3 (3.9)	3 (6.4)	6 (4.8)	
	Mean (SD)	Mean (SD)	Mean (SD)	
Age	14.67 (0.47)	14.70 (0.46)	14.68 (0.47)	
Height (cm)	171.66 (9.52)	165.49 (8.37)	169.27 (9.54)	
Weight (kg)	62.55 (12.26)	56.73 (7.90)	60.35 (11.15)	
OPAQ MVPA (mins/day)	80.22 (41.26)	65.21 (50.61)	74.30 (45.54)	
Single-item measure (days/wk)	4.90 (1.61)	4.26 (1.64) 4.65 (1.64		
Accelerometer MVPA $480 \text{mins}/dav^2$	28.29 (9.85)	19.54 (7.75)	24.74 (9.99)	
Accelerometer MVPA 600mins/day ³	30.46 (10.13)	20.72 (8.49)	26.54 (10.59)	

¹Percentage of participants in study. ²Accelerometer output MVPA mins/day (Inclusion criteria: 3 weekdays, 480mins wear time/day). ³Accelerometer output MVPA mins/day (Inclusion criteria: 3 weekdays, 600mins wear time/day).

Figure 3.1 - Bland-Altman Plot for the OPAQ with 95% confidence limits comparing mean self-reported MVPA and inter-trial difference (trial 1 and trail 2)



Figure 3.2 - Bland-Altman Plot for the single-item measure with 95% confidence limits comparing mean self –reported MVPA and inter-trial difference (trial 1 and trail 2)



CHAPTER 4: ADOLESCENT PEDOMETER PROTOCOLS: EXAMINING REACTIVITY, TAMPERING AND PARTICIPANTS' PERCEPTIONS

Preface: This chapter examines the current adolescent pedometer protocols and focuses on the current issues with objective monitoring that are an inherit threat to reliability and validity. This research is novel as it was first of its kind provide insights into why participants may not comply with objective monitoring protocols by exploring their perceptions with self-report measures.

Citation:

Scott, J. J., Morgan, P. J., Plotnikoff, R. C., Trost, S. G., & Lubans, D. R. (2014). Adolescent pedometer protocols: examining reactivity, tampering and participants' perceptions. *Journal of Sports Sciences*, *32* (2), 183-190.

Published in the Journal of Sport Sciences (accepted 11 June 2013).

4 Adolescent pedometer protocols: examining reactivity, tampering and participants' perceptions

4.1 Abstract

Background: Little is known about reactivity and tampering in adolescent pedometer-based research.

Methods: The sample included adolescents (N = 123, Age range = 14 to 15 years) from three secondary schools in NSW, Australia. Schools were randomized to one of three pedometer protocols: i) Daily sealed pedometer group: steps recorded daily by research team, ii) Unsealed pedometer group: steps recorded daily by participants and iii) Weekly sealed pedometer group: step counts recorded by research team at the end of monitoring period. Participants wore pedometers (Yamax CW700) and accelerometers (ActiGraph GT3X+) simultaneously for seven days. Repeated measures analysis of variance was used to examine potential reactivity. Bivariate correlations between step counts and accelerometer output were calculated to explore potential tampering.

Results: The correlation between accelerometer output and pedometer steps/day was strongest among participants in the weekly sealed group (r = 0.82, $p \le 0.001$), compared to the unsealed (r = 0.63, $p = \le 0.001$) and daily sealed (r = 0.16, p = 0.324) groups. The daily sealed ($p \le$ 0.001) and unsealed (p = 0.003), but not the weekly sealed (p = 0.891), groups showed evidence of reactivity

Conclusion: The results suggest that reactivity and tampering does occur in adolescents and contrary to existing research, and that pedometer protocol may influence participant behaviour.

4.2 Introduction

Although there has been a proliferation in the use of objective measuring devices for measuring physical activity, they are subject to reactivity and potential tampering [9, 100, 159]. Reactivity is a potential threat to the accurate assessment of physical activity and can be defined as a change in normal activity pattern when participants are aware that they are being monitored [141]. Studies examining reactivity in children and adults have produced mixed findings.

Although many studies have failed to identify reactivity [11, 58, 141, 270], others have found a feedback effect resulting in a change of activity usual pattern [9, 100, 159]. For reactivity to be identified, it is expected that participants will exhibit an increase in activity at the start of the monitoring and then return to a more stable pattern once they are accustomed to wearing the devices [270]. Ozdoba and colleagues used sealed and unsealed pedometers with 4th grade children (N = 45) for 14 days and found that over the period, reactivity was not present [58]. Clemes and colleagues examined reactivity in adults and concluded that the reactivity was present but may only last a few days [9, 159, 161], contrary to other findings in young adults [160, 270, 271].

Device tampering (e.g. shaking) and non-adherence to monitoring protocols are additional threats to the accuracy of physical activity assessment using pedometers. Surprisingly, research exploring pedometer tampering is sparse and it is not known if different pedometer monitoring protocols influence participant behaviour. A recent study examining adolescent adherence to a pedometer protocol found that 30 out of the 43 participants self-reported tampering with their pedometers [100]. Adolescents remain a difficult group to measure due to their poor compliance with monitoring protocols [115] and further understanding of their behaviours while being monitored and their perceptions of the monitoring process, may help to improve the accuracy of physical activity assessment in this population.

To the authors' knowledge no previous study has examined reactivity in adolescents and only one study has examined pedometer tampering among adolescents [100]. The current study attempts to compare existing objective protocols to identify the method which provides the most accurate representation of adolescents' physical activity. The primary aim of this study was to investigate adolescents' potential reactivity and tampering while wearing pedometers by comparing different monitoring protocols to accelerometer output. A secondary aim was to explore adolescents' perceptions and experiences relating to wearing pedometers. Such information may help guide monitoring protocols and improve participant compliance.

4.3 Methods

4.3.1 Study design

This study recruited six physical education (PE) classes (approximately 20 students in each class) from three secondary schools in the Hunter Region, NSW, Australia. Schools were randomized to one of three pedometer protocols and all participants wore pedometers (Yamax Digi-Walker CW700, Japan) and accelerometers (ActiGraph GT3X+, Pensacola, USA) simultaneously for seven consecutive days. The study was approved by the relevant institutional review boards and all participants provided written informed consent. The sample included adolescent boys (n = 76) and girls (n = 47) (Mean_{age} = 14.7 ± 0.5). Participants were blinded to the study aims. One way ANOVA's and Chi-square analyses indicated that there were no significant differences in age, height, cultural background or country of birth between groups (p > 0.05) (Table 4.1). Of the 160 dispersed forms, 123 returned consent (i.e., 82% response rate).

4.3.2 Measures

Trained research assistants conducted all assessments, which were completed at the study schools using the same instruments at each time point.

4.3.3 Pedometers

The Yamax Digi-Walker pedometer is recognized as one of the most accurate pedometers in terms of correctly estimating step counts and distance covered [122, 126, 141]. The accuracy of the pedometers was initially assessed by research assistants, using a brief walking test [272]. The Yamax Digi-Walker has high inter-instrument reliability (r = 0.96) when estimating step counts of children in a school setting [121, 174]. This study used the Yamax Digi-Walker CW-700 as it automatically resets itself each day and has the ability to record and store data for seven days.

4.3.4 Accelerometers

ActiGraph (GT3X+) accelerometers were used to measure physical activity level as the reference measure to validate the accuracy of pedometer step counts. Counts per minute (CPM)

data were derived from the vertical axis only. CPM was used in the current study because it is the most widely used physical activity outcome in accelerometer studies as it is not dependent on cut-points or adjusted for wear-time [260]. Accelerometers and pedometers were dispersed on an adjustable elastic waist belt so that it could be tightly fastened to reduce inaccurate measures [255]. The pedometers were clipped onto the same belt, adjacent to the accelerometer.

4.3.5 Pedometry behaviour questionnaire (PBQ)

This questionnaire was developed specifically for this study. The brief questionnaire was composed of 10 questions, each scored on a 5-point Likert-type scale. The first five questions focused on behaviours that the students exhibited whilst wearing pedometers, with responses ranging from 1 = never to 5 = every day. i) 'I opened my pedometer even though it was sealed ii) 'I wore my pedometer throughout the day, other than swimming/showering', iii) 'When I wore a pedometer I was more active than normal', iv) 'I shook my pedometer to increase my step count' and iv) 'I purposely wrote down more steps than I actually did' (if applicable). The next five questions explored students' perceptions of wearing pedometers with responses ranging from 1 = strongly disagree to 5 = strongly agree. i) 'I wanted to get a high step count to impress the researchers' ii) 'I wanted to get a high step count to impress my peers' iii) 'I found that wearing a pedometer was uncomfortable iv) 'I found that wearing a pedometer was a waste of time. The questionnaire was administered on the last day of the measuring phase (day 8).

4.3.6 Procedures

Data were collected in the fourth school term of 2011 (i.e. October to December). Prior to measuring, schools were randomized to one of the three pedometer protocols. Face-to-face contact with the participants occurred in their normal PE lessons. On the first day of the measuring period, a member of the research team explained the general purposes and functions of pedometers and accelerometers. Participants were blinded to the study aims and were told that the purpose of the study was to provide an estimate of their 'usual' physical activity [252].

A member of the research team explained the importance of correct placement and the method of recording step counts if relevant, which was dependent on their designated protocol:

- 1. *Daily sealed group:* Participants in this group wore a pedometer that was sealed with a sticker. At the start of each school day they took their pedometer to an allocated classroom and met a researcher who removed the small sticker, recorded the daily step count and attached a new sticker to the pedometer. Participants were permitted to see their recorded count. Step counts for the weekend were recorded on the Monday morning.
- Unsealed group: Participants in this group wore an unsealed pedometer for seven days. At the end of each day, participants recorded their steps on a personal log sheet.
- 3. *Weekly sealed group:* Participants in this group wore pedometers sealed with a sticker for seven days so that the step count could not be seen by the participant for the monitoring period. Daily step counts were recorded by the researchers when pedometers were re-collected at the school after seven days.

On the morning of the first measuring day, pedometers were checked by researchers that they were functioning properly. All participants were given an activity belt (pedometer and accelerometer) and a log folder (to log step counts and removal of activity belt). As part of consent, participants were required to provide a personal mobile phone number. A large majority of the participants had their own personal mobiles (87.3%). The remaining 16.3% provided a parental/guardian contact number. Each morning participants were sent a brief text message to remind them to put on their activity belt each day and also ways to correctly log their step count (if appropriate).

On the seventh day of measuring, students were sent an additional text message reminding them to return their measuring devices the following day. To encourage participants to bring back activity belts on the collection day, gift bags (containing a drink bottle and stickers) were provided on return of undamaged pedometers and accelerometers. When

participants returned their activity belts on Day 8, they then completed the Pedometry Behaviour Questionnaire.

4.3.7 Treatment of Data

Currently there is no consensus regarding the treatment of extreme high or low pedometer scores for adolescents. Based on the previous studies, we applied inclusion cut-points of < 1000 and \geq 30,000 steps/day, which were originally developed for children [102, 144, 272]. The replacement of missing pedometer data followed the procedures Kang and Rowe's previous studies [144, 273]. Data were then cleaned and merged, for analysis using IBM SPSS, 2010, version 19 (IBM Company SPSS. Inc.). From the 96 participants included in the analysis, there were nine scores of over 30,000 steps (1.2%) and 43 scores of less than 1000 steps (5.8%) These extreme values were deleted and replaced with the appropriate weekday or weekend personal mean. Days with no activity were excluded. Non-wear time was determined by strings of consecutive 0 counts \geq 20minutes. Participants were included in data replacement if they had at least three valid days of data. Neither pedometer nor accelerometer data were adjusted in terms of self-reported non-ambulatory and water activity.

The accelerometer data were downloaded using Actilife 5, 2011, version 5.7.4 (ActiGraph, Pensacola). A 15-second epoch length was applied to the data. Activity counts/minute (CPM) were then calculated and imported into SPSS for analysis. While four days of monitoring is recommended as the minimum [119], there is a growing recognition for a more flexible framework with researchers selecting alternative thresholds depending on their objectives and the population studied [258]. Because the current study was focused on comparing different protocols rather than estimating habitual physical activity, a three-day inclusion criteria with 480min wear time/day was applied to maximize the sample size [n = 96 (not inclusive of a weekend day) and n = 54 (inclusive of a weekend day)].

Analysis

Data were checked for normality and satisfied the criteria. Alpha levels were set at p < 0.05. The data file was split by group (i.e., pedometer protocol) and repeated measures analysis of variance (ANOVA) was used to explore potential reactivity during weekdays. To explore potential pedometer tampering (e.g. shaking), bivariate correlations between mean step counts and mean CPM were calculated. We hypothesized that participants in the unsealed groups would shake their pedometers, resulting in a weaker correlation between step counts and accelerometer CPM. One way analysis of variance with Bonferroni post-hoc procedures were used to compare participants' perceptions of, and behaviours while wearing monitoring devices between groups. Independent samples t-tests were used to explore gender differences.

4.4 **Results**

4.4.1 Reactivity

Both the daily sealed and unsealed groups showed evidence of reactivity (initial increase of physical activity followed by levelling off) over the monitoring period (Table 4.2). The decline of physical activity in the daily sealed group occurred between Day 3 and Day 4 ($p \le 0.001$). This decline occurred earlier in the unsealed group, between days 2 and 3 ($p \le 0.001$). The weekly sealed group showed no significant differences over the monitoring period (p = 0.886).

4.4.2 Tampering

To examine tampering of pedometers, correlations between accelerometer output (mean CPM) and mean steps/day were calculated (Table 4.3). Correlations between weekday CPM and pedometer steps/day were strongest among participants in the weekly sealed group (r = 0.82, $p \le 0.001$), compared to the unsealed (r = 0.63, $p \le 0.001$) and daily sealed (r = 0.16, p = 0.324) groups. Correlations in the weekly sealed group were again strongest with the inclusion of a weekend day (r = 0.81, $p \le 0.001$).

4.4.3 Self-reported behaviours and perceptions

In total 111 participants satisfactorily completed the PBQ (90.2%). There were no significant gender differences in participants' self-reported behaviours and perceptions (p > 0.05). Participants in the weekly sealed group were less likely to report a '*desire to impress the research team*' with their step counts than those in the daily sealed (p = 0.021) and unsealed (p

= 0.003) groups. Participants in the daily sealed group reported that they did more activity than normal whist being measured (p = 0.010). For all other responses, there were no significant differences between groups (p > 0.05). Of the participants that completed the questionnaire, 49% reported shaking their pedometers to increase their step count. In regards to participants' perceptions of the monitoring process, 40% indicated that they did not like wearing pedometers, 81% found wearing a pedometer uncomfortable and 69% reporting that they found wearing a pedometer embarrassing at times.

4.5 Discussion

The primary aim of this study was to investigate adolescents' reactivity and tampering while wearing pedometers by comparing different monitoring protocols to accelerometer output. Our findings suggest that reactivity and tampering occur in adolescents and the protocol selected for pedometer monitoring impacts on behaviour and compliance. Our secondary aim was to explore adolescents' perceptions and experiences whilst wearing pedometers. Based on our findings, it appears that adolescents find objective measuring devices uncomfortable and embarrassing and a large proportion report a desire to impress their peers and/or researchers with elevated step counts.

In previous studies, ICC and repeated measures ANOVA have typically been used to explore reactivity to pedometer monitoring. The assumption is that participants will increase their activity at the start of the monitoring period and then return to a more stable pattern once they are accustomed to wearing the devices [270]. In the current study, we found evidence of reactivity among students in the weekly unsealed group, suggesting they increased their activity at the start of the study period. However, studies have found substantial intra-individual variation in objectively measured physical activity in youth [274]. Consequently, changes in physical activity over the week due to sports training or physical education might be misclassified as reactivity. These findings highlight the need for more rigorous methodologies examining reactivity in different populations. To explore the possibility of pedometer tampering, adolescents were required to wear accelerometers and pedometers simultaneously for seven days. Previous studies have used accelerometers to validate pedometer step counts [91, 275]. A recent study of free-living adolescents found moderate to strong correlations (r = 0.65 - 0.74) between pedometer step counts and CPM [276], but no previous study has used this method to compare different pedometer monitoring protocols in youth. Based on our previous trials and anecdotal evidence [183, 193], we hypothesized that unsealed group would provide the least accurate pedometer values. The relationship between CPM and steps was non-significant in the daily sealed group and this group was also significantly more likely to report a desire to impress the research team. Whilst non-significant (p = 0.33), the trend showed that those in the daily sealed group participants were more likely to shake their pedometers (DS mean = 2.05, SD = 1.21; US mean = 1.68, SD = 1.02; WS mean = 1.77, SD = 1.17). These findings suggest that adolescents who have daily contact with members of a research team as part of the monitoring process are more likely to exhibit reactivity, even if the pedometers are sealed.

As hypothesized, there was evidence of reactivity in the unsealed group, yet step counts remained relatively stable across the monitoring period in the weekly sealed group. This may be due to the sealing of pedometers and the removal of step count feedback. Although a pattern of reactivity was evident in the daily sealed and unsealed groups, we cannot be certain that this was reactivity and not a 'day to day' difference in physical activity, although this seems unlikely as the weekly sealed groups showed no significant differences across days. The unsealed groups were required to log their own step counts which resulted in the most missing pedometry data of all groups. Correlations of step counts and CPM were strongest in the weekly sealed group implying that this was the most reliable protocol of the three and reduced pedometer tampering. Correlation was weakest in the daily sealed group (r = 0.16) indicating higher tampering most likely due to daily contact with researchers. This was reinforced when the inclusion of a weekend day was added to the analysis and the correlation was raised to be similar to that of the unsealed group (r = 0.54).

Responses from the questionnaire revealed that almost half of students that completed the questionnaire admitted to tampering with their pedometer with no statistical difference between boys and girls. Furthermore, a third of all participants reported that they did more physical activity than normal while wearing the pedometers. Few studies have examined participants' behaviours and perceptions of the monitoring process and an improved understanding of the measurement barriers may improve the accuracy of pedometer protocols. These findings suggest that reactivity and tampering are threats to the accuracy of objective physical activity measurement in adolescents.

Bulk internet text messaging proved to be a very useful strategy to remind participants to wear measuring devices log step counts and to return activity devices at the end of the measuring phase. Instead of giving the students a log sheet, they were provided a log 'folder' and encouraged to personalize it. Participants also chose various stickers to put on the top of their measuring devices to ensure that the devices were not worn upside down to reinforce accurate measurement.

The strengths of this study include the simultaneous assessment of physical activity using accelerometers and pedometers and the triangulation of findings using questionnaires. Despite these strengths, there are some limitations that should be noted. First, compliance to accelerometer protocols was lower than anticipated (78%) and we subsequently adapted our wear-time criteria to include as many subjects as possible in the analyses, a longer monitoring time frame may have more accurately detected reductions in step counts (i.e. reactivity). Second, pedometers were sealed with stickers and although, researchers could tell if the pedometers had been tampered with more stringent sealing measures may be necessary to prevent pedometers from being opened. We encourage manufacturers to produce lockable pedometers for research purposes. Thirdly, we did not assess participants' behaviour with regard to the accelerometer. However, this was not part of the research design. Finally, we did not use a cross-over design which would have provided us with an opportunity to assess the adolescents using each of the three monitoring protocols (although cross-over designs have their own inherent weaknesses).

4.6 Conclusions and Recommendations

Contrary to existing research [58, 141, 144, 270], our results suggest that the protocol selected for pedometer monitoring impacts on behaviour and compliance. With regard to adolescent behaviour, half of the students self-reported tampering; 40% state that they didn't like the process; 81% found it uncomfortable and 69% found it embarrassing. Clearly, more qualitative research is warranted in this area to learn more about the adolescent population and their perceptions of monitoring processes. Weekly sealed pedometer step counts were found to have the strongest association with accelerometer CPM and there was no evidence of reactivity in this group. Based on our findings, we recommend seven days of sealed pedometer monitoring using pedometers capable of storing at least 7 days of step count data in their internal memory. While longer monitoring periods may reduce reactivity, this may not be feasible among adolescents who often feel burdened by the monitoring process. Further research on pedometer protocol and the development of guidelines to improve the accuracy of adolescent pedometer protocols is clearly warranted.

Table 4.1: Characteristics of sample across three groups

Pedometer Protocol	Daily sealed		Unsealed		Weekly sealed	
Number of participant in sample	n = 46		n = 44		n = 33	
Number of participants meeting inclusion criteria (480mins, 3 valid days of wear time).	n = 39		n = 31		n = 26	
	Mean	SD	Mean	SD	Mean	SD
Age (years)	14.7	0.44	14.6	0.48	14.7	.49
Country of birth ¹	46	100%	42	95%	32	97%
Language spoken at home ²	43	94%	42	95%	33	100%
Ethnicity						
Australian	37	80%	33	75%	29	88%
Aboriginal/Torres-strait Islander	-	-	2	5%	-	-
European	7	15%	5	12%	2	6%
Other	-	-	4	9%	2	6%
Height (cm)	171	8.80	168	10.71	169	8.49
Weight (kg)	59	8.48	64	13.40	59	10.26

¹Participants born in Australia.

²Participants who speak English at home.

Pedometer protocol	Day 1	Day 2	Day 3	Day 4	F	<i>p</i> - value	n
	Mean (SD)	Mean (SD)	Mean (SD)	Mean (SD)			
Daily sealed					7.587	≤ 0.001	39
	11486	10819	12041	8935			
	(4991)	(5345)	(5818)	(3986)			
Unsealed	10213	11244	8387	9820	4.991	0.003	31
	(3684)	(5345)	(3917)	(4232)			
Weekly sealed	8715	8189	8487	8590	0.215	0.886	26
	(3321)	(3888)	(8487)	(5205)			

 Table 4.2: Changes in daily step counts and reactivity by pedometer group

Note. Repeated measures analysis of variance used to test for reactivity.

			Pedometer protoc	ol
		Daily sealed	Unsealed	Weekly sealed
Step/day ¹		10833 (3682)	9915 (3388)	8495 (3380)
CPM ¹		554 (145)	426 (133)	393 (117)
Weekdays only	n	39	30	26
	<i>p</i> -value	0.342	\leq 0.001	≤ 0.001
	r	0.156	0.632	0.822
Including weekend day	n	18	20	15
	<i>p</i> -value	0.021	0.021	\leq 0.001
	r	0.540	0.514	0.810

Table 4.3: Relationship between mean steps/day and accelerometer counts per minute

Note. Bivariate correlation used.

¹Means (standard deviation) reported

CHAPTER 5: YOUNG PEOPLES' PERCEPTIONS OF THE OBJECTIVE PHYSICAL ACTIVITY MONITORING PROCESS: A QUALITATIVE EXPLORATION

Preface: This chapter examines adolescents' perceptions of objective physical activity monitoring. This builds on the finding of the previous chapter by providing a more in-depth look at adolescents' compliance to physical activity protocols. This research is novel as it was first of its kind provide insights into why participants may not comply with objective monitoring protocols by exploring the participants' perceptions through focus groups.

Citation:

Scott, J. J., Hansen, V. Morgan, P. J., Plotnikoff, R. C., & Lubans, D. R (2017). Young Peoples' Perceptions of the Objective Physical Activity Monitoring Process: A Qualitative Exploration, *Health Education Journal*, doi: 10.1177/0017896917734576

Published in the Health Education Journal (Accepted 09 September 2017)
5 Young Peoples' perceptions of the objective physical activity monitoring process: a qualitative exploration

5.1 Abstract

Objective: To explore young peoples' perceptions of pedometers and investigate behaviours exhibited whilst being monitored.

Design: Qualitative design using six focus groups with participants (mean age 14.7).

Setting: Study participants (n= 24) were randomly selected from a previous study of 123 young people age 14-15 years, from three secondary schools in New South Wales (NSW), Australia.

Methods: Participants wore pedometers (Yamax CW700) and accelerometers (ActiGraph GT3X+) simultaneously for seven days. Accelerometer output was used to categorize participants into one of six focus groups (three boys groups, 3 girls groups) i) low active (<30 mins MVPA/day), ii) Medium active (30-60mins MVPA/day), iii) High active (≥60 mins MVPA/day). Participants were questioned on their perceptions of the monitoring process and the behaviors' that they exhibited whilst wearing pedometers. A hybrid approach to data analysis identified key concepts which were thematically analysed.

Results: The two main themes observed were: i) participants' perceptions of the monitoring process and, ii) behaviour exhibited while being monitoring. Overall, participants' attitudes towards objective monitoring were positive. A large proportion reported changing their levels of physical activity during the monitoring process, and 87.5% of the focus group participants reported shaking their pedometers to increase their step counts. The medium and high active groups reported changing their activity patterns more than the low active groups.

Conclusion: Our findings are consistent with previous quantitative studies that suggest reactivity and tampering are commonplace among young people. Pedometers may have more utility as an intervention strategy for increasing activity rather than a method for assessing habitual activity levels.

Key Words: Focus Group, Pedometers, Young People, Behaviours, Physical Activity

5.2 Background

Pedometers are regularly used to measure habitual physical activity, assist in identifying the determinants of physical activity and to evaluate physical activity interventions [10, 93, 247]. Despite their ubiquity, little is known regarding young peoples' perceptions of, and behaviours during, pedometer-based research [100, 277]. Despite the challenges of ensuring young peoples' adherence to objective physical activity monitoring protocols [222, 278], there is a paucity of research examining their perceptions of the monitoring process [61]. As objective monitoring is regarded as the gold standard for measuring physical activity in population level studies [65], there is a clear need to improve our understanding of the monitoring process in young people.

Current research on pedometer protocols has identified 'reactivity' and 'tampering' across different population groups of children, young people and adults [9, 141, 172]. '*Reactivity*' is defined as a change in normal activity patterns when the participants are aware their physical activity level is being monitored [141]. '*Tampering*' is the term generally used when participants purposefully shake their pedometers to increase their step count; this is also sometimes referred to as 'cheating' [100]. Reactivity and tampering are an inherent threat to accurate physical activity measurement [172]. Kahan and colleagues studied adolescent adherence to pedometer protocols and found that 30 out of the 43 participants self-reported tampering with their pedometers during the monitoring process [100]. With growing quantitative research to support the existence of reactivity and tampering among young people, qualitative research may help improve our understanding of why participants feel the need to change their activity patterns during monitoring.

A recent qualitative study by Kirby and colleagues (n = 35) concluded that children's and young peoples' attitudes toward accelerometers were mixed [61]. Whilst some reported positive experiences, such as feeling 'special' and attaining increased attention, others reported that they perceived they were at higher risk of being bullied whilst wearing the devices and that they were uncomfortable. A further study focusing on the utility of accelerometers in child obesity interventions used parental interviews (n= 12) to collect information on participants' perceptions of the monitoring process and reported similar reasons for non-adherence including comfort issues and higher risk of stigmatisation [279]. As young peoples' adherence to monitoring protocols is such a complex phenomenon, further research in this area has been recommended [61, 100, 115].

Most pedometers provide direct feedback of activity level in an easily understood 'stepcount'; previous research has shown that this feedback effect can enhance the amount of reactivity and tampering [61, 100, 141, 159]. However, no previous study has employed focus groups to investigate young peoples' perceptions of pedometers, which due to device feedback, affect young peoples' adherence to protocol differently to that of accelerometers [172, 277]. As pedometers are still being used in physical activity research to collect data on population groups and evaluate interventions, further research into the amount of reactivity and tampering and its effect on validity is warranted. To provide deeper analysis of issues surrounding objective physical activity monitoring, rich qualitative data derived from focus groups may provide useful and important insights. This will assist researchers in their attempts to accurately measure physical activity.

Numerous studies have demonstrated that young boys engage in more physical activity than girls [280-282]. Recent research has suggested that social relationships, environmental factors and biological differences may be responsible for the sex-based differences in physical activity levels [283]. To the authors' knowledge, no previous studies have investigated if participants' perceptions of objective monitoring process differ according to sex and physical activity levels. Further research is warranted to address these gaps in the literature.

The primary aim of this study was to explore young peoples' perceptions of pedometers and investigate the physical activity behaviours exhibited by participants during the monitoring process. A secondary aim was to investigate if participants' perceptions differed depending on physical activity level.

5.3 Methods

5.3.1 Study design and procedures

The focus group sample (n = 24) for this study was randomly selected from a previous study that investigated different pedometer protocols, which used a convenience sample of 123 young people aged 14-15 years, from three secondary schools in New South Wales (NSW), Australia. The 123 participants were divided into three pedometer protocol groups: i) daily sealed group: wore sealed pedometers and step counts were checked by researchers daily, ii) weekly sealed group: wore sealed pedometers and step counts were checked by researchers at the end of the week, iii) unsealed group: wore unsealed pedometers and step counts were checked by researchers at the end of the week, iii) unsealed group: wore unsealed pedometers and step counts were checked by researchers at the end of the week. Regardless of the pedometer protocol, all participants were told not to look at or shake their pedometer, and also not to change their normal activity pattern [172].

Participants wore pedometers (Yamax Digi-Walker CW700, Kumamoto City, Japan) and accelerometers (ActiGraph GT3X+, Pensacola, USA) simultaneously for seven days. Contact with participants was made in their Physical Education (PE) classes. Participants were blinded to the study aims and were told that the researchers were attempting to gain an estimate of their normal activity pattern. They were also told that the researchers would be the only people who would see their step counts and it was confirmed with the students that their activity level (step counts) would not be compared across individuals or classes. Accelerometers were worn solely to categorize participants into focus groups based on accelerometer-determined moderate to vigorous physical activity (MVPA).

5.3.2 Measures

5.3.3 *Pedometry/accelerometry*

The pedometers were pre-set with date and time to record a 7-day period. Yamax CW700 pedometers were chosen, as they are able to record steps over a period of seven days. The pedometer has an in-built 24-hour reset and records the step count daily. The accelerometers were preset to collect data at 15second epochs. After the monitoring period, the data were

downloaded using ActiLife5 software version 6.5.3 (ActiGraph, Pensacola, USA). Evenson cutpoints were then applied to obtain accelerometer-determined time spent in MVPA [284].

5.3.4 Focus Groups

A qualitative design using a focus group methodology was applied due to the likely benefits of group interaction which may elicit information and insights that may be less accessible during individual interviews [285]. This benefit of focus groups may be particularly true of groups in which members possess some level of group affinity and connection, as was the case in the current study [285]. Additionally, a focus group methodology is considered particularly appropriate for use with children and young people, and considered the method of choice when there is a concern of socially desirable responding [286].

All participants from the larger sample (n = 123) provided written assent and parental consent to be involved in the study and focus groups if were selected after the monitoring phase. After the seven days of objective monitoring, six focus groups, each consisting of four participants were conducted to determine participants' perceptions of the monitoring process. Participants in all of three pedometer groups were given the same instructions. More specifically, participants were told not to open their pedometer and or look at their step counts. Random selection was stratified to have equal number of boys and girls. Smaller groups comprising of 4 to 6 provide an ideal environment where the participants for each focus. Data saturation was achieved (n = 4). None of the participants that were selected to participant in the focus groups declined to be involved.

Participants were categorised as follows (each having one boy and one girl group):

- Low-active groups: participants in these two focus groups achieved an average of <30 minutes/day in MVPA over their monitoring period.
- Medium-active groups: participants in these two focus groups achieved an average of 30-60 minutes/day in MVPA over their monitoring period.

High-active groups: participants in these two focus groups achieved an average of ≥60
minutes/day in MVPA over their monitoring period.

The trained researcher who administered the focus groups was not associated with the involved schools and had no prior relationships with the students (participants). On average, the focus groups ran for between 20-25minutes. Interviews were conducted following the development of questions designed by the research team to facilitate discussion and reflection of the overall aims of the qualitative study component. The focus group questions were developed and peer-reviewed by three Professors with extensive experience physical activity research. Of particular relevance to this project, the research team are familiar with the challenges of measuring physical activity in children and young people using objective monitoring devices.

The questions were specifically designed to explore the participants' general perceptions of wearing pedometers, perceived need to, and reasons for, altering their step count, the perceived honesty of others, as well as views on the usefulness of assessing physical activity using pedometers. Further questions explored a range of behaviours relating to pedometer use, such as reasons for removing, opening, shaking or lending out the pedometer, as well as changes to physical activity levels during the monitoring period. Participant honesty and prevention of socially desirable responding was promoted by several means including an overt statement of facilitator independence as well as through iterative questioning. Focus group participants were told that they would not get into 'trouble' for providing honest answers, and that when the responses were transcribed all names would be removed and replaced with ID numbers so no individuals could be identified. Prior to conducting the study, focus group questions and study design proceeded through rigorous internal peer-review and the university and educational institutional ethics approval process.

5.3.5 Analysis

The focus groups were digitally recorded with the participants' consent and transcribed verbatim. To classify each individual within each focus groups, each participant was allocated

an ID number. The first letter represented the level of activity (L = low, M = medium and H= high); the second letter identified sex (B = boy, G = girl) and third, a number between one and four (e.g., LB1). A computer program (NVIVO 10, Burlington, USA) was used to assist with the organisational aspects of data analysis. Analysis was conducted by a qualitative researcher, who was independent to the main study. To ensure reliability, a second researcher then conducted separate thematic analysis for categorisation of themes. Intercoder reliability analysis revealed a good level of agreement the two coders (k =0.72, p <0.001) [289].

A hybrid approach to data analysis was adopted [290, 291] allowing for identification and refinement of the key concepts which had been developed as part of the study framework, as well as identification of concepts arising from in vivo coding. Initially, codes or labels were formulated which were firmly grounded in the overall study aims, with the purpose being the development of a taxonomy which would enable a detailed description of the domains that characterised the multifaceted attitudes, perceptions and experiences of participants. The developing coding scheme was revised and further expanded during the initial open coding process. Following coding of all focus group transcripts, the key domains, and the dimensions that emerged within these, were thematically analysed and described in detail.

5.4 Results

The focus groups participants had similar characteristics to the main study sample in terms of sex, age, physical activity level. The participants self-reported characteristic data were imported and analyzed using IBM Statistics (SPSS 12 Inc. Chicago, USA) software with alpha levels set at (p < 0.05). One way ANOVA's and Chi-square analyses indicated that there were no significant differences in self-reported height and weight, or body mass index (BMI) (p > 0.05), (see Table 1). The qualitative analysis yielded a detailed and descriptive account of a range of domains surrounding the participants' attitudes towards, and behaviours relating to, pedometer use.

5.4.1 Attitudes relating to pedometers and their usefulness

5.4.2 General perception of wearing pedometers

Overall, pedometer use was well received among participants, with no distinct differences between participant groups, with the exception of some girls of varying activity levels, who were bothered by wearing them. Generally, participants had taken an interest in wearing the pedometers from the perspective of self-monitoring. While close to half of the participants had been somewhat annoyed by wearing them initially (i.e., bulky, rubbing, uncomfortable), almost all reported having quickly become accustomed to it, with the pedometer becoming part of daily routine;

"...I noticed the first day, then after that it was just a daily routine put it on, take it off... go for a swim, take it off beforehand, put it back on afterwards. It was pretty much normal day except for putting it on." (LB4)

"I found it really annoying, it's like this massive bulge just hanging... out of the side of you (LG3)"

5.4.3 Perceived need to impress others

Apart from one low active girl, who acknowledged occasionally feeling the need to impress the researchers by exhibiting an increased step count; the other participants across the focus groups did not report a perceived need to increase their step counts to impress either teachers or researchers. However, a number of participants in the high active boys and high active girls' groups as well as medium active girls group, admitted to having engaged in some level of peer competition to achieve higher than normal step counts to impress their peers.

"We just all competed for who had the highest each day ... we were just sort of all trying to get the best." (HG2)

A number of participants alluded to the importance of self-monitoring their step count in determining whether they would feel the need to "catch up [to their peers]" by engaging in more physical activity or artificially increasing their step count;

"I looked at mine like all the time and especially ...if I wanted to see who was doing the best with our steps we'd check it...[I would increase my steps by] walking and then if I was losing, I'd probably stand out there and jump because it like skips a couple of times." (MG2)

Additionally, four boys and girls across all activity levels reported they felt a need to engage in some form of informal self-competition:

"...I felt like impressing myself, I wanted to impress myself with the amount of steps I could get." (MG1)

5.4.4 Perceived dishonesty amongst others

Analysis revealed no clear differences between sexes or activity levels, the vast majority of participants perceived most of their peers to have engaged in some level of dishonest behaviour, with the only anomaly being a more positive appraisal of other participants' honesty given by the high active boys group. In this group, there was a predominant belief that a majority would have refrained from artificially increasing their step count. In the other groups, the perceived level of dishonesty among other members of the study population ranged from 'half' to 'most', with dishonest behaviour seen mainly to have manifested itself as engaging in more sport than usual, pedometer shaking, deliberate non-purposeful physical activity such as walking on the spot, jumping up and down etc. with the commonly perceived notion of this being largely due to peer competition.

"I think it's half and half, some people would try and do more exercise... I know for a fact a couple of my friends were shaking it or doing a lot more exercise than they normally would". (LB4)

A couple of female participants alluded to dishonesty being associated with some level of guilt, while honesty on the contrary was promoted by fear of negative consequences (i.e., getting into trouble);

"...I think some of them ...did it right, they didn't shake it or anything but then the other half did shake it because...they wanted to get higher [steps] ...when I shook it...I felt bad, I wanted you [the researcher], to get the right information." (MG2)

Some participants had perceived everyone to have changed their behaviour in some way, with a few questioning whether assessment of physical activity levels is at all possible without somehow affecting the behaviour being measured;

"Everyone would probably change because you can't act the same when you're being measured as when you're not, you kind of feel like you need to impress someone or be more active just so you're not the...laziest kid." (HB1)

5.4.5 Perceived usefulness of assessing physical activity with pedometers

While the participants acknowledged that the majority of pedometer results may not represent an accurate picture of participants 'usual' or 'normal' activity levels due to dishonest behaviours, most participants felt that the use of pedometers still afforded a meaningful measure of physical activity levels; which by some in particular were seen as useful for raising peoples' awareness of their own activity level, for assessing within-person changes over time, or for exploring differences between groups;

"No, I don't think they're a waste of time because it shows how many steps people take every day and how active we are...for the people that aren't cheating, they're actually doing it properly, it's good for them." (LB4)

However, while approximately a quarter of participants did not perceive pedometers to be a useful measurement tool due to the high prevalence of dishonest behaviour, another quarter felt that their usefulness was solely dependent on the honesty of the group.

5.4.6 Behaviours relating to pedometer use

5.4.7 Reasons for removing pedometers

The participants provided a range of reasons for removing the pedometers during the study phase. Apart from the logistical/practical ones of showering, sleeping, water activities and contact sports, some participants had (or knew of others who had) removed them due to discomfort;

"Some people... they just didn't want to wear it, ... one person said that they wore it two days of the whole week so some people just didn't want to wear it at all." (LG2)

A small number of girls (across activity groups) identified social reasons (embarrassment) for temporarily removing their pedometers. One medium active boy reported removing his pedometer due to his dislike of being measured.

"I felt embarrassed [wearing the pedometer], it felt like everyone was seeing me wearing it (MG3)"

Apart from four female participants, who admitted to letting others wear their pedometer (mainly physically active friends or siblings) to increase their step count, the remainder reported not having lent out the pedometers at any stage with the main reason being that it would not show their own results and hence have both moral implications as well as negatively impact the validity (and interest-value) of the results;

"I wanted to get my accurate reading and I wanted to see how fit I am on a daily basis so I chose not to give it to anyone else." (LG3)

5.4.8 Tampering with pedometers

Even though all participants involved in the pedometer monitoring were told not to open or shake their pedometer, 87.5% of focus group participants reported shaking their pedometers to increase their step count. Only three participants reported not having intentionally opened their pedometers. The remainder admitted to having opened the unit (ranging from only 'once' to

'often'). The main reason given was simply being curious as to how many steps they had done, while others alluded to peer interest, peer competition or peer pressure as having contributed to them opening their units to check step counts;

"Yeah I did [open it] because I just wanted to see the step count and all my friends would be like, what are you are up to now? then they'll ask 10 minutes later, what are you up to now?, ...so I did." (HG3)

"Yeah, I opened it...I was curious to check my step count and I was quite proud of it actually...not for opening it because that was rebellion...and I kind of did feel pressured because other people...all my friends they're like "Oh, open it" so I was just like, do it." (LG3)

5.4.9 Changes in physical activity levels while wearing pedometers

More than half of the participants, especially within the high active groups (both sexes) and medium active girls group, revealed having engaged in more physical activity than normal during the measuring phase. Amongst those few participants who did offer an explanation, some admitted to having been motivated by peer competition. However, it appeared that the increased efforts mainly were a result of knowing their activity levels were being measured.

"Yeah... we just wanted to get [our steps] up and because... you want it to be high, then higher again...you just want it to go up every day." (HG3).

"I'd tried to stay on longer for touch footy or bowl more in cricket [or] something like that, to just keep my steps up...because I was getting measured, I guess" (HB1)

None of the low active boys taking part in the focus group reported changing their normal activity levels. The reasons given for not doing so included having forgotten that they were wearing the pedometer and not feeling the need to increase their activity level. All other participants admitted to changing their normal activity pattern in some way. While this mostly comprised increased activity during sports for the high active boys, the remainder of the groups of both sexes mostly reported increased incidental and low intensity intentional physical activity such as walking (n = 11); more general physical activity such as running and jumping (n = 4).

"...I would help more around the house and walk back and forth to do things... usually I would stay home but now I go out and stuff like that." (MG1)

5.5 Discussion:

The aim of this study was to explore young peoples' perceptions of pedometers and gain insight into the way young people behave while wearing pedometers. An additional aim was to investigate if participants' perceptions differed depending on their physical activity level. Responses from the focus groups indicated that the majority of participants' enjoyed wearing pedometers. However, some found them uncomfortable and embarrassing to wear which affected their level of adherence. Participants in the medium and high active focus groups reported changing their activity patterns more than the low active groups. The study findings not only support this existence of reactivity and tampering, but also provide future insight into why young people subconsciously or consciously change their activity patterns when being measured.

Missing data due to non-adherence to monitoring protocols is a widespread limitation of objective physical activity measurement [87, 144]. Participants were therefore questioned as to why they removed the pedometers. Most responses were expected (i.e. water activities, showering, contact sports.); other common responses were that it was 'embarrassing, annoying and/or uncomfortable'. This shows the need for more comfortable and less inhibiting devices to improve adherence in young populations. Emerging research suggests that participants consider wrist-worn devices to be more comfortable, less embarrassing and burdensome to wear than hip-worn devices [292-294]. Wrist-worn objective monitoring could address some of the issues that arose in the focus groups in relation to size, comfort and confidence in how they are perceived by their peers. However, it should be noted that some participants enjoyed wearing

Chapter 5: Adolescents' perceptions of pedometer monitoring

the pedometers, with some even reporting that they forgot that they were wearing it once they had it on.

Overall, participants' attitudes towards objective monitoring were positive. Most believed that pedometers are a useful tool for measuring young peoples' physical activity patterns. A large proportion of the participants reported purposely changing their normal activity and stated that they believed most of their peers did the same. The majority reported having done so through increasing incidental physical activity (e.g., walking on the spot while watching TV) and planned physical activity (e.g., going for a walk or jog), as well as increased effort and time in usual sports participation (e.g., running harder than they usually would during a game).

Most of the participants in the low active groups (both boys and girls) reported that they did not attempt to increase their step count during the monitoring period. The reported reasons for not changing their activity pattern ranged from having forgotten about the pedometer to not perceiving a need to increase their activity levels. It is has been reported that low active participants are less compliant with physical activity intervention protocols due to fear of the way they will be perceived by the researcher or peers [279]. On the contrary, our research suggests that the low active participants were less likely to change their activity patterns than the medium and high active participants.

Pedometer tampering has been identified as a threat to the accurate measurement of physical activity in young people [100, 172]. This study revealed that 21 out of the 24 (87.5%) participants in the focus groups reported shaking their pedometers to increase their step counts. As a high proportion of participants reported shaking their pedometers and changing their activity patterns as a result of being measured, future pedometer research in young peoples should examine various protocols that attempt to limit tampering and reactivity. For example, a pedometer protocol where participants were informed prior to the monitoring period that pedometer shaking would be detected may limit tampering. Alternatively, pedometers may have

Chapter 5: Adolescents' perceptions of pedometer monitoring

more utility as an intervention strategy for increasing activity rather than a method for assessing habitual behaviour in young people.

Although there is growing support for the existence of reactivity and tampering in young people [10, 172, 277], there is little research into how and why participants may change their behaviour whilst being measured. Our findings suggest the majority of participants in the study were reactive to the monitoring process, and reported most of their peers were also. This was interesting, as there was no reinforcement or incentive for participants to increase their step count; participants were told the study aim was to gain an estimate of their 'normal' activity pattern. Many participants reported 'peer competition' as another motivator to change their activity patterns. Others also reported they liked knowing what their step count was and were motivated to complete more daily physical activity than normal. Although this study is in a different context (not an intervention), it potentially confirms previous research that pedometers can be used as a useful tool to motivate young peoples to increase physical activity levels, although without knowledge and behavioural skills it is questionable whether this would be maintained [93]. Our findings indeed highlight the need for researchers to evaluate the study protocol when using pedometers prior to collecting data as it affects adherence, level of reactivity and amount of tampering. Greater understanding of participant perceptions and behaviours exhibited will lead to more efficient and accurate data collection.

Despite the novelty of this study, there are some limitations that should be noted. First, the study sample may not be fully generalizable to the target population, more qualitative studies in different young people populations are needed. Second, the monitoring period only seven days, a longer monitoring period would allow researchers to determine if reactivity and tampering would taper or be sustained. Third, triangulation with objective pedometer data was not completed. Future studies are encouraged to complete triangulation of data sources to improve conformability of findings. Fourth, due to feasibility limitations, the focus group questions were not pilot tested. Finally, participants were not questioned on pedometer protocol or whether behaviours differed on each day of the week, or in particular the first day of

monitoring. Although a large proportion of the participants reported changing their normal activity patterns and tampering with their pedometers, we did not ask the participants about daily differences of reactivity and tampering. It is therefore difficult to determine if reactivity and tampering are ongoing or taper off over time. Future research in this population group should investigate whether reasons for changing activity pattern and pedometer tampering is consistent throughout the monitoring phase.

5.6 Conclusion:

This study is novel as it provides insights into young peoples' perceptions of, and behaviours during, the objective physical activity monitoring process. Our findings are consistent with previous quantitative studies that suggest reactivity and tampering are commonplace among young boys and girls. Of note, the vast majority of participants reported changing their activity patterns and tampering with the devices. In addition, there was unanimous agreement among participants that their peers also changed their activity patterns. Although the majority of participants believed that pedometers have value as a research tool, their behaviours and previous research [93] suggest that pedometers have more utility as an intervention strategy for increasing activity rather than a method for assessing habitual activity levels. As the research remains relatively sparse in this area, to properly understand participants' behaviours and perceptions, the authors recommend further qualitative research related to physical activity monitoring across populations groups and in free-living conditions.

	Sample (n = 123)		Focus group sample (n= 24)		Daily sealed group (n = 46)		Unsealed group (n = 44)		Weekly sealed group (n = 33)	
	Mean	SD	Mean	SD	Mean	SD	Mean	SD	Mean	SD
Age (years)	14.7	0.4	14.7	0.5	14.7	0.4	14.6	0.5	14.7	0.5
Height (cm)	169	9.6	168	9.9	171	8.8	168	10.7	169	8.5
Weight (kg)	60.4	11.3	62.7	13.4	58.8	8.5	63.6	13.4	58.7	10.3
Body Mass Index (BMI)	21.1	3.0	22	3.9	19.8	3.8	22.7	3.4	20.3	3.5
	n	%	n	%	n	%	n	%	n	%
Country of birth ¹	118	97	23	95.9	46	100	42	95	32	97
Language spoken at home ²	116	96	23	95.9	43	94	42	95	33	100
Ethnicity ³										
Australian	102	84.4	23	96	37	80.5	33	75	29	88
Aboriginal/Torres-strait Islander	2	1.6	1	4	0	0	2	4.5	0	-
European	14	11.6	0	0	7	14.5	5	11.4	2	6
Other	3	2.4	0	0	0	0	4	9.1	2	6

Table 5.1: Characteristics of larger and focus group sample

¹Participants born in Australia (missing n=3) ²Participants who speak English at home (missing n=1) ³Ethnicity (missing n=2)

Table 5.2	: Examples	of focus	group	questions
1 4010 2.2	• L'Aumpres	or rocus	Stoup	questions

Juestion
ell me what you thought of wearing a pedometer?
vid you feel that you had to get a high step count to impress the researchers, teachers
r your friends? Why or why not?
o you think that most people are honest or dishonest about their step count?
o you think that trying to find physical activity levels with pedometers is a waste of
me? Why or Why not?
bid you do more physical activity than normal when wearing a pedometer? Why or
'hy not?
vid you attempt to increase your step count in anyway? If so, how?
/hat were the other kids doing? Do you think they changed their behaviours because
ney were being measured?
id you let anyone else wear your pedometer? Why/why not?

CHAPTER 6: COMPARABILITY AND FEASIBILITY OF WRIST- AND HIP-WORN ACCELEROMETERS IN FREE-LIVING ADOLESCENTS

Preface: This chapter investigates that feasibility and concurrent validity the wrist-worn (GENEActiv) accelerometer in comparison to the hip-worn (ActiGraph GT3X+) in free-living adolescents. This chapter presents particularly novel findings as it uses 7 days of accelerometry in free living adolescents and self-report information to explore participants preferred accelerometer placement site on the body, which was more comfortable and if they would be willing to wear it on the hip or the wrist again. The findings from this chapter are important for researchers attempting to increase adolescent compliance to measurement protocols.

Citation:

Scott, J. J., Rowlands, A. V., Morgan, P. J., Plotnikoff, R. C., & Lubans, D. R (2017).
Comparability and feasibility of wrist- and hip-worn accelerometers in free-living adolescents, *Journal of Science and Medicine in Sport,* doi: 10.1016/j.jsams.2017.04.017 (Article in press)

Published in the Journal of Science and Medicine in Sport (Accepted 16 April 2017).

6 Comparability and feasibility of wrist- and hip-worn accelerometers in free-living adolescents

6.1 Abstract

Objective: To determine the comparability and feasibility of wrist- and hip-worn accelerometers among free-living adolescents.

Design: 89 adolescents (age = 13 - 14y old) from eight secondary schools in New South Wales (NSW), Australia wore wrist-worn GENEActiv and hip-worn ActiGraph (GT3X+) accelerometers simultaneously for seven days and completed an accelerometry behaviour questionnaire.

Methods: Bivariate correlations between the wrist- and hip-worn out-put were used to determine concurrent validity. Paired samples t-test were used to compare minutes per day in moderate-to-vigorous physical activity (MVPA). Group means and paired sample t-tests were used to analyse participants' perceptions of the wrist- and hip-worn monitoring protocols to assist with determining the feasibility.

Results: Wrist-worn accelerometry compared favourably with the hip-worn in average activity (r = 0.88, p < 0.001) and MVPA (r = 0.84 p < 0.001, mean difference = 3.54 mins/day, SD = 12.37). The wrist-worn accelerometer had 50% fewer non-valid days (75 days, 12%) than the hip-worn accelerometer (n = 152, 24.4%). Participants reported they liked to wear the device on the wrist (p < 0.001), and that it was less uncomfortable (p = 0.024) and less embarrassing to wear on the wrist (p < 0.001). Furthermore, that they would be more willing to wear the device again on the wrist over the hip (p < 0.001).

Conclusions: Our findings reveal there is a strong linear relationship between wrist- and hipworn accelerometer out-put among adolescents in free-living conditions. Adolescent compliance was significantly higher with wrist placement, with participants reporting that it was more comfortable and less embarrassing to wear on the wrist.

Key words: ActiGraph, GENEActiv, Physical Activity, Compliance, Perceptions, Youth.

6.2 Introduction

Poor adherence to accelerometer monitoring protocols and subsequent missing data are major issues for researchers [104, 114, 140]. Low compliance reduces the sample size and subsequent statistical power, while high compliance is desirable because it provides a more accurate representation of habitual activity patterns [295]. Adolescents, in particular, have been a challenging population to measure with accelerometers [115, 172]. Reasons for poor compliance among adolescents include: dissatisfaction with the size and comfort of devices [172]; unwanted attention and increased risk of being bullied [140]; and feelings of embarrassment [61, 172]. Various strategies have been employed by researchers to increase compliance to accelerometer monitoring protocols, including: i) increasing the amount of researcher contact [296], ii) calls and SMS reminders [297], iii) activity logs [224], iv) gifts and cash incentives [140]. However, even with these strategies, compliance to accelerometer monitoring protocols among adolescents is poor, especially in longitudinal and experimental studies that require individuals to wear devices on multiple occasions [278, 298].

There is clearly an urgent need to reconsider accelerometer-monitoring protocols with adolescent populations. While the ActiGraph accelerometer is the most commonly used validated accelerometer in physical activity research [114], it is typically worn on the hip rather than the wrist and removed whilst sleeping, resulting in non-wear time prior to sleep time and after waking [260]. ActiGraph recently released the GT9X accelerometer which is designed to be worn at the wrist, which reflects a recent shift toward wrist-worn activity monitoring. Emerging research suggests that participants consider wrist worn devices to be less burdensome, resulting in higher levels of compliance [292, 293]. Notably, the National Health and Nutrition Examination Survey (NHANES) 2011–2012 revealed wear time was 100% greater for wrist-worn accelerometers in comparison to the previous years, when monitors were worn on the hip [157]. Such findings highlight the potential for using wrist-worn accelerometry increase participant's compliance to protocols.

In recent years, there has been a proliferation in the number of commercially available accelerometers designed for both hip and wrist placement (e.g. Fitbit[®], Smartband[®], Archos Activity Tracker[®], Vivofit[®]). While this has helped to reduce the cost of accelerometers and increase their use in large-scale research, it has introduced new challenges in the interpretation and comparability of accelerometer outputs [110]. The GENEActiv is a relatively new accelerometer, and laboratory studies using calibration with oxygen consumption, have shown that this wrist-worn device can accurately assess children's and adults' physical activity intensity [156, 299]. Moreover, a recent field-based study [104] compared the wrist-worn GENEActiv and the hip-worn ActiGraph GT3X+ monitors in children and found higher compliance for the wrist-worn device, regardless of the wear-time criteria applied. In terms of concurrent validity, the authors reported a strong positive association between output from the two accelerometers (MVPA, r = 0.83, p < 0.001).

To the authors' knowledge, no previous study has examined the acceptability and comparability of wrist- and hip-worn accelerometers among adolescents in free-living conditions. Therefore, the primary aim of this study was to test the comparability and feasibility of wrist- and hip-worn accelerometers in a sample of free-living adolescents. A secondary aim was to compare wear-time, missing data and participant perceptions of the wrist and hip device placement in this population. Improving our understanding of adolescents' perceptions of the monitoring process is vitally important, and will help guide researchers to improve the accuracy of assessment in a sub-population who have been largely neglected in physical activity research.

6.3 Methods

Data for the current study were collected during baseline assessments as part of the existing Switch-off 4 Healthy Minds' (S4HM) cluster randomised controlled trial [1]. S4HM was a recreational screen-time reduction intervention targeting male and female adolescents in Grade 7 (first year of secondary school) in eight independent schools in NSW, Australia (2014). Ethics approval for the study was obtained from the Human Research Ethics Committees of the University of Newcastle, Newcastle-Maitland Catholic Schools Office and the Diocese of

Broken Bay. All students in grade 7 were invited to be involved in the study and were considered eligible to participate in the S4HM study if they self-reported ≥ 2 hrs/day of recreational screen time per day. Students who satisfied the eligibility criteria and provided signed written informed parental consent, were invited to participate in the study. Of the 322 eligible participants, every third student from each school (n = 113) was randomly selected and invited to participant in this study component. Data were collected in New South Wales (NSW), Australia in April/May 2014 (Term 2 of the school year). Participants were asked to wear both wrist- and hip-worn accelerometers simultaneously for seven full days and complete an accelerometry behaviour questionnaire that was designed for the purpose of this study.

Participants wore the Gravity Estimator of Normal Everyday Activity (GENEActiv), seismic acceleration sensor, dynamic range +/- 8g, ActivInsights, Cambridgeshire, UK) on their non-dominant wrist. Using the GENEA software (version 2.2), the devices were initialized to collect tri-axial acceleration data at a sample rate of 100 Hz. Participants also wore the ActiGraph GT3X+ (monolithic differential capacitance sensor, dynamic range +/-6g, ActiGraph LLC, Pensacola, FL, USA) on the non-dominant hip. Using the Actilife5 (version 6.5.3) software the ActiGraph GT3X+ devices were initialized to collect tri-axial acceleration data at a sample rate of 80 Hz. This study used different sampling frequencies for each monitoring device, however due to the nature of the signal processing (summarising output over 15 s epochs) this difference in sampling frequency would not have impacted on the output [111]. The accelerometry behaviour questionnaire was based on a previous pedometer questionnaire designed to examine participants' perceptions of the objective monitoring process [172]. The questionnaire consisted of eight questions, each scored on a 5-point Likert- scale. The questions explored students' perceptions of wearing accelerometers with responses ranging from 1 = strongly disagree to 5 = strongly agree. The wording of the items was derived from a previous pedometer questionnaires designed for adolescents. They were then reviewed by academics that have an expertise in physical activity interventions for input and feedback. The questionnaire was administered on the last day of the monitoring phase. Responses from the

Chapter 6: Comparability and feasibility of wrist- and hip-worn accelerometers

accelerometry behaviour questionnaire were used to provide insights into participants' perceptions of the monitoring process and to compare feasibility of the wrist- and the hip-worn devices among adolescents in free-living conditions.

The monitoring process was a total of nine days from dispersal to collection of the monitors, the first and last day were excluded from the analysis as they were only partial monitoring days, leaving seven-days. This study used a 24hour/day wear-time accelerometer protocol, compliance has been shown to be higher with 24 h/day protocols for both wrist-worn [300] and hip-worn monitors [301]. Both accelerometers were time and date synchronised using the same clock to start recording at 00:01am on Day 1 and finish recording at 11:59pm on Day 9. On completion of the monitoring period, participants returned both accelerometers and completed the accelerometry behaviour questionnaire. The GENEActiv data were downloaded with GENEActiv software (version 2.2). R-package GGIR version 1.2-2 (http://cran.r-project.org) and was used to process and analyse GENEActiv .bin files (This includes autocalibration using local gravity as a reference) [302]. The software was used to detect abnormally high values and non-wear time, and to calculate the average magnitude of dynamic acceleration (Euclidean Norm minus 1 g, ENMO) over 15-seconds epochs, with negative values rounded up to zero.

$$ENMO = \sum \sqrt{x^2 + y^2 + z^2} - g$$

The cut-points applied to calculate MVPA based on ENMO values were taken from a recent study by Phillips et al [299] and adjusted for the 100 Hz sampling frequency and 15-second epochs. The adjusted ENMO value used in the current study was \geq 20gs. Individual days were classed as invalid and excluded if wear-time was less than 10h. The detection of non-wear followed the procedures of Van Hees and colleagues [302]. Non-wear was estimated using the standard deviation (SD) and value range of each axis, calculated for 60 minute windows with 15-minute moving increments. If the SD on two of the three axes was less than 13mg or the

value range was less than 50mg, the time window was classified as non-wear. In this study, there was no reclassification activity or imputation of missing data for the GENEActiv data, nor the GT3X+ data. Mean daily activity (ENMO, mg) and MVPA were the output variables used in the analysis.

The hip-worn ActiGraph GT3X+ data were downloaded with the Actilife5 (version 6.5.3) software. The GT3X files were converted to 15s epoch AGD files for analysis of count data. The data were cleaned and scored using Actilife5 software (version 6.5.3). Non-wear was defined as \geq 20 minutes of consecutive zero counts [303]. To remain consistent with previous studies, the wear-time criteria for both monitoring devices was \geq 10hours/day on \geq 3 days/week. Any participants that did not meet the minimum wear-time criteria were excluded from the analysis [114]. MVPA was estimated by applying the commonly used adolescent Evenson cutpoints to the vertical count data (i.e. \geq 2296 CPM) [284]. Mean daily activity (average daily vector magnitude counts (VM, cpm) and daily MVPA (mins/day) were the output measures used in analysis.

The mean daily activity and MVPA (mins/day) data, along with responses to the accelerometry behaviour questionnaire were imported and analysed using IBM Statistics (SPSS 12 Inc. Chicago, IL) software and alpha levels set at p < 0.05. The data for the GT3X+ and GENEActiv were matched on concurrent valid whole days where participants were wearing both devices at the same time (at least three valid week days of wear-time at \geq 10hrs/day on both wrist- and hip-worn accelerometer). A more sensitive approach such as epoch matching may have allowed a more accurate assessment of concurrent validity, however this approach may lack ecological validity. Pearson bivariate correlations between the wrist- and hip- (daily mean physical activity and daily MVPA) accelerometer output were calculated to examine the relationship of the wrist- and hip-worn accelerometer data over the monitoring period. Paired samples t-tests were used to explore individual difference in MVPA mins/day for week and weekend days. Frequency analysis was used to reveal the amount of days that participants did

not wear either device; or wore one device and not the other. Responses to the accelerometry behaviour questionnaire were coded and imported into SPSS to be analysed quantitatively. Analysis of group means and paired sample *t*-tests were used to analyse the differences in participants' perceptions of the monitoring process.

6.4 Results

A total of 113 participants were involved in the monitoring process, of which 89 (41 boys, 48 girls) met the inclusion criteria (at least three valid week days of wear-time at \geq 10hrs/day on both wrist- and hip-worn accelerometer) and were included in the concurrent validity analysis. If the participant wore only once device and not the other, this day was removed and excluded from the analysis and treated as missing data (n = 132 days removed, 21.18%). Only 57 participants provided both wrist and hip accelerometry data for weekend days. Due to one participant being absent at the time of questionnaire completion, 112 participants completed the accelerometry behaviour questionnaire. The questionnaire responses were only included in the analysis if the participant had met the accelerometry inclusion criteria, the remaining (n = 23) were excluded.

Pearson bivariate correlations (shown in Table 1) revealed strong associations between the wrist- and hip-worn output in both daily mean activity (r = 0.88, p < 0.001, 95% CI = 0.82 -0.93) and MVPA (r = 0.84, p < 0.001, 95% CI = 0.77 - 0.89) over the 7 days. When analysing the weekdays only, the correlations were strong but slightly lower in both daily mean activity (r= 0.84, p < 0.001, 95% CI =0.76 - 0.89) and MVPA (r = 0.79, p < 0.001, 95% CI = 0.72 - 0.85). Furthermore, weekend days only, whilst lower again, there was moderate association for daily mean activity (r = 0.71, p < 0.001, 95% CI = 0.56 - 0.82) and MVPA (r = 0.53, p < 0.001, 95% CI = 0.37 - 0.71). Paired samples t-tests (shown in Table 2) revealed a low mean MVPA difference between the wrist and hip output on both weekdays (mean difference = 3.54 mins/day, p = 0.001) and weekend days (mean difference = 1.57 mins/day, p = 0.632). Tests for proportional bias indicated that there was a correlation between mean physical activity and the difference in estimates for acceleration (r = 0.18, p > 0.05) and MVPA (r = 0.46, p < 0.001).

Participant compliance to the monitoring protocols ranged from 3-7 days and is presented in Table 3. Overall the hip-worn accelerometer (152 days, 24.4%) had twice as many non-valid (missing) days than the wrist-worn accelerometer (75 days, 12%). In boys, there was minor difference in compliance between wrist and hip (wrist = 43 days, hip = 56 days), whereas for girls; there were three times as many non-valid days for the hip-worn accelerometer (n = 96), compared to the wrist-worn accelerometer (n = 32). Paired samples t-tests of non-valid (missing) days revealed that the mean difference between wrist- and hip-worn data was 11days (SD = 10.6, p = 0.031), for weekdays only (mean difference = 5.4, SD = 4.5, p > 0.05) and weekend days only (mean difference = 25, SD = 7.1, p > 0.05). Analysis of the excluded data due to participants only wearing one device revealed that participants were three times more likely to wear the wrist-worn accelerometer (n = 33 missing days) than only wear the hip-worn accelerometer (n = 99 missing days).

Participants reported a preference for the wrist-worn accelerometer (mean = 3.18, SD = 0.10), compared to the hip-worn accelerometer (mean = 2.51, SD = 1.01). Participants reported wearing the device on the wrist to be less uncomfortable and less embarrassing (mean = 1.93, SD = 0.10) to wear, compared to the hip-worn accelerometer (mean = 3.35, SD = 1.06; mean = 2.42, SD = 1.20, respectively). Participants reported they would be more willing to wear the wrist-worn accelerometer (mean = 3.65, SD = 1.06) than the hip-worn accelerometer (mean= 2.74, SD = 1.33) in future assessments. Participants, particularly adolescent females, reported the wrist-worn accelerometer to be more comfortable (p = 0.032), and less embarrassing (p < 0.001), to wear than the waist-worn accelerometer, but there were no other statistical differences between the wear sites based on mean scores for both sexes.

6.5 Discussion

This study investigated the concurrent validity and feasibility of wrist- and hip-worn accelerometers in a free-living adolescent population. We found that the wrist-worn accelerometer output compared favourably with the hip-worn accelerometer output in both mean daily activity and MVPA. In addition, the participants reported that they liked wearing the wrist-worn accelerometer more than the hip-worn accelerometer and would be more willing to wear it again on the wrist over the hip. Furthermore, there was three times as much missing data for the hip-worn accelerometer than that of the wrist-worn accelerometer.

By comparing the wrist-worn (GENEActiv) accelerometer output to the previously validated hip-worn (ActiGraph GT3X+) output under free-living conditions, this study provides an important contribution. Our findings revealed that the physical activity outcomes from the wrist-worn accelerometer were strongly associated with hip-worn accelerometer output for both physical activity patterning (r = 0.88, p < 0.001) and ranking of activity level (r = 0.84, p < 0.001) 0.001) within the sample, however absolute values differed. In MVPA minutes, there was also a low mean difference (3.54 mins/day weekdays, 1.57 mins/day weekend days) between the devices with the hip-worn accelerometer estimating slightly higher activity on both weekdays and weekend days. A strength of this study was that it reported the mean difference (minutes) between the wrist- and hip-worn accelerometers, which has not previously been done in previous adult and child GENEActiv validation. These findings may have relevance for researchers interested in evaluating physical activity intervention effects, but it is important to note that the results are dependent on the selected cut-points, population group and monitor used. Our results support previous findings reported in children and adults, where the wristworn GENEActiv compared well to the hip-worn ActiGraph GT3X+ in both acceleration and MVPA [104]. Moreover, the small mean difference between the monitors provides a unique contribution to the literature.

However, our findings differ to the previous findings of GENEActiv studies that concluded that the wrist worn GENEActiv had higher physical activity estimates than the hip worn ActiGraph [104, 111]. Further equivalency studies are required to determine the interchangeability of the devices, application of cut-points and site placement of the accelerometer. To remain consistent with the literature, the application of cut-points was based on previous studies in adolescents that have used ActiGraph and GENEActiv accelerometers to quantify activity [145, 284, 304]. For the GENEActiv monitor, this study used the Phillips et al [299] left wrist cut-points. Evenson cut-points were applied to the ActiGraph GT3X+ data. As cut-points are developed specifically for each accelerometer monitoring device and are both protocol- and population-specific, direct comparison of devices is very difficult, and further research into the application of different cut-points and the influence on estimates of sedentary, light, moderate and vigorous physical activity is clearly warranted.

A recent study in free-living adult women compared physical activity estimates for both hip and wrist site placements and concluded that there was only moderate correlation between the two sites [305]. A further study in an older adult population compared physical activity estimates for different wear-time protocols and hip and wrist placement with GT3X+ accelerometers. Findings revealed that wear-time adherence for the hip and wrist only varied by 2.7%, however, activity estimates for hip and wrist were statistically different and varied by as much as 41% [306]. These results differed to a study that compared GT3X+ accelerometer activity estimates at both the hip and wrist in pre-school aged children, which found a strong correlation between hip and wrist (r = 0.81, p < 0.01) accelerometer output. However, but large systematic bias with wide limits of agreement were observed [307]. Differences in accelerometer protocol and data reduction techniques for both hip and wrist accelerometers may explain the current inconsistencies found in the literature. Standardisation of hip and wrist accelerometer protocols are clearly warranted.

A recent study concluded that the GENEActiv can accurately assess physical activity intensities in children when worn at the hip or wrist [299]. Although previous research has also highlighted decoupling differences in hip and wrist accelerations depending on the type of activity and level of intensity in children [308]. This becomes more complex in free-living conditions when hip and wrist accelerations can be more disproportionate [157]. As the research is currently limited on preference for site placement in adolescents [104, 309], our study compared the wrist and hip placement site in a free-living adolescent population. It was expected that the hip and wrist accelerometer would be subjected to slightly different movements, which would account for some minor degree of error. Of note, correlations were lower on weekend days in comparison to weekdays. This finding may reflect the way that young people spend their weekends. For example, emerging research suggests that children are more active during weekdays, while on weekends they spend large amounts of time sedentary engaged in seated screen-based recreation [104, 157]. It has also been reported in the literature that activity in free-living adolescents is different on weekend days in comparison to weekdays where activity is commonly more routinized and structured [310]. These differences of daily activity patterns may also influence the accuracy of both hip and wrist activity estimates.

Previous research has identified the challenges of assessing physical activity using objective measures in adolescent populations [115, 277]. Studies in children have revealed that compliance is higher when devices are worn on the wrist, in comparison to hip placement [104]. This study was designed to confirm this finding among adolescents in free-living conditions and assess participants' perceptions of the two placement sites. To determine feasibility, adolescents self-reported their perceptions of the two placements sites and the research team compared accelerometer protocol compliance. Compliance to the monitoring protocol was operationalized as whole days that the monitor was worn, rather than periodic removal. Our results revealed twice as much non-valid (full days) for the hip-worn accelerometer (n = 152, 24.4%), compared to the wrist-worn accelerometer (75 days, 12%). The boys (wrist = 43, hip = 56) had fewer total missing days for the wrist-worn accelerometer. In girls, there was three times more missing

days on the hip-worn accelerometer than the wrist-worn accelerometer (wrist = 32, hip = 96). Hence, in both sexes there was less missing data for the wrist-worn accelerometer indicating higher compliance to the seven-day protocol. The results also revealed that participants were three times more likely to wear only the wrist-worn accelerometer.

This study is novel as it used a self-reported questionnaire, to not only investigate participants' preferred accelerometer site placement, but also investigated some of the potential reasons why. Our findings showed that adolescents found the wrist-worn device to be more comfortable and less embarrassing to wear. Interestingly, there were sex differences; girls reported that they found the hip-worn accelerometer more embarrassing to wear than the boys. Previous research has shown that girls have higher dissatisfaction with body image and concerns with body changes than boys [311], which could be a reason for preferring to wear the device on the wrist rather than around their waist. This may be an important finding as adolescent girls are a target population for physical activity interventions due to low activity levels [140, 222].

Both the hip-worn GT3X+ and the wrist-worn GENEActiv devices are robust, waterproof, light weight and have long battery life. An additional advantage of the GENEActiv is its watch-like appearance. Indeed, the design and appearance of the accelerometer may be a key determinant in increasing adolescent compliance, especially in girls who have previously shown poor adherence to physical activity intervention protocols and reported a dislike to the physical appearance of accelerometers [61, 224]. Wrist-worn GENEActiv accelerometer data correlated with the previously validated hip-worn GT3X+data (mean activity: r = 0.88, p <0.001; MVPA: r = 0.84, p < 0.001), and also daily compliance was far higher. The participants reported they liked wearing the accelerometer on the wrist more than the hip, and reported a higher willingness to wear it again on the wrist over the hip. This may be an important finding, as previous physical activity intervention research in adolescents have found poorer compliance to accelerometer protocols in post-test and follow-up assessments [224, 278]. As non-

compliance to accelerometer protocols is such a complex issue in the adolescent population [114, 140], further investigation into reasons why participants choose to wear one device over the other or simply not comply with protocol is warranted.

To our knowledge, this the first study to examine the concurrent validity and feasibility of wrist- and hip worn accelerometers among adolescents in free-living conditions. Despite the importance of our study findings, some limitations should be noted. First, the sample was relatively small and findings may not be generalizable to the entire adolescent domain. Second, for MVPA, the cut-points used for each device may have affected the classification of intensity of activity. Third, Other than sex, no other participant characteristics were analysed for association with wear time. Fourth, this study primarily focused on comparability of hip and wrist accelerometer placement rather than the equivalency of the ActiGraph GT3X+ and the GENEActiv; further research using raw accelerometer data is warranted to determine the interchangeability of the two monitoring devices. Finally, the accelerometry behaviour questionnaire was not piloted with a group of adolescents prior to data collection and has not been fully validated in adolescents.

6.6 Conclusion

The wrist-worn accelerometer (GENEActiv) showed good concurrent validity when compared to the previously validated hip-worn accelerometer (ActiGraph GT3X+) in both daily mean activity and MVPA. Our findings reveal there is a strong linear relationship between the wrist-and hip-worn accelerometer output and that daily wear-time compliance was far higher for the wrist-worn accelerometer. Overall, adolescents reported a preference for the wrist-worn accelerometer to be more comfortable and less embarrassing to wear and importantly, that they would be more willing to wear it again on the wrist rather than the hip. The use of wrist-worn accelerometer may assist researchers to increase participant compliance to accelerometer protocols in free-living adolescents. The authors recommend further adolescent physical activity studies utilise

Chapter 6: Comparability and feasibility of wrist- and hip-worn accelerometers

wrist-worn accelerometry to increase the probability of higher compliance to protocol, as greater wear time will provide a more accurate assessment of habitual physical activity.

		Wrist vs Hip			
		Mean acceleration	MVPA		
		(VM vs ENMO mgs)	(Minutes)		
All days	r	0.88*	0.84*		
	p-value	< 0.001	< 0.001		
	n	89	89		
Weekdays only	r	0.84*	0.79*		
	p-value	< 0.001	< 0.001		
	n	89	89		
Weekend days only	r	0.71*	0.53*		
	p-value	< 0.001	< 0.001		
	n	58	58		

Table 6.1: Relationship between the wrist- and hip-worn accelerometer output

*Significant at 0.05

Chapter 6: Comparability and feasibility of wrist- and hip-worn accelerometers

	Wrist	Hip		
	Mean (SD)	Mean (SD)	Mean difference (SD)	<i>p</i> -value
Weekday MVPA mins/day (n= 89)	31.1 (19.2)	34.6 (19.1)	3.5 (12.4)	< 0.001
Weekend days mins/day (n= 57)	32.7 (21.5)	34.3 (27.3)	1.6 (24.2)	0.632

Fable 6.2: Paired sample t-	tests to explore individual	differences in MVPA
-----------------------------	-----------------------------	---------------------

		Wrist				Hip		
		All	Boys	Girls	All	Boys	Girls	
	n	89	41	48	89	41	48	
Monday	(Days missing)	0	0	0	1	1	0	
	(%)	0	0	0	1.2	2.4	0	
Tuesday	(Days missing)	5	3	2	10	2	8	
	(%)	5.6	7.3	4.2	11.2	4.9	16.7	
Wednesday	(Davs missing)	13	7	6	17	7	10	
5	(%)	14.6	17.1	12.5	19.1	17.1	20.8	
Thursday	(Days missing)	10	7	3	23	8	15	
	(%)	11.2	17.1	6.3	25.8	19.5	31.3	
T ' 1		24	10	10	20	11	17	
Friday	(Days missing)	24	12	12	28	11	1/	
	(%)	27.0	29.3	25.0	31.5	26.8	35.4	
Saturday	(Days missing)	16	8	8	36	14	22	
	(%)	18.0	19.5	16.6	40.4	34.2	45.8	
		_						
Sunday	(Days missing)	7	6	1	37	13	24	
	(%)	7.9	14.6	2.1	41.6	31.7	50.0	
Total days missing/week		75	43	32	152	56	96	
Total missing/week (%)		12.1	6.9	5.1	24.4	9.0	15.4	

Table 6.3: Comparison of non-valid (missing) days








Note: Moderate-vigorous physical activity (MVPA) reported in minutes

7 CHAPTER 7: DISCUSSION

7.1 Overview

As this thesis is presented as a series of publications, the findings for each of the research aims have been reported and discussed comprehensively in the preceding chapters. Therefore, the purpose of this chapter is to synthesise these findings and address the thesis overall aim and study objectives of the thesis.

The overall aim of this thesis was to investigate current physical activity measurement protocols to assess physical activity levels in adolescents.

The study objectives of this thesis were to:

- Assess the test-retest reliability of a single-item physical activity questionnaire for adolescents;
- Determine the concurrent validity of a single-item physical activity measure for adolescents, by testing it against accelerometer output;
- Explore the impact of different pedometer monitoring protocols on compliance, reactivity and tampering in a sample of adolescents;
- Explore adolescents' perceptions of wearing pedometers and investigate behaviours exhibited by participants while wearing pedometers;
- Test the comparability of hip- and wrist-worn accelerometers in the free-living adolescent population; and
- Compare wear-time, missing data and participant perceptions of hip- and wrist-worn accelerometers in the adolescent domain.

Furthermore, this chapter highlights the strengths and limitations of the studies conducted in this thesis, and provides a series of evidence-based recommendations for future research and practice.

This chapter includes the following sections:

- 1. Section 7.3: reliability and validity of a single-item physical activity questionnaire;
- Section 7.4: pedometer monitoring protocols in adolescents: compliance, reactivity and tampering;
- 3. Section 7.5: a qualitative investigation of adolescents' perceptions of the pedometer monitoring process; and
- 4. Section 7.6: comparability and feasibility of wrist- and hip-worn accelerometers in adolescents.

7.2 Research gaps addressed by this thesis

This thesis addressed the following existing gaps within the evidence base:

- Current physical activity questionnaires designed for young people vary in length and complexity [15]. There is a need for standardised self-report measures designed specifically for adolescents that are not burdensome for participants to complete. Short physical activity questionnaires have been validated in both child and adult populations [12, 53, 248, 312]. However, no single-item screening measure for adolescents has been validated against both an existing physical activity questionnaire and accelerometer output. Further research is warranted to determine the utility of single-item physical activity questionnaires for adolescents.
- 2. While there have been numerous pedometer validation studies [10, 313] and reviews of existing methodological elements in this domain [16, 87, 95], there is a lack of consensus regarding the optimal pedometer protocol for measuring habitual physical activity in adolescents.
- 3. Few studies have investigated adolescents' perceptions of, and behaviours exhibited, during objective pedometer monitoring [61]. Consequently, there is little evidence in relation to adolescents' perceptions of pedometer monitoring and the behaviours that they exhibit during the monitoring phase in free living conditions.

4. Little is known regarding the comparability of physical activity estimates from wristand hip-worn accelerometers in free-living adolescents. In addition, no previous study has investigated participants' perceptions of both devices among adolescents in freeliving conditions.

7.3 Validity and reliability of a single-item physical activity questionnaire

7.3.1 Overview of findings

With the growth of physical activity research in the last two decades, there has been a large number of physical activity questionnaires developed [314]. It has been suggested that shorter questionnaires may reduce the amount of missing data, reporting inaccuracies and participant fatigue [81]. As existing questionnaires vary in length and complexity [15], more simplified questionnaires may prove to be particularly useful in adolescents, who have shorter attention spans than adults [80].

Chapter 3 investigated the test–retest reliability and concurrent validity of a single-item physical activity questionnaire in a sample of adolescents, by comparing it to an existing physical activity questionnaire (OPAQ) and accelerometer-determined MVPA. The single-item questionnaire for the study was modified from a previous single-item measure that was validated in adults [12]. The single item questionnaire asked participants to report the amount of days in the past week in which they had engaged in 60 minutes of MVPA/day. Overall, the single-item measure compared well with the more comprehensive physical activity questionnaire (OPAQ) in both test–retest reliability and concurrent validity.

It should be noted that although the ActiGraph GT3X+ accelerometer was used as the criterion measure to validate the single-item questionnaire, accelerometers quantify physical activity differently to self-report measures. The single item measure asks respondents to report time spent in MVPA in the last 7days. However, the OPAQ, ask respondents to report time spent in a particular activity (i.e. type of activity, start/finish time). This can then be converted

into a MET count based on an intensity of the activity type [264]. Time-spent in various intensities can then be calculated which was completed for the OPAQ. However, inaccuracies can occur as this assumption is that the intensity of activity stays the same for the duration reported, researchers could therefore significantly over report activity intensity with the application of a MET count. This could account for the overestimation of self-reported MVPA in comparison to the accelerometer determined MVPA.

The single-item measure was found to have good repeatability (ICC = 0.75). Previous research has reported that reliability of physical activity questionnaires is often poorly assessed, and ICC is considered preferable to Spearman's rank or Pearson's correlation [14]. This is encouraging, as the research suggests that ICC values > 0.70 are considered to be acceptable [315], and a recent review of physical activity questionnaires in youth reported that most existing questionnaires were found to have moderate reliability (i.e., ICC = 0.64 - 0.80). The study in Chapter 3 was strengthened as the reliability testing also addressed proportional bias and typical error, as ICC only tests consistency between trials [13]. Pearson correlations between inter-trial difference and the mean of the trials revealed that the OPAQ had slight proportional bias (r = -0.17, 95% CI = -0.43 - 0.10, p = 0.139) and no relationship between participants' MVPA inter-trial difference for the single-item measure (r = 0.08, 95% CI = -0.12 - 0.26, p = 0.465). Although there were significant linear relationships between the single item measure and accelerometer output, absolute differences were large.

For concurrent validity, self-reported MVPA from the single-item questionnaire was then compared with participants' accelerometer-determined MVPA and self-reported MVPA from the OPAQ. Bivariate correlations between the single-item questionnaire and accelerometers and self-reported physical activity were r = 0.50 and r = 0.44, respectively. Recent reviews of youth physical activity questionnaires that have been validated against accelerometers reported that correlation coefficients ranged from 0.25 - 0.41 [14] and from 0.30 - 0.40 [71]. Thus, findings from this study suggest that the single-item questionnaire had comparable reliability and superior concurrent validity, compared with existing physical activity measures in adolescents.

7.3.2 Strengths and limitations

This study was novel as it tested a single-item physical activity measure in an adolescent population by validating it against accelerometer output and self-reported data from an existing physical activity questionnaire. Previous reviews of the validity of physical activity questionnaires have noted that reliability is often not assessed appropriately, as the majority of studies have used Spearman's and Pearson's correlations [57, 114]. In an attempt to overcome the limitations of previous research, this study assessed reliability in three ways (that is, repeatability, change in mean and proportional bias).

There are some limitations that should be noted. First, the convenience sample used for this study was small and relatively homogenous, which may limit the generalisability of findings. Second, adherence to accelerometer protocols was lower than desired, and the inclusion criteria were subsequently modified to maximise the sample size. The low adherence to the seven-day monitoring protocol meant that it was could not be determined whether participants met the recommended physical activity guidelines, based on output from both the single-item measure and the accelerometer. Third, this study did not randomise the order of questionnaire completion. Finally, due to feasibility restrictions, it was not possible to test the reliability and validity in a broader age range of adolescents (i.e., ages 12 - 18). Consequently, further testing of this measure in other adolescent populations is warranted.

7.3.3 Implications for practice and recommendations for future research

The single-item measure is simple to administer and appears to have utility as a screening tool for adolescents, to determine whether or not they are meeting physical activity guidelines. This measure does not, however, provide the detail of more comprehensive questionnaires (e.g., activity type, duration, intensity, time of day/week, daily activity patterns) and may lack the necessary sensitivity to identify and quantify associations with health outcomes and physical activity invention effects.

Physical activity questionnaires have many limitations. There is a vast number of
existing physical activity questions that vary in accuracy, type, length and complexity.
Researchers should base their choice of physical activity questionnaire on both study
aims and the age of the study population. Further research into the appropriate length
and detail of physical activity questionnaires will lead to more efficient and accurate
physical activity data. Due to the noted limitations with self-report measures,
researchers should consider using an objective method for assessing physical activity in
addition to any self-report questionnaire.

7.4 Pedometer monitoring protocols in adolescents: compliance, reactivity and tampering

7.4.1 Overview of findings

Research in child and adult populations suggests that reactivity does not exist if pedometers are sealed [58, 160, 270]. Moreover, research on pedometer tampering in adolescents is sparse, and the level of tampering based on the type of pedometer protocol administered has not been investigated [100]. The aim of the study in Chapter 4 was to investigate adolescents' reactivity and tampering while wearing pedometers, by comparing different monitoring protocols to accelerometer output. A secondary aim of Chapter 4 was to explore adolescents' perceptions of wearing pedometers and investigate some of the behaviours they may exhibit while being measured.

In the study discussed in Chapter 4, participants were randomly allocated to one of three pedometer protocols (daily sealed, weekly sealed and unsealed pedometers). Repeated measures ANOVA was used to explore potential reactivity during week days. There was evidence of reactivity in both unsealed groups, yet step counts remained relatively stable across the monitoring period in the weekly sealed group. This was most likely due to sealing of the pedometer and minimising adolescents' step count feedback. Of note, participants in the daily sealed group reported a higher desire to impress the researchers by changing their normal activity pattern, most commonly by inflating their step counts. This was an important finding, as participants in this group were required to report to a researcher each day to have their step counts recorded. Almost half (49%) of participants reported tampering with their pedometers (shaking to increase step-counts) during the seven days of monitoring. Furthermore, 40% indicated they did not like wearing pedometers, with 81% finding them uncomfortable, and 69% finding them embarrassing to wear at times. These statistics provide some insight into why non-compliance with pedometer monitoring protocols is so high among adolescents. These findings also highlight the need for less burdensome and more discrete and comfortable devices, to increase compliance in this population.

The findings clearly demonstrate that reactivity and tampering occur in adolescents and that the protocol selected for pedometer monitoring may influence compliance. Correlations between step counts and accelerometer counts were strongest in the weekly sealed group, suggesting that this was the most reliable protocol of the three. Correlations with accelerometer output were weakest in the daily sealed group, suggesting potential tampering, perhaps due to daily contact with researchers. The results revealed that adolescents find objective measuring devices uncomfortable and embarrassing to wear, and a large proportion reported a desire to impress their peers and/or researchers by inflating their normal step count (e.g. by increasing activity or shaking their pedometer). Almost half the participants self-reported tampering with their pedometers.

7.4.2 Strengths and limitations

This study used seven days of objective monitoring to evaluate three different pedometer protocols. The protocols differed not only with various restrictions in the nature of feedback of step counts, but also in the level of contact with the researchers. A major strength of this study was the simultaneous assessment of physical activity using accelerometers and pedometers and the triangulation of findings using questionnaires. Indeed, the self-report questionnaire data provided valuable insights into the way that adolescents behaved during pedometer monitoring and some of the reasons for changing their activity patterns.

Despite these strengths, there are some limitations that should be noted. First, the compliance with objective monitoring protocols was lower than expected. The wear-time criteria was modified to include as many participants as possible for the analysis. Second, a pattern of reactivity was evident in the daily sealed and unsealed groups, but it is not definitive that it was not a 'day-to-day' difference in physical activity (e.g., participation in physical education or sport training). This is unlikely, however, as the weekly sealed group showed no significant differences across days. Third, pedometers were only sealed with stickers. The CW700 pedometers were selected due to their ability to record steps over seven days. However, these devices are not lockable and cannot be sealed with 'zip-ties'. Stickers were used to give

researchers daily access to step counts in the daily sealed group and also to enable researchers to tell if the pedometers had been opened or tampered with. Of note, many of pedometers that were returned displayed evidence of tampering (i.e., torn sticker). Finally, due to feasibility limitations, this study did not use a cross-over design, which would have provided an opportunity to assess the adolescents using each of the three monitoring protocols.

7.4.3 Implications for practice and recommendations for future research

- As the weekly sealed pedometer monitoring protocol was the most accurate and least influenced by reactivity and tampering, researchers should use at least a seven-day pedometer monitoring timeframe with sealed pedometers.
- Due to the level of reactivity and self-reported tampering with pedometers, researchers may instead use accelerometers, which normally do not provide a step count feedback to participants. The level of researcher contact may influence participants' desire to impress. Hence, reducing the contact between researchers and participants may lower the chance of participants feeling the need to change their normal activity patterns. It is therefore recommended that researchers use pedometers that have the ability to store data in their internal memory for seven days (or longer). These pedometers are required to be time set prior to monitoring; however, once initialised, they will automatically restart the step-count each day. This eliminates the risk of accidental reset and the requirement for participants to self-report daily step counts.
- Pedometer sealing is recommended to limit the feedback effect. Stickers are useful to determine whether the pedometer has been opened or tampered with; however, more stringent measures (such as the use of lockable monitors) will ensure that the pedometer is not opened for the entire monitoring period. Furthermore, manufacturers may need to consider the production of lockable pedometers for research purposes.
- This study revealed that a third of participants reported completing more activity than normal when being measured. The feedback of step counts appears to be an added

advantage of pedometers, and may still be a useful motivator for adolescent physical activity interventions.

- Manufacturers of pedometers and accelerometers for research purposes should endeavour to design devices that possess the ability to detect shaking through machine learning. This may enable researchers to exclude additional step counts derived from tampering (shaking) and therefore obtain a more accurate assessment of activity.
- Adolescents reported a variety of reasons for changing their activity pattern while being measured, which may differ from those of child and adult populations (e.g., peer competition). More research in this area is warranted to learn more about adolescents' perceptions of the monitoring process and reasons why they feel the need to change their activity patterns while being measured. Additional strategies are needed to incentivise adolescents to adhere to objective monitoring protocols.

7.5 A qualitative investigation of adolescents' perceptions of the pedometer monitoring process

7.5.1 Overview of findings

Although pedometers are commonly used in adolescent populations [85, 193, 247], little is known about adolescents' behaviours during the monitoring process and their perceptions of the monitoring process. Therefore, the aim of the study in Chapter 5 was to explore adolescents' perceptions of pedometers and investigate the physical activity behaviours exhibited by participants during the monitoring process. A secondary aim was to investigate whether participants' perceptions differed depending on their physical activity level.

Six focus groups were conducted to obtain participants' perceptions of the pedometer monitoring process in order to gain novel insights. Participants were purposely selected and organised into focus groups based on sex (boy/girl) and physical activity level (low/medium/high), as determined by accelerometers. The in-depth qualitative analysis yielded some unique insights about adolescents' attitudes towards, and behaviours relating to, pedometer use. Multiple themes were developed to: (i) provide insight into participants' perceptions of pedometer monitoring; (ii) identify the behaviours that participants exhibited while wearing pedometers; and (iii) investigate reasons why participants felt the need to inflate their step counts during the monitoring phase. Although some participants reported that the monitors were uncomfortable and/or embarrassing to wear, generally the attitudes towards objective monitoring were positive and the majority considered pedometers to be a useful tool for measuring physical activity.

Study findings suggest that most participants involved in the focus groups reported changing their activity pattern during the monitoring phase, with almost all participants believing their peers did the same. Many reported that peer competition was a strong motivator for changing normal activity patterns; however, this was more common in the medium- and high-active groups. Participants reported that they modified their normal activity pattern through increasing incidental physical activity (e.g., walking on the spot while watching TV)

and planned physical activity (e.g., going for a walk or jog), as well as increased effort and time in usual sports participation (e.g., running further than they usually would during a game). It was reinforced at the start of the monitoring phase that there was no incentive for a high step count and that researchers were attempting to obtain participants' 'normal' habitual activity. Participants were encouraged not to change their activity pattern, but for the majority of participants the opposite occurred.

In addition, 21 of the 24 participants in the focus groups reported shaking their pedometers to increase their step counts. 'Peer competition' was reported as a motivator for shaking the pedometer. This finding revealed the importance of sealing pedometers for adolescents if researchers are attempting to obtain accurate assessments of activity patterns. If the aim of pedometer use is to promote and increase physical activity in adolescent participants, then unsealed pedometers may be of great value.

The findings in Chapter 5 not only reveal the existence of reactivity and tampering in free-living adolescents, but also provide insight into why adolescents subconsciously or consciously change their activity patterns when being measured. The majority of participants in the focus groups reported changing their normal patterns while being measured, and believed their peers did the same. The findings indicate that reasons for changing activity patterns were not related to sex or physical activity level. As there is limited existing qualitative research in this area, further research investigating participants' perceptions of objective monitoring is warranted.

7.5.2 Strengths and limitations

A qualitative design using a focus group methodology was applied due to the likely benefits of group interaction, which elicits information and insights that may be less accessible during individual interview. The analysis was conducted using NVIVO software (QSR international Pty. Ltd, Melbourne, Australia) by a qualitative researcher, who was independent from the main study. To ensure reliability, a second researcher then conducted separate content analysis for

categorisation of themes. Finally, for consistency with data collection, the same researcher conducted all focus group interviews at all time-points.

Despite these study strengths, there are some limitations that should be noted. First, the study involved 24 participants from a homogenous sample of adolescents; therefore, the findings may not be generalisable to all adolescents. Second, although a period of seven days of objective monitoring is commonly recommended when measuring adolescents [120], this timeframe is relatively short when attempting to understand habitual physical activity. A longer monitoring period may allow researchers to determine whether reactivity and tampering are sustained or if they taper off over time. Finally, participants were not asked specific questions in relation to 'day to day' differences in amount of physical activity [316]. To further understand reactivity, future qualitative studies should investigate the key factors that contribute to day to day differences in perceptions and behaviours, to determine whether these would remain constant or taper.

7.5.3 Implications for practice and recommendations for future research

- A large proportion of adolescents reported changing their activity pattern while being measured. Future research should attempt to tailor pedometer protocols for the population group and the study aims. For example, adolescents may require a more stringent protocol than adults or children and, potentially, greater incentives to comply with study requirements. Adolescents need to be reminded before and during the monitoring phase that they should not change their activity patterns. Researchers could also advise the participants that shaking their device could be detected.
 - Peer-competition is clearly an issue when attempting to accurately quantify activity in adolescents. The study findings suggest that peer competition may increase reactivity and tampering. Researchers need to devise strategies that deter adolescents from engaging in peer-competition, by providing incentives to comply with monitoring protocols.

- Many of the participants reported that the pedometers were annoying and/or uncomfortable to wear. Again, this shows the need for more comfortable and less inhibiting devices. Adolescents, especially girls, reported that the pedometer was embarrassing to wear. In the adolescent population, the appearance of the pedometer may be a very important factor in improving compliance, maybe more so in than child and adult populations. As the monitors available on the commercial market are designed for consumer appeal rather than for scientific research purposes; they may be a useful tool for increasing compliance with monitoring protocols in adolescent populations. Future qualitative research should investigate the effect that appearance of the pedometer has on compliance in the adolescent domain. Researchers should also trial the use of more discreet monitoring devices that can be hidden under clothing, and popular commercially available wearable trackers (such as Fitbits, Jawbones etc.) to improve protocol compliance.
 - The study findings highlight the need for researchers to evaluate measurement protocols when using pedometers in pilot studies. Greater understanding of participants' perceptions and behaviours during the monitoring process may help improve measurement accuracy and efficiency.

7.6 Comparability and feasibility of wrist- and hip-worn accelerometers in adolescents

7.6.1 Overview of findings

Missing data due to non-compliance with protocols is a major weakness of objective monitoring using pedometers and accelerometers [104, 114, 140], particularly in adolescent populations [115, 147]. Accelerometers have traditionally been worn at the hip, and issues such as discomfort, size and feelings of embarrassment are some of the reasons adolescents have chosen not to adhere to accelerometer monitoring protocols [61, 140]. Accelerometers such as the GENEActiv have been developed specifically to be worn on the wrist. The study in Chapter 6 investigated the comparability and feasibility of wrist- and hip-worn accelerometers in free-living adolescents. The findings suggest that the wrist-worn GENEActiv output compared favourably to that of the hip-worn ActiGraph (GT3X+) for both mean daily activity (r = 0.88, p < 0.001) and MVPA/day (r = 0.84, p < 0.001). The results also revealed that there was a low mean difference between the wrist-mounted and hip-mounted monitors (3.54 minutes/day on week days).

Overall, there were twice as many missing days for the hip-worn accelerometer, when compared with the wrist-worn accelerometer. In boys, compliance was slightly higher for the wrist-worn device, whereas girls had three times as many missing days when the monitor was worn at the hip. Analysis of participants' choosing to wear only one device instead of both (as per the accelerometer protocol), revealed that participants were three times as likely to wear the wrist-worn accelerometer only. This is an important contribution to the literature, as it highlights the potential use of wrist-worn accelerometers to address the widespread issue of low compliance levels in adolescents.

Participants reported a preference for wearing the device on the wrist. Furthermore, they reported the device to be less uncomfortable and less embarrassing to wear when worn on the wrist. These findings are important, as the placement site, appearance and comfort of the monitor are factors that clearly affect the level of compliance with accelerometer protocols in

adolescents. It is important that researchers assess the appearance of the device and the effect it may have on compliance during assessment, as these findings reinforce previous findings in the literature that some adolescents, especially girls, feel at higher risk of being teased when wearing monitoring devices [61]. The findings of this study are consistent with the previous research, and highlight the importance of comfort and appearance of the accelerometers and the effect this can have on compliance with monitoring protocols in adolescents.

In recent years, there has been a large increase in the number of commercially available physical activity monitors [17]. The commercially available monitoring devices are predominantly designed to be worn on the wrist; potentially showing the shift from hip-worn accelerometer measurement. These commercial available activity monitors potentially reveal a new way to increase adolescent compliance in free-living conditions. The commercially available Fitbit[®] and Jawbone[®] accelerometers have recently been trialled in laboratory based studies, by comparing them with the previously-validated ActiGraph GT3X+ and GTX9. While there have only been two studies completed in youth populations, findings from these studies indicate high inter-device reliability ($r \ge 0.80$) [17]. Although these monitors show promise, the findings are only preliminary and further research in free-living conditions is required [317].

Although commercial wrist-worn accelerometers have been used in physical activity interventions, a recent systematic review have revealed that they have commonly been combined with other intervention approaches such as goal setting and research feedback, rather than for activity measurement [317]. Preliminary feasibility studies are consistent with the findings of the study in Chapter 6, highlighting the importance of design and comfort factors in children and adolescents [318, 319]. It has also been reported in the literature that the feedback is an important feature for adolescents, to motivate participants to be more active and comply with intervention monitoring protocols [317]. Many of the commercially available activity monitors use wireless connectivity to smartphone apps and provide participants with instant feedback about how much activity they have accumulated. Similar to pedometers, such devices

may have limited utility for measuring habitual physical activity due to reactivity. There is currently a paucity of studies in this area of commercially available accelerometers, and no recommendations exist for physical activity researchers attempting to limit the feedback effect and potential reactivity. Further research in this area is warranted.

The findings in Chapter 6 also revealed that participants self-reported that they would be more willing to the wear the monitoring device again on the wrist, rather than at the hip. This is an important finding, as longitudinal and intervention studies have shown that compliance is often poorer in subsequent assessment periods [115, 298, 320]. Overall, these findings suggest that free-living adolescents have a preference for wrist-worn accelerometers over hip-worn. Based on the findings in Chapter 6, the wrist-worn GENEActiv accelerometer, and others like it, may assist research to increase adolescent compliance with accelerometer protocols.

7.6.2 Strengths and limitations

This was the first study to examine the comparability and feasibility of wrist- and hip-worn accelerometers in adolescents in free-living conditions. A further strength and novelty of this study was that it used a self-report questionnaire to investigate adolescents' preferred accelerometer site placement and some of the potential reasons 'why'. A third strength was that this study used both mean activity and MVPA/day to compare the two monitoring placement sites. Fourth, experienced and trained researchers completed all of the assessments at all time-points.

Although this study had many strengths, there are some limitations that should be noted. First, the sample was relatively homogenous and study findings might not be generalisable to other populations of adolescents. Second, the accelerometer wear-time inclusion criteria were modified (that is, from 10 to 8 hours/day and from 4 to 3days/week) to increase participant numbers in the analyses. Third, as two brands of monitoring devices were used, different cut-points for each device were applied to classify intensity of activity for MVPA/day analysis. Fourth, non-wear time was only analysed in full days, and periodic

removal throughout the day was not assessed. Finally, the accelerometer behaviour questionnaire was developed for the current study and has not been validated.

7.6.3 Implications for practice and recommendations for future research

- The study findings suggest a strong linear association between the wrist- and hip-worn accelerometer outputs. However, due to the small sample size and limited analysis, further equivalency studies are required to determine the interchangeability of the ActiGraph GT3X+ and the GENEActiv accelerometers.
- As the study used two different brands of accelerometers (GENEActiv and ActiGraph), further comparability studies on accelerometer site placement should be completed with the same brands.
- Participants reported a preference for wearing the GENEActiv device on the wrist. Importantly, they found it more comfortable and less embarrassing to wear than the ActiGraph. These are very important findings, as adolescent compliance with accelerometer monitoring protocols is inherently poor. Appearance of the accelerometer is clearly important for adolescents, especially girls, who report a higher rate of feeling embarrassed when wearing the devices. Comfort and design appears to be a paramount issue affecting compliance. This highlights the need for more comfortable and discreet accelerometers to be developed for physical activity research purposes in adolescents, to address the current widespread non-adherence issue. When designing accelerometer protocols for specific study aims, researchers should be selective about the size and appearance of the accelerometer, as this has an effect on participant adherence. A further recommendation may be to use accelerometers that can be hidden under clothing, to reduce the chance of participants' feeling embarrassed during the monitoring period.
- The findings from this study indicate that comfort and appearance of the accelerometer are factors that can influence compliance in adolescents. Researchers could potentially trial a variety of types of wrist-worn accelerometer that differ in design, including

commercially available activity monitors, in an attempt to increase adolescent compliance with monitoring protocols. There is currently a paucity of research to determine the utility of commercially available accelerometers in physical activity measurement research across population groups. In addition, caution should be exercised when using commercially available monitors that provide instant feedback, as this may increase potential reactivity.

Researchers are encouraged to use wrist worn accelerometers when measuring physical activity in free-living adolescents. This may be particularly important in intervention and longitudinal studies where the participants are required to wear the monitoring devices on multiple occasions. The findings suggest that participants are not only more likely to comply with protocols when the device is worn on the wrist, but also that they would be more willing to wear it again on the wrist.

7.7 Concluding remarks

The accurate assessment of physical activity in adolescents is a challenging but important endeavour. The lack of consistent measurement protocols in the field of physical activity measurement is a result of the interplay that exists between cost/burden and accuracy/quality of data. Subsequently, researchers' choice of physical measure is not solely based on accuracy of the measure, it is dependent on a number of factors associated with the research aims, study design, sample and resource availability.

Physical activity questionnaires should be selected based on the specific study aims/design, and, importantly, the age and cognitive capacity of participants. Future researchers are encouraged to conduct additional testing of shorter, less complex self-report measures with adolescents, in an attempt to reduce participant burden and improve accuracy. The findings of the studies in this thesis suggest that reactivity to objective monitoring devices and pedometer tampering do exist in adolescents and are an inherent risk to validity. Strategies to limit reactivity and tampering should be considered by researchers during studies when attempting to accurately assess adolescents' physical activity patterns.

Further qualitative research may help to increase adolescents' compliance with objective monitoring protocols. In an attempt to increase compliance in this population, it is recommended that future research use wrist-worn accelerometers. The refinement of existing measurement tools, and the improvement of objective monitoring protocols in the adolescent domain, will consequently lead to more accurate assessments of adolescents' habitual activity patterns. This will be of particular importance in the improvement and success of physical activity interventions for this target population.

8 **Bibliography**

- Babic, M.J., et al., *Rationale and study protocol for 'Switch-off 4 Healthy Minds'* (S4HM): A cluster randomized controlled trial to reduce recreational screen time in adolescents. Contemp Clin Trials, 2015. 40: p. 150-158.
- Department of Health. Does your child get 60minutes of physical activity every day?
 2014 [cited 2017 18 February]; Available from: http://www.health.gov.au/internet/main/publishing.nsf/content/.
- Australian Government. National strategy for young Australians. 2010 [cited 2017 18 February]; Available from:

http://www.youthpolicy.org/national/Australia_2010_National_Youth_Strategy.pdf.

- 4. World Health Organisation. *Adolescents: health risks and solutions*. 2017 [cited 2017 18 February]; Available from: http://www.who.int/mediacentre/factsheets/fs345/en/.
- World Health Organisation. *Health Topics: Physical activity*. 2011 [cited 2013 5th June]; Available from: http://www.who.int/topics/physical_activity/en/
- Ainsworth, B.E., et al., 2011 Compendium of physical activities: a second update of codes and MET values. Med Sci Spor Exerc, 2011. 43(8): p. 1575-1581
- World Health Organisation, What is moderate-intensity and Vigorous-intensity physical activity?, in Global Strategy on Diet, Physical Activity & Health. 2014 [cited 2017 18
 February] Available from: http://www.who.int/dietphysicalactivity/physical_activity_intensity/en/
- Rowlands, A.V. and R. Eston, *The measurement and interpretation of children's physical activity*. J. Sports Sci. Med., 2007. 6(3): p. 270-276
- Clemes, S.A. and N.K. Deans, *Presence and duration of reactivity to pedometers in adults*. Med Sci Spor Exerc, 2012. 44(6): p. 1097-101.
- McNamara, E., Z. Hudson, and S.J. Taylor, *Measuring activity levels of young people:* the validity of pedometers. Br. Med. Bull., 2010. 95: p. 121-137.

- Oliver, M., et al., *Pedometer accuracy in physical activity assessment of preschool children*. J. Sports Sci. Med., 2007. 10(5): p. 303-310.
- 12. Milton, K., F. Bull, and A. Bauman, *Reliability and validity testing of a single-item physical activity measure*. Br J Sports Med, 2010. **45**(3): p. 203-208.
- Poppel, M.V., et al., *Physical activity questionnaires for adults: a systematic review of measurement properties.* Sports Med, 2010. 40(7): p. 565-600.
- Helmerhorst, H.J., et al., A systematic review of reliability and objective criterionrelated validity of physical activity questionnaires. Int J Behav Nutr Phy, 2012. 9: p. 103 epub doi: 10.1186/1479-5868-9-103.
- 15. Chinapaw, M.J., et al., *Physical activity questionnaires for youth: a systematic review of measurement properties.* Sports Med, 2010. **40**(7): p. 539-563.
- 16. Schmidt, M., et al., *Practical considerations when using pedometers to assess physical activity in population studies.* Res Q Exerc Sport, 2007. **78**(3): p. 162-170.
- Evenson, K.R., M.M. Goto, and R.D. Furberg, *Systematic review of the validity and reliability of consumer-wearable activity trackers*. Int J Behav Nutr Phy, 2015. 12: p. 159, epub doi: 10.1186/s12966-015-0314-1.
- Myers, J., *The Health Benefits and Economics of Physical Activity*. Curr Sports Med Rep, 2008. 7(6): p. 314-316.
- 19. World Health Organisation. Global Recommendations on Physical Activity and Health.
 2010 [2014 29 June] Available from: http://www.who.int/dietphysicalactivity/factsheet_recommendations/en/.
- 20. Behrens, T.K., et al., *Objectively assessed physical activity among Tongans in the United States.* Res Q Exerc Sport, 2011. **82**(3): p. 565-569.
- Janssen, I., & LeBlanc, A. G., Systematic review of the health benefits of physical activity and fitness in school-aged children and youth. Int J Behav Nutr Phy, 2010.
 7(40): p. 7-40

- 22. Hallal, P.C., et al., *Global physical activity levels: surveillance progress, pitfalls, and prospects.* The Lancet, 2012. **38**(9838): p. 247-257.
- Reiner, M., et al., Long-term health benefits of physical activity a systematic review of longitudinal studies. BMC Public Health, 2013. 13(1): p. 813. epub doi: 10.1186/1471-2458-13-813
- World Health Organisation, *Global Strategy on Diet, Physical Activity and Health: Physical activity and young people*. 2016, [cited 2017 February 18] Available from: http://www.who.int/dietphysicalactivity/strategy/eb11344/strategy_english_web.pdf.
- 25. Department of Health. Australia's Physical Activity and Sedentary Behaviour
 Guidelines. 2014 [cited 2015 27 June]; Available from: http://www.health.gov.au/internet/main/publishing.nsf/Content/health-central.htm.
- 26. United States Department of Health and Human Services. *Physical activity guidelines for Americans, Children and Adolescents*. 2008 [cited 2013 16th June]; Available from: http://www.health.gov/paguidelines/guidelines/chapter3.aspx
- 27. Canadian Society for Exercise Physiology, *Canadian physical activity guidelines for youth (12-17 years)*. 2011, [cited 2013 16 June] Available from: http://www.csep.ca/CMFiles/Guidelines/bcfiles/BC_CSEP_PAGuidelinesyouth_en.pdf.
- 28. Department of Health Physical Activity Health Improvement and Protection, Stay Active: A report on physical activity from the four home countries'. 2011 [cited 2013 24 June]Available from: https://www.gov.uk/government/publications/start-active-stayactive-a-report
- 29. Lee, I.M., et al., *Impact of Physical Inactivity on the World's Major Non-Communicable Diseases*. Lancet, 2012. **380**(9838): p. 219-229.
- 30. Bauman, A.E., et al., *Correlates of physical activity: why are some people physically active and others not?* Lancet, 2012. **380**(9838): p. 258-271.
- 31. Heath, G.W., et al., *Evidence-based intervention in physical activity: lessons from around the world*. Lancet, 2012. **380**(9838): p. 272-281.

- 32. Pratt, M., et al., *The implications of megatrends in information and communication technology and transportation for changes in global physical activity*. Lancet, 2012.
 380(9838): p. 282-293.
- 33. Kohl, H.W., et al., *The pandemic of physical inactivity: global action for public health*.Lancet, 2012. **380**(9838): p. 294-305.
- 34. Das, P. and R. Horton, *Physical activity-time to take it seriously and regularly*. Lancet, 2016. 338(10051): p. 1254-1255.
- Andersen, L.B., J. Mota, and L. Di Pietro, Update on the global pandemic of physical inactivity. Lancet, 2016. 388(10051): p. 1255-1256.
- 36. Ekelund, U., et al., Does physical activity attenuate, or even eliminate, the detrimental association of sitting time with mortality? A harmonised meta-analysis of data from more than 1 million men and women. Lancet, 2016. 388(10051): p. 1302-1310.
- 37. Ding, D., et al., *The economic burden of physical inactivity: a global analysis of major non-communicable diseases.* Lancet, 2016. **388**(10051): 1311-1324.
- 38. Australian Bureau of Statistics. *Physical Activity in Australia: A Snapshot 2007-08*.
 2011 [cited 2015 27th May]; Available from: http://www.abs.gov.au.
- 39. The Australian Department of Health, Australia's Physical Activity and Sedentary Behaviour Guidelines. 2014 [cited 2015 27th May]; Available from: http://www.health.gov.au/internet/main/publishing.nsf/Content/pasb
- 40. United States Department of Health & Human Services, *Healthy people 2010:* understanding and improving health (2nd Ed.). 2000, Washington D.C: U.S.
 Government Printing Office.
- 41. Econtech, *Economic modelling of the net costs associated with non-participation in sport and physical activity.* 2007 Econtech, Canberra.
- 42. Kimm, S.Y.S., et al., *Relation between the changes in physical activity and body-mass index during adolescence: a multicentre longitudinal study.* Lancet, 2005. **366**(9482): p. 301-307.

- Dalene, K.E., et al., Secular and longitudinal physical activity changes in populationbased samples of children and adolescents. Scand J Med Sci Sports, 2017, epub, doi: doi: 10.1111/sms.12876.
- 44. Allison, K.R., et al., *The decline in physical activity among adolescent students: a cross-national comparison*. Can J Public Health, 2007. **98**(2): p. 97-100.
- 45. Nader, P.R., et al., *Moderate-to-vigorous physical activity from ages 9 to 15 years*.JAMA, 2008. **300**(3): p. 295-305.
- 46. Cancer Council. Prevalence of meeting physical activity recommendations in australian secondary students. 2010 [20 June 2013]; Available from: www.cancervic.org.au/module_research/module_research_projects/secondary-studentsdiet-nassda.html.
- 47. Dumith, S.C., et al., *Physical activity change during adolescence: a systematic review and a pooled analysis.* Int J Epidemiol, 2011. **40**(3): p. 685-698.
- 48. Cataldo, R., et al., *Impact of physical activity intervention programs on self-efficacy in youths: a systematic review*. ISRN Obes, 2013. 2013: p. 586497. epub doi: 10.1155/2013/586497
- 49. Rachele, J.N., et al., *Practical physical activity measurement in youth: a review of contemporary approaches.* World J Pediatr, 2012. **8**(3): p. 207-216.
- 50. Dollman, J., et al., A hitchhiker's guide to assessing young people's physical activity:
 Deciding what method to use. J Sci Med Sport, 2009. 12(5): p. 518-525.
- 51. Lamonte, M.J. and B.E. Ainsworth, *Quantifying energy expenditure and physical activity in the context of dose response*. Med Sci Sports Exerc, 2001. **33**(6 Suppl): p. S370-378; discussion S419-420.
- Sirard, J.R. and R.R. Pate, *Physical activity assessment in children and adolescents*.
 Sports Med, 2001. **31**(6): p. 439-454.
- 53. Milton, K., S. Clemes, and F. Bull, *Can a single question provide an accurate measure of physical activity?* Br J Sports Med, 2013. **47**(1): p. 44-48.

- 54. Wareham, N.J. and K.L. Rennie, *The assessment of physical activity in individuals and populations: why try to be more precise about how physical activity is assessed?* Int J Obes Relat Metab Disord, 1998. 22(Suppl 2): p. S30-38.
- 55. Bauman, A., et al. *Physical activity measures for children and adolescents recommendations on population surveillance. an evidence check rapid review.* 2010
 [cited 25 2012 June]; Available from:

https://www.saxinstitute.org.au/publications/evidence-check.

- 56. Kohl, H., J. Fulton, and C. Caspersen, *Assessment of physical activity among children and adolescents: A review and synthesis.* Prev Med, 2000. **31**(2): p. S54-76.
- Sallis, J.F. and B.E. Saelens, Assessment of physical activity by self-report: Status, limitations, and future directions. Res Q Exerc Sport, 2000. 71(2): p. S1-14.
- Ozdoba, R.S., C.B. Corbin, and G.C. LeMasurier, *Does reactivity exist in children when measuring activity levels with unsealed pedometers?* Ped Exerc Sci, 2004. 16(2): p. 158-166.
- 59. Lockwood, J., et al., Comparison of a physical activity recall questionnaire with accelerometry in children and adolescents with obesity: a pilot study. Pediatr Obes, 2016 epub, doi: 10.1111/ijpo.12172.
- Audrey, S., et al., Adolescent perspectives on wearing accelerometers to measure physical activity in population-based trials. Eur J Public Health, 2012 23(3): p. 475-480.
- Kirby, J., et al., Young people's views on accelerometer use in physical activity research: findings from a user involvement investigation. ISRN Obes, 2012 (Article ID 948504): epub, doi: 10.5402/2012/948504.
- Loprinzi, P.D. and R.E. Davis, *Recent temporal trends in parent-reported physical activity in children in the united states, 2009 to 2014.* Mayo Clin Proc, 2016. **91**(4): p. 477-481.

- Arundell, L., et al., Contribution of the after-school period to children's daily participation in physical activity and sedentary behaviours. PLoS One, 2015. 10(10): p. e0140132, epub, doi: 10.1371/journal.pone.0140132.
- 64. Prince, S., et al., A comparison of direct versus self-report measures for assessing physical activity in adults: a systematic review. Int J Behav Nutr Phys Act., 2008.
 5(56): epub, doi: 10.1186/1479-5868-5-56.
- Trost, S., State of the art reviews: the measurement of physical activity in children and adolescents. Am J Lifestyle Med, 2007. 1(4): p. 299-314.
- 66. Leblanc, A. and I. Janssen, *Difference between self-reported and accelerometer measured moderate-to-vigorous physical activity in youth*. Ped Exerc Sci, 2010. 22(4):
 p. 523-534.
- Robbins, L., et al., *Physical activity, self-efficacy, and perceived exertion among adolescents*. Res Nurs Health, 2004. 27(6): p. 435-446.
- Basterfield, L., et al., Surveillance of physical activity in the UK is flawed: validation of the Health Survey for England Physical Activity Questionnaire. Arch Dis Child, 2008.
 93(12): p. 1054-1058.
- 69. Mota, J., et al., *Validation of a physical activity self-report questionnaire in a Portuguese pédiatrie population.* Ped Exerc Sci, 2002. **14**(3): p. 269-276.
- McMurray, R., et al., *Comparison of a computerized physical activity recall with a triaxial motion sensor in middle-school youth*. Med Sci Sports Exerc, 1998. **30**(8): p. 1238-1245.
- Foley, L., et al., Self-report use-of-time tools for the assessment of physical activity and sedentary behaviour in young people: systematic review. Obes Rev, 2012. 13(8): p. 711-722.
- Hagstromer, M., et al., Concurrent validity of a modified version of the international physical activity questionnaire (IPAQ-A) in European adolescents: The HELENA study. Int J Obes (Lond), 2008. 32(Suppl 5): p. S42-48.

- Piddle, S.J., et al., An assessment of self-reported physical activity instruments in young people for population surveillance: Project ALPHA. Int J Behav Nutr Phys Act, 2011.
 8(1): epub, doi: 10.1186/1479-5868-8-1.
- 74. Kowalski, K.C., P.R.E. Crocker, and N.P. Kowalski, *Convergent validity of the physical activity questionnaire for adolescents*. Ped Exerc Sci, 1997. **9**(4): p. 342-352.
- 75. Caspersen, C.J., M.A. Pereira, and K.M. Curran, *Changes in physical activity patterns in the United States, by sex and cross-sectional age.* Med Sci Sports Exerc, 2000. 32(9):
 p. 1601-1609.
- 76. Prochaska, J.J., J.F. Sallis, and B. Long, *A physical activity screening measure for use with adolescents in primary care.* Arch Pediatr Adolesc Med, 2001. **155**(5): p. 554-559.
- World Health Organisation. Social determinants of health and well-being among young people. Health Behaviour in School-aged Children (HBSC) study: international report from the 2009/2010 survey. 2012 [cited 2017 August]; Available from: http://www.euro.who.int/en/health-topics/Life-stages/child-and-adolescent-health/health-behaviour-in-school-aged-children-hbsc/.
- 78. Australian Bureau of Statistics. *Australian Health Survey: Nutrition and Physical Activity, 2011-12.* 2015 [cited August 2017; Available from: http://abs.gov.au/AUSSTATS/abs@.nsf/Lookup/4324.0.55.002Main+Features12011-12.
- 79. Lubans, D.R., K. Sylva, and Z. Osborn, Convergent validity and test-retest reliability of the Oxford Physical Activity Questionnaire for secondary school students. Behaviour Change, 2008. 25(1): p. 23-34.
- Garmy, P., U. Jakobsson, and P. Nyberg, Development and psychometric evaluation of a new instrument for measuring sleep length and television and computer habits of Swedish school-age children. J. Of Sch. Nurs, 2012. 28(2): p. 138-143.
- Ganassali, S., *The influence of the design of web survey questionnaires on the quality of responses*, Survey Research Methods, 2008. 2(1): p. 21-32.

- McKenzie, T.L., *The use of direct observation to assess physical activity. In G. Welk* (*Ed.*) *Physical activity assessments for health-related research.* Human Kinetics, , 2002 Champaign, IL: p. 179-195.
- 83. Montoye, H.J., Kemper, H., Saris, W., & Washburn, R. A., *Measuring physical activity and energy expenditure*. Human Kinetics, 1996, Champaign, IL
- 84. LaPorte, R.E., H.J. Montoye, and C.J. Caspersen, *Assessment of physical activity in epidemiologic research: problems and prospects*. Public Health Reports, 1985. 100(2): p. 131-146.
- 85. Lubans, D., Hesketh, K., Cliff, D., Barnett, L., Salmon, J., Dollman, J., Hardy, L., *A systematic review of the validity and reliability of sedentary behaviour measures used with children and adolescents.* Obes Rev, 2011. **12**(10): p. 781-799.
- Behrens, T.K., et al., *Accuracy of step recording in free-living adults*. Res Q Exerc Sport, 2007. 78(5): p. 542-547.
- 87. Craig, C., et al., *Process and treatment of pedometer data collection for youth: the Canadian physical activity levels among youth study*. Med Sci Sports Exerc, 2010.
 42(3): p. 430-435.
- Tudor-Locke, C., et al., *Pedometry methods for assessing free-living youth*. Res Q
 Exerc Sport, 2009. 80(2): p. 175-184.
- Ramirez-Marrero, F.A., et al., *Comparison of methods to estimate physical activity and energy expenditure in African American children*. Int J Sports Med, 2005. 26(5): p. 363-371.
- 90. Tudor-Locke, C., et al., *Utility of pedometers for assessing physical activity*. Sports Med, 2002. 32(12): p. 785-808.
- Beets, M.W., et al., Convergent validity of pedometer and accelerometer estimates of moderate-to-vigorous physical activity of youth. J Phys Act Health, 2011. Suppl 2: p. S298-305.

- 92. Eston, R.G., A.V. Rowlands, and D.K. Ingledew, *Validity of heart rate, pedometry, and accelerometry for predicting the energy cost of children's activities*. J Appl Physiol (1985) 1998. 84(1): p. 362-71.
- 93. Lubans, D.R., P.J. Morgan, and C. Tudor-Locke, A systematic review of studies using pedometers to promote physical activity among youth. Prev Med, 2009. 48(4): p. 307-315.
- 94. McClain, J.J. and C. Tudor-Locke, *Objective monitoring of physical activity in children: considerations for instrument selection*. J Sci Med Sport, 2009. 12(5): p. 526-533.
- 95. Corder, K., et al., *Assessment of physical activity in youth*. J Appl Physiol (1985), 2008.
 105(3): p. 977-987.
- Beets, M.W., et al., *Pedometer-measured physical activity patterns of youth: a 13-country review*. Am J Prev Med, 2010. 38(2): p. 208-216.
- 97. Kilanowski, C.K., A.R. Consalvi, and H. Epstein, *Validation of an electronic pedometer for measurement of physical activity in children*. Ped Exerc Sci, 1999. **11**(1): p. 63-68.
- 98. Clemes, S.A. and S.J. Biddle, *The use of pedometers for monitoring physical activity in children and adolescents: measurement considerations*. J Phys Act Health, 2013. 10(2): p. 249-262.
- 99. Miller, R., W. Brown, and C. Tudor-Locke, But what about swimming and cycling? how to "count" non-ambulatory activity when using pedometers to assess physical activity. J Phys Act Health, 2006. 3(3): p. 257-266.
- 100. Kahan, D. and N. Nicaise, Walk as directed! adolescent adherence to pedometer intervention protocols. J Phys Act Health, 2011. 9(7): p. 962-969.
- 101. Laurson, K.R., et al., Evaluation of youth pedometer-determined physical activity guidelines using receiver operator characteristic curves. Prev Med, 2008. 46(5): p. 419-424.
- Tudor-Locke, C., et al., *Expected values for pedometer-determined physical activity in youth*. Res Q Exerc Sport, 2009. 80(2): p. 164-174.

Bibliography

- 103. Tudor-Locke, C., Hatano, Y., Pangrazi, R. P., & Kang, M., *Revisiting how many steps are enough?*". Med Sci Sports Exerc, 2008. 40(7 Suppl): p. S537-543.
- 104. Fairclough, S.J., et al., *Wear compliance and activity in children wearing wrist and hipmounted accelerometers.* Med Sci Sports Exerc., 2016. **48**(2): p. 245-253.
- 105. Chen, Y.K. and D.R. Basset, *The technology of accelerometry-based activity monitors: current and future*. Med Sci Sports Exerc, 2005. **37**(11 Suppl): p. 490-500.
- 106. John, D. and P. Freedson, *Actigraph and Actipal physical activity monitors: A peek under the hood.* Med Sci Sports Exerc, 2012. **44**(1 Suppl 1): p. S86-89.
- 107. Ridgers, N.D., et al., Agreement between activPAL and ActiGraph for assessing children's sedentary time. Int J Behav Nutr Phys Act, 2012. 9(15) epub doi: 10.1186/1479-5868-9-15.
- 108. Carr, L.J. and M.T. Mahar, Accuracy of intensity and inclinometer output of three activity monitors for identification of sedentary behavior and light-intensity activity. J Obes, 2012. epub, doi: 10.1155/2012/460271.
- 109. Broderick, J.M., et al., *A guide to assessing physical activity using accelerometry in cancer patients*. Support Care Cancer, 2014. **22**(4): p. 1121-1130.
- Welk, G.J., J. McClain, and B.E. Ainsworth, *Protocols for evaluating equivalency of accelerometry-based activity monitors*. Med Sci Sports Exerc., 2012. 44(1 Suppl) p. S39-49.
- 111. Rowlands, A.V., et al., *Comparability of measured acceleration from accelerometrybased activity monitors.* Med Sci Spor Exerc, 2015. **47**(1): p. 201-210.
- Yang, C. and Y. Hsu, A review of accelerometry-based wearable motion detectors for physical activity monitoring. Sensor, 2010. 10(8): p. 7772-7788.
- 113. Reilly, J.J., et al., *Objective measurement of physical activity and sedentary behaviour: review with new data*. Arch Dis Child, 2008. **93**(7): p. 614-619
- Cain, K.L., et al., Using accelerometers in youth physical activity studies: a review of methods. J Phys Act Health, 2013. 10(3): p. 437-450.

Bibliography

- 115. Troiano, R.P., et al., *Physical activity in the United States measured by accelerometer*.Med Sci Sports Exerc, 2008. 40(1) p.181-188.
- Lonsdale, C., Rosenkranz, R. R., Peralta, L. R., Bennie, A., Fahey, P., & Lubans, D. R.
 , A systematic review and meta-analysis of interventions designed to increase moderateto-vigorous physical activity in school physical education lessons. Prev Med, 2013.
 56(2): p. 152-161.
- 117. Mattocks, C., et al., Use of accelerometers in a large field-based study of children: protocols, design issues, and effects on precision. J Phys Act Health, 2008. 5(Supp 1):
 p. S98-S111.
- 118. Dössegger, A., et al., *Reactivity to accelerometer measurement of children and adolescents*. Med Sci Sports Exerc, 2014. 46(6): p. 1140-1146.
- 119. Trost, S., et al., *Using objective physical activity measures with youth: How many days of monitoring are needed?* Med Sci Sports Exerc, 2000. **32**(2): p. 426-431.
- 120. Trost, S.G., K.L. McIver, and R.R. Pate, *Conducting accelerometer-based activity assessments in field-based research*. Med Sci Sports Exerc, 2005. **37**(11 Suppl): p. S531-543.
- Beets, M.W., M. Patton, and S. Edwards, *The accuracy of pedometer steps and time during in children*. Med Sci Sports Exerc, 2005. **37**(3): p. 513-520.
- 122. Crouter, S.E., et al., *Validity of 10 electronic pedometers for measuring steps, distance, and energy cost.* Med Sci Sports Exerc, 2003. **35**(8): p. 1455-1460.
- 123. Schneider, P.L., et al., *Accuracy and reliability of 10 pedometers for measuring steps over a 400-m walk*. Med Sci Sports Exerc, 2003. **35**(10): p. 1779-84.
- 124. Behrens, T.K. and M.K. Dinger, *Comparisons of accelerometer and pedometer determined steps in free living samples.* J Phys Act Health, 2011. **8**(3): p. 390-397.
- 125. Rowlands, A.V., et al., *The effect of accelerometer epoch on physical activity output measures.* J Exerc Sci Fit, 2006. **4**(1): p. 52-58.
- Basset, D.R., et al., Accuracy of five electronic pedometers for measuring distance walked. Med Sci Sports Exerc, 1996. 28(8): p. 1071-1077.

- 127. Nilsson, A., et al., Assessing physical activity among children with accelerometers using different time sampling intervals and placements. Pediatr Exerc Sci, 2002. 14(1): p. 87-96.
- McClain, J.J., et al., *Epoch length and accelerometer outputs in children: comparison to direct observation*. Med Sci Sports Exerc, 2008. 40(12): p. 2080-2087.
- 129. Baquet, G., et al., *Improving physical activity assessment in prepubertal children with high-frequency accelerometry monitoring: a methodological issue*. Prev Med, 2007.
 44(2): p. 143-147.
- Berman, N., et al., Spectral and bout detection analysis of physical activity patterns in healthy, prepubertal boys and girls, A J Human Biol, 1998. 10(3): p. 289–297.
- Banda, J.A., et al., Effects of Varying Epoch Lengths, Wear Time Algorithms, and Activity Cut-Points on Estimates of Child Sedentary Behavior and Physical Activity from Accelerometer Data. PLoS One, 2016. 11(3): epub doi: 10.1371/journal.pone.0150534
- Edwardson, C.L. and T. Gorely, *Epoch length and its effect on physical activity intensity*. Med Sci Sports Exerc, 2010. 42(5): p. 928-934.
- 133. Strath, S.J., K.A. Pfeiffer, and M.C. Whitt-Glover, Accelerometer use with children, older adults, and adults with functional limitations. Med Sci Sports Exerc, 2012. 44(1 Suppl 1): p. S77-85.
- 134. Kim, Y., M.W. Beets, and G.J. Welk, Everything you wanted to know about selecting the "right" ActiGraph accelerometer cut-points for youth, but...: a systematic review. J Sci Med Sport, 2012. 15(4): p. 311-321.
- 135. Puyau, M.R., et al., *Validation and calibration of physical activity monitors in children*.Obes Res, 2002. 10(3): p. 150-157.
- 136. Guinhouya, C.B., et al., *Moderate-to-vigorous physical activity among children: discrepancies in accelerometry-based cut-off points*. Obesity (Silver Spring), 2006.
 14(5): p. 774-777.

- Watson, K.B., et al., *Comparison of accelerometer cut points to estimate physical activity in U.S. adults.* J Sports Sci, 2014. **32**(7): p. 660-669.
- Mota, J., M. Valente, and L. Aires, *Accelerometer cut points and youth physical activity prevalence*. Eur Phys Educ Rev, 2007. 13(3): p. 287-299.
- 139. Trost, S.G., et al., *Comparison of accelerometer cut points for predicting activity intensity in youth.* Med Sci Sports Exerc, 2011. **43**(7): p. 1360-1368.
- 140. Sirard, J.R. and M.E. Slater, *Compliance with wearing physical activity accelerometers in high school students.* J Phys Act Health, 2009. **6**(Supp 1): p. S148-1455.
- 141. Vincent, S. and R. Pangrazi, *Does reactivity exist in children when measuring activity levels with pedometers?* Pediatr Exerc Sci, 2002. **14**(1): p. 56-63.
- 142. Tudor-Locke, C. and D.R. Bassett, Jr., *How many steps/day are enough? Preliminary pedometer indices for public health.* Sports Med, 2004. **34**(1): p. 1-8.
- 143. Tudor-Locke, C., et al., *Pedometer-determined step-count guidelines for classifying walking intensity in a young ostensibly health population*, Can J Appl Physiol, 2005. **30**(6): p. 666-676.
- 144. Rowe, D.A., et al., *Measuring physical activity in children with pedometers: Reliability, reactivity, and replacement of missing data.* Pediatr Exerc Sci, 2004. **16**(4): p. 343-354.
- 145. Ward, D.S., et al., *Accelerometer use in physical activity: best practices and research recommendations*. Med Sci Sports Exerc, 2005. **37**(11 Suppl): p. S582-588.
- Howie, E.K. and L.M. Straker, *Rates of attrition, non-compliance and missingness in randomized controlled trials of child physical activity interventions using accelerometers: A brief methodological review.* J Sci Med Sport, 2016 19(10) p. 830-836.
- 147. Catellier, D.J., et al., *Imputation of missing data when measuring physical activity by accelerometry*. Med Sci Sports Exerc, 2005. **37**(11 Suppl): p. S555-562.
- Toftager, M., et al., Accelerometer data reduction in adolescents: effects on sample retention and bias. Int J Behav Nutr Phys Act., 2013. 10(1): epub doi: 10.1186/1479-5868-10-140.
- Riddoch, C.J., et al., *Physical activity levels and patterns of 9-and 15-yr-old European children*. Med Sci Sports Exerc, 2004. 36(1) p. 86-92.
- 150. Coevering, P.V., et al., *Feasibility of using accelerometers to measure physical activity in young adolescents*. Med Sci Sports Exerc., 2005. **37**(5): p. 867-871.
- 151. Kwon, S. and K.F. Janz, *Tracking of accelerometry-measured physical activity during childhood: ICAD pooled analysis.* Int J Behav Nutr Phys Act., 2012. 9(68): epub doi: 10.1186/1479-5868-9-68.
- 152. Perry, M.A., et al., *Utility of the RT3 triaxial accelerometer in free living: An investigation of adherence and data loss.* Appl Ergon, 2010. **41**(3): p. 469-476.
- Crouter, S., P. Schneider, and D. Basset, *Spring level versus piezo electric pedometer* accuracy in overweight and obese adults. Med Sci Sports Exerc, 2005. **37**(10): p. 1673-1679.
- 154. Abel, M., *The effect of pedometer tilt angle on pedometer accuracy*. I J Fitness, 2008.
 4(1): p. 51-57.
- 155. Rosenberger, M.E., et al., *Estimating activity and sedentary behavior from an accelerometer on the hip or wrist*. Med Sci Sports Exerc, 2013. **45**(5): p. 964-975.
- 156. Hildebrand, M., et al., *Age-group comparability of raw accelerometer output from wrist and hip-worn monitors.* Med Sci Sports Exerc., 2014. **46**(9): p. 944-951.
- 157. Rowlands, A.V., et al., *Children's physical activity assessed with wrist- and hip-worn accelerometers*. Med Sci Sports Exerc, 2014. **46**(12): p. 2308-2016.
- 158. Wang, J. and J. Quek, *Measures of reliability and validity of school-based pedometer step counts in Singaporean children*. Asian J Exerc Spor Sci, 2005. **2**(1): p. 17-24.
- Clemes, S.A., N. Matchett, and S. Wane, *Reactivity: an issue for short-term pedometer studies?* Br J Sports Med, 2007. 42(1): p. 68-70.
- Matevey, C., et al., *Lack of reactivity during pedometer self-monitoring in adults*. Meas
 Phys Educ Exerc Sci, 2006. 10(1): epub doi: 10.1207/s15327841mpee1001_1.
- Clemes, S.A. and R.A. Parker, *Increasing our understanding of reactivity to pedometers in adults*. Med Sci Sports Exerc, 2009. 41(3): p. 674-680.

- 162. Inchley, J., L. Cuthbert, and M. Grimes, An investigation of the use of pedometers to promote physical activity, and particularly walking, among school-aged children: Review of evidence and scoping study. 2007, Edinburgh, Scotland: Child and Adolescent Health Research Unit.
- 163. Rowlands, A.V., Eston, R. G., & Ingledew, D. K., Measurement of physical activity in children with particular reference to the use of heart rate and pedometry. Sports Med, 1997. 24(4): p. 258-272.
- 164. Rowlands, A.V., Eston, R. G., & Ingledew, D. K., *Relationship between activity levels, aerobic fitness, and body fat in 8- to 10-yr-old children.* J Appl Physiol, 1999. 86(4): p. 1428-1435.
- 165. Tudor-Locke, C., et al., *How many steps/day are enough? For children and adolescents*. Int J Behav Nutr Phys Act., 2011. (8)78: epub doi:10.1186/1479-5868-8-78.
- 166. Tudor-Locke, C., W.D. Johnson, and P.T. Katzmarzyk, *Accelerometer-determined steps per day in US children and youth*. Medicine and Science in Sports and Exercise, 2010.
 42(12): p. 2244-2250.
- Barreira, T.V., et al., *Cadence patterns and peak cadence in US children and adolescents: NHANES, 2005-2006.* Med Sci Sports Exerc, 2012. 44(9): p. 1721-1727.
- Tudor-Locke, C., et al., A step-defined sedentary lifestyle index: < 5000 steps/day. Appl
 Physiol Nutr Metab, 2012. 38(2): p. 100-114.
- 169. Strong, W.B., et al., *Evidence based physical activity for school-age youth*. J. Pediatr., 2005. 146(6): p. 732-737.
- 170. Bravata, D.M., et al., *Using pedometers to increase physical activity and improve health: A systematic review.* JAMA, 2007. **298**(19): p. 2296-3204.
- 171. Tudor-Locke, C. and L. Lutes, Why do pedometers work? A reflection upon the factors related to successfully increasing physical activity. Sports Med, 2009. 39(12): p. 981-993.

- 172. Scott, J.J., et al., *Adolescent pedometer protocols: examining reactivity, tampering and participants' perceptions.* J Sports Sci, 2014. **32**(2): p. 183-190.
- 173. Tudor-Locke, C. and D.R. Bassett, Jr., *How many steps/day are enough? Preliminary pedometer indices for public health.* Sports Med, 2004. **34**(1): p. 1-8.
- Barfield, J.P., D.A. Rowe, and T. Michael, *Interinstrument consistency of the Yamax Digi-Walker in elementary school children*. Meas Phys Educ Exerc Sci, 2004. 8(2): p. 109-116.
- 175. Carnethon, M.R., M. Gulati, and P. Greenland, *Prevalence and cardiovascular disease correlates of low cardiorespiratory fitness in adolescents and adults*. JAMA, 2005.
 294(23): p. 2981-2988.
- 176. Nielsen, G.A. and L.B. Andersen, *The association between high blood pressure*, *physical fitness, and body mass index in adolescents*. Preventive Medicine, 2003. 36(2):
 p. 229-234.
- Andersen, L.B., et al., *Physical activity and clustered cardiovascular risk in children: a cross-sectional study (The European Youth Heart Study)*. Lancet, 2006. 368(9532): p. 299-304.
- 178. Eisenmann, J., et al., Utility of pedometer step recommendations for predicting overweight in children. Int J Obes, 2007. 31(7): p. 1179-1182.
- 179. Dencker, M., et al., *Daily physical activity related to body fat in children aged 8-11 years*. J. Pediatr., 2006. 149(1): p. 38-42.
- 180. Duncan, M.J., et al., *The relationship between pedometer-determined physical activity, body mass index and lean body mass index in children*. Int J Pediatr Obes, 2010. 5(5):
 p. 445-450.
- 181. McCormack, G.R., et al., *BMI-referenced cut-points for recommended daily pedometerdetermined steps in Australian children and adolescents*. Res Q Exerc Sport, 2011.
 82(2): p. 162-167.
- 182. Tudor-Locke, C., et al., *BMI-referenced standards for recommended pedometerdetermined steps/day in children.* Prev Med, 2004. **38**(6): p. 857-864.

154.

- 183. Lubans, D.R. and P.J. Morgan, Evaluation of an extra-curricular school sport program promoting lifestyle and lifetime activity. J Sports Sci, 2008. 26(5): p. 519-529.
- 184. Le Masurier, G.C. and C.B. Corbin, *Step counts among middle school students vary with aerobic fitness level.* Res Q Exerc Sport, 2006. **77**(1): p. 14-22.
- 185. Rowlands, A.V., R.G. Eston, and D.K. Ingledew, *Relationship between activity levels*, aerobic fitness, and body fat in 8- to 10-yr-old children. J Appl Physiol, 1999. 86(4): p. 1428-1435.
- Pate, R.R., J.R. O'Neill, and F. Lobelo, *The evolving definition of sedentary*. Exerc.
 Sport Sci. Rev., 2008. 36(4): p. 173-178.
- 187. Matthews, C.E., et al., Amount of time spent in sedentary behaviors in the United States, 2003–2004. Am J Epidemiol, 2008. 167(7): p. 875-881.
- 188. De Greef, K., et al., A cognitive-behavioural pedometer-based group intervention on physical activity and sedentary behaviour in individuals with type 2 diabetes. Health Educ Res, 2010. 25(5): p. 724-736.
- 189. Craig, C.L., C. Cameron, and C. Tudor-Locke, *CANPLAY pedometer normative reference data for 21,271 children and 12,956 adolescents*. Med Sci Sports Exerc, 2013. 45(1): p. 123-129.
- 190. Tudor-Locke, C., et al., *Revisiting How Many Steps are Enough?*. Med Sci Sports Exerc, 2008. 40(7): p. S537-543.
- 191. Craig, C.L., et al., *Descriptive epidemiology of youth pedometer-determined physical activity: CANPLAY.* Med Sci Sports Exerc, 2010. **42**(9): p. 1639-1643.
- 192. Richardson, C.R., et al., *A meta-analysis of pedometer-based walking interventions and weight loss*. Ann Fam Med, 2008. **6**(1): p. 69-77.
- 193. Lubans, D.R., et al., *Effects of integrating pedometers, parental materials, and email support within an extracurricular school sport intervention.* J Adolesc Health, 2009.
 44(2): p. 176-83.
- 194. Horne, P.J., et al., *Increasing children's physical activity: A peer modelling, rewards and pedometer-based intervention*. Eur J Clin Nutr, 2009. **63**(2): p. 191-198.

- 195. Schofield, L., W.K. Mummery, and G. Schofield, *Effects of a controlled pedometerintervention trial for low-active adolescent girls*. Med Sci Sports Exerc, 2005. **37**(8): p. 1414-1420.
- 196. Goldfield, G.S., et al., *Effects of open-loop feedback on physical activity and television viewing in overweight and obese children: A randomized, controlled trial.* Pediatrics 2006. 118: p. e157-e166 doi: 10.1542/peds.2005-3052.
- Southard, D.R. and B.H. Southard, *Promoting physical activity in children with MetaKenkoh*. Clin Invest Med, 2006. **29**(5): p. 293-297.
- 198. Michie, S. and C. Abraham, *Interventions to change health behaviours: Evidence-based or evidence-inspired?* Psychol Health, 2004. **19**(1): p. 29-49.
- 199. Dishman, R.K., et al., *Worksite physical activity interventions*. Am J Prev Med, 1998.
 15(4): p. 344-361.
- 200. Hardeman, W., et al., A causal modelling approach to the development of theory based behavior change programmes for trial evaluation. Health Educ Res, 2005. 20(6): p. 676-687.
- 201. Lubans, D.R. and P.J. Morgan, Impact of an extra-curricular school sport program on determinants of objectively measured physical activity among adolescents. Health Educ. J, 2008. 67(4): p. 305-319.
- 202. Lubans, D.R., et al., *Exploring the mechanisms of physical activity and dietary behavior change in the Program X intervention for adolescents*. J Adolesc Health, 2010. 47(1): p. 83-91.
- 203. Zizzi, S., et al., *Impact of a three-week pedometer intervention on high school students' daily step counts and perceptions of physical activity*. Am J Health Educ., 2006. 37(1):
 p. 35-40.
- 204. Lubans, D.R., C. Foster, and S.J.H. Biddle, *A review of mediators of behavior in interventions to promote physical activity among children and adolescents*. Prev Med, 2008. 47(5): p. 463-470.

- 205. van Stralen, M.M., et al., Exploring the efficacy and moderators of two computertailored physical activity interventions for older adults: a randomized controlled trial. Ann Behav Med, 2010. 39(2): p. 139-150.
- Brown, H., et al., A systematic review of intervention effects on potential mediators of children's physical activity. BMC Public Health, 2013. 13(165) epub doi: 10.1186/1471-2458-13-165.
- 207. Tsiros, M.D., et al., *Cognitive behavioral therapy improves diet and body composition in overweight and obese adolescents*. Am J Clin Nutr, 2008. **87**(5): p. 1134-1140
- Berry, D., et al., *An intervention for multiethnic obese parents and overweight children*.
 Appl Nurs Res, 2007. 20(2): p. 63-71.
- 209. Deci, E.L. and R.M. Ryan, *Handbook of Self-determination Research*. 2002, Rochester, New York: University of Rochester Press.
- Goldfield, G.S., et al., *Open-loop feedback to increase physical activity in obese children*. Int J Obes Relat Metab Disord, 2000. 24(7): p. 888-892.
- 211. Biddle, S. and N. Mutrie, *Psychology of physical activity: determinants, well-being and interventions*. 2001, London: Routledge.
- 212. Santrock, J.W., Adolesence. 2005, Boston: McGraw-Hill.
- 213. Lubans, D.R., et al., *Description and evaluation of a social cognitive model of physical activity behaviour tailored for adolescent girls*. Health Educ Res, 2011. 27(1): p. 115-128.
- Rhodes, R.E. and C.R. Nigg, Advancing physical activity theory: a review and future directions. Exerc. Sport Sci. Rev., 2011. 39(3): p. 113-119.
- 215. Sallis, J.F., K. Kraft, and L.S. Linton, *How the environment shapes physical activity: A transdisciplinary research agenda*. Am J Prev Med, 2002. 22(3): p. 208. epub doi: 10.1016/S0749-3797(01)00435-4
- Welk, G.J., *The youth physical activity promotion model: a conceptual bridge between theory and practice*. Quest, 1999. **51**(1): p. 5-23.

- 217. Weiss, M.R., *Motivating kids in physical activity*. Pres Counc Phys Fit Sports Res Dig, 2000. 3(11): p. 1-8.
- 218. Norman, G.J., et al., *A review of eHealth interventions for physical activity and dietary behavior change*. Am J Prev Med, 2007. **33**(4): p. 336-345.
- 219. Weerakkody and Niranjala D., *Mobile phones and children : an Australian perspective*, in *INSITE 2008 : Proceedings of the 2008 informing science + information technology education conference, informing science institute*. 2008: Varna, Bulgaria. p. 459-475.
- Lenhart, A., *Teens, smartphones & texting*. 2012, Pew Internet and American Life Project: Washington, DC.
- 221. Wiehe, S.E., et al., *Using GPS-enabled cell phones to track the travel patterns of adolescents*. Int J Health Geogr., 2008. **7**(22): epub doi:10.1186/1476-072X-7-22.
- 222. Lubans, D.R., et al., *Preventing obesity among adolescent girls: One-year outcomes of the Nutrition and Enjoyable Activity for Teen Girls (NEAT Girls) cluster randomized controlled trial.* Arch Pediatr Adolesc Med, 2012. **166**(9): p. 821-827.
- 223. Dewar, D., et al., *The Nutrition and Enjoyable Activity for Teen Girls study: A cluster randomized controlled trial.* Am J Prev Med, 2013. **45**(3): p. 313-317.
- 224. Dewar, D., et al., *Exploring changes in physical activity, sedentary behaviors and hypothesized mediators in the NEAT girls group randomized controlled trial.* J. Sci. Med. Sport, 2014. 17(1): p. 39-46.
- 225. Smith, J.J., et al., *Rationale and study protocol for the 'Active Teen Leaders Avoiding Screen-time' (ATLAS) group randomized controlled trial: An obesity prevention intervention for adolescent boys from schools in low-income communities.* Contemp Clin Trials, 2013 **37**(1): p. 106-119.
- Thompson, D., et al., *Texting to increase physical activity among teens (TXT Me!): Rationale, design, and methods*. Contemp Clin Trials, 2014 3(1) epub doi: 10.2196/resprot.3074.

- 227. Centers for Disease Control & Prevention, *School health guidelines to promote healthy eating and physical activity*. Morbidity & Mortality Weekly Report, 2011. 60(5): p. 1-76.
- 228. Lonsdale, C., et al., A systematic review and meta-analysis of interventions designed to increase moderate-to-vigorous physical activity in school physical education lessons.
 Prev Med, 2013. 56(2): p. 152-161.
- 229. Kriemler, S., et al., Effect of school based physical activity programme (KISS) on fitness and adiposity in primary schoolchildren: cluster randomised controlled trial. Br Med J 2010. 340: epub doi: 10.1136/bmj.c785.
- Scruggs, P., Middle school physical education physical activity quantification: A pedometer steps/min guideline. Res Q Exerc Sport, 2007. 78(4): p. 284-292.
- 231. Scruggs, P., *Pedometer steps/min in physical education: Does the pedometer matter?* J Sci Med Sport, 2013. 16(1): p. 36-39.
- 232. Riley, N., P. Morgan, and D.R. Lubans, *Preliminary findings of the E.A.S.Y.*(*Encouraging Activity to Stimulate Young*) minds feasibility study: A curriculum-based physical activity integration program in the primary school. J Sci Med Sport, 2012. 15
 (Suppl 1): p. S90, epub doi: 10.1016/j.jsams.2012.11.217.
- 233. Stewart, J.A., et al., *Exercise level and energy expenditure in the TAKE 10! in-class physical activity program.* J Sch Health, 2004. **74**(10): p. 397-400.
- 234. Mahar, M.T., et al., *Effects of a classroom-based program on physical activity and ontask behavior*. Med Sci Sports Exerc, 2006. **38**(12): p. 2086-2094.
- Ridgers, N.D., G. Stratton, and S.J. Fairclough, *Physical activity levels of children during school playtime*. Sports Med, 2006. 36(4): p. 359-371.
- Beets, M.W., et al., *After-school program impact on physical activity and fitness: a meta-analysis.* Am J Prev Med, 2009. 36(6): p. 527-537.
- Bassett, D.R., et al., *Estimated energy expenditures for school-based policies and active living*. Am J Prev Med, 2013. 44(2): p. 108-113.

- 238. Van Acker, R., et al., *The association between socio-ecological factors and having an after-school physical activity program.* J Sch Health, 2012. **82**(9): p. 395-403.
- 239. Nichol, M.E., W. Pickett, and I. Janssen, *Associations between school recreational* environments and physical activity. J Sch Health, 2009. **79**(6): p. 247-254.
- Lubans, D.R., et al., Rationale and study protocol for the Supporting Children's Outcomes using Rewards, Exercise and Skills (SCORES) group randomized controlled trial: A physical activity and fundamental movement skills intervention for primary schools in low-income communities. BMC Public Health, 2012. 12: epub doi: 10.1186/1471-2458-12-427.
- Huberty, J., M. Beets, and A. Beighle, *Effects of a policy-level intervention on children's pedometer-determined physical activity: preliminary findings from Movin' Afterschool.* J Public Health Manag Pract., 2013. 19(8) p. 525-528.
- 242. Haskell, W., *Physical activity by self-report: A brief history and future issues.* J Phys Act Health, 2012. **9**(Suppl 1): p. S5-10.
- 243. Sirard, J.R., et al., *Evaluation of two self-report measures of physical activity with accelerometry in young adults.* J Phys Act Health, 2013. **10**(1): p. 85-96.
- 244. Cliff, D.P., J.J. Reily, and A.D. Okely, *Methodological considerations in using accelerometers to assess habitual physical activity in children aged 0–5 years*. J Sci Med Sport., 2009. **12**(12): p. 557-567.
- 245. Adams, S.A., et al., *The effect of social desirability and social approval on self-reports of physical activity*. Am J Epidemiol, 2005. **161**(4): p. 389-398.
- 246. Baranowski, T., *Validity and reliability of self report measures of physical activity: An information-processing perspective.* Research Quarterly for Exercise and Sport, 1988.
 59: p. 314-327.
- 247. Hardy, L.L., et al., *A hitchhiker's guide to assessing sedentary behaviour among young people: Deciding what method to use.* J Sci Med Sport, 2013. **16**(1): p. 28-35.

- 248. Jackson, A.W., J.R. Morrow, and H.R. Bowles, *Construct validity evidence for singleresponse items to estimate physical activity levels in large sample studies.* Res Q Exerc Sport, 2007. **78**(2): p. 24-31.
- 249. Iwai, N., S. Hisamichi, and N. Hayakawa, *Validity and reliability of single-item questions about physical activity*. J Epidemiol., 2001. **11**(5): p. 211-218.
- Li, S., E. Carlson, and K. Holm, Validation of a single-item measure of usual physical activity. Percept Mot Skills., 2000. 91(2): p. 593-602.
- 251. Smith, B.J., A.L. Marshall, and N. Huang, *Screening for physical activity in family practice: evaluation of two brief assessment tools*. Am J Prev Med, 2005. **29**(4): p. 256-264.
- 252. Beets, M.W., *The pursuit of the reactivity: A need for a closer look*. Med Sci Sports Exerc, 2006. 10(4): p. 265-267.
- 253. Machado-Rodrigues, A.M., et al., Concurrent validation of estimated activity energy expenditure using a 3-day diary and accelerometry in adolescents. Scand J Med Sci Sports, 2012. 22(2): p. 259-264.
- 254. Telford, A., J. Salmon, and D. Jolley, *Reliability and validity of physical activity questionniares for children: The children's leisure activities study survey.* Pediatr Exerc Sci, 2004. 16(3): p. 64-78.
- 255. Duncan, J.S., et al., *Effects of age, walking speed, and body composition on pedometer accuracy in children.* Res Q Exerc Sport, 2007. **78**(5): p. 420-428.
- 256. Edwardson, C.L. and T. Gorley, *Epoch length and its effect on physical activity intensity*. Med Sci Sports Exerc, 2009. **42**(5): p. 928-934.
- 257. Anderson, C.B., M. Hagstromer, and A. Yngve, Validation of the PDPAR as an adolescent diary: effect of accelerometer cut points. Med Sci Sports Exerc, 2005. 37(7): p. 1224-1230.
- 258. Matthews, C.E., et al., *Best practices for using physical activity monitors in populationbased research.* Med Sci Sports Exerc, 2012. **44**(1): p. 68-76.

- 259. Riddoch, C.J., L.B. Anderson, and N. Wedderkopp, *Physical activity levels and patterns of 9- and 15-yr-old European children*. Medical Science Sports Exercise, 2004.
 36: p. 86-92.
- Rowlands, A.V., Accelerometer Assessment of Physical Activity in Children: An Update. Pediatr Exerc Sci , 2007. 19(3): p. 252-266.
- 261. Ridgers, N.D., et al., Validity of a breif self-report instument for assessing compliance with physical activity guidelines amongst adolescents. J Sci Med Sport, 2012. **15**(2): p. 136-141.
- Troped, P.J., et al., *Reliability and validity of YRBS physical activity items among middle school students*. Med Sci Sports Exerc, 2007. **39**(3): p. 416-425.
- 263. Economos, C.D., et al., *Development and testing of the BONES physical activity survey for young children*. BMC Musculoskelet Disord., 2010. 11(195) epub doi: 10.1186/1471-2474-11-195.
- 264. Ridley, K., B.E. Ainsworth, and T.S. Olds, *Development of a compendium of energy expenditures for youth*. Int J Behav Nutr Phys Act., 2008. 10(5): epub doi: 10.1186/1479-5868-5-45.
- 265. Ridley, K. and T.S. Olds, Assigning energy costs to activities in children: a review and synthesis. Med Sci Sports Exerc, 2008. 40(8): p. 1439-1446.
- 266. Hopkins, W.G., *Measures of reliability in sports medicine and science*. Sports Med, 2000. 30(1): p. 1-15.
- 267. Ridley, K., T.S. Olds, and A. Hill, *The multimedia activity recall for children and adolescents (MARCA): development and evaluation*. Int J Behav Nutr Phys Act, 2006.
 26(3): epub doi: 10.1186/1479-5868-3-10.
- 268. Welk, G.J., C.B. Corbin, and D. Dale, *Measurement issues in the assessment of physical activity in children*. Res Q Exerc Sport, 2000. **71**(Suppl 2): p. S59-73.
- 269. Trost, S., Objective measurement of physical activity in youth: current issues future directions. Exerc Sport Sci Rev., 2001. 29(1): p. 32-36.

- 270. Behrens, T. and M. Dinger, *Motion sensor reactivity in physically active young adults*.Res Q Exerc Sport., 2007. 78(2): p. 1-8.
- Eastep, E., et al., *Does augmented feedback from pedometers increase adults walking behaviour?* Percept Mot Skills., 2004. 99(2): p. 392-402.
- 272. Tudor-Locke, C. and A.M. Myers, *Methodological considerations for researchers and practitioners using pedometers to measure physical (ambulatory) activity.* Res Q Exerc Sport., 2001. **72**(1): p. 1-12.
- 273. Kang, M., et al., An experimental determination of the best missing value recovery method in assessing physical activity using pedometers. Res Q Exerc Sport., 2003.
 74(1): p. 1-25.
- 274. Mattocks, C., S. Leary, and A. Ness, *Intraindividual variation of objectively measured physical activity in children*. Med Sci Sports Exerc, 2007. **39**(4): p. 622-629.
- 275. Tudor-Locke, C., et al., *Comparison of pedometer and accelerometer measures of freeliving physical activity*. Med Sci Sports Exerc, 2002. **34**(12): p. 2045-2051.
- 276. Cuberek, R., et al., A comparison of two motion sensors for the assessment of freeliving physical activity of adolescents, Int J Environ Res Public Health., 2010. 7(4): p. 1558-1576.
- 277. Lubans, D.R., et al., *Using pedometers for measuring and increasing physical activity in children and adolescents: The next step.* Am J Lifestyle Med, 2015. **9**(6): p. 418-427.
- 278. Smith, J.J., et al., *Smart-phone obesity prevention trial for adolescent boys in lowincome communities: The ATLAS RCT.* Pediatrics, 2014. **134**(3): p. 723-731.
- 279. Robertson, W., et al., Utility of accelerometers to measure physical activity in children attending an obesity treatment intervention. J Obes, 2011. Article ID 398918 epub doi: 10.1155/2011/398918.
- 280. Pearson, N., et al., *Patterns of adolescent physical activity and dietary behaviours*. Int J Behav Nutr Phys Act., 2009. 6(45): epub doi: 10.1186/1479-5868-6-45.

- 281. Bergier, J., et al., *Physical activity of Polish adolescents and young adults according to IPAQ: a population based study.* Ann Agric Environ Med, 2012. **19**(1): p. 109-115.
- 282. Sanchez, A., et al., Patterns and correlates of physical activity and nutrition behaviors in adolescents. Am J Prev Med, 2007. 32(2): p. 124-30.
- 283. Telford, R.M., et al., Why Are Girls Less Physically Active than Boys? Findings from the LOOK Longitudinal Study. PLoS ONE, 2016. 11(3): p. e0150041, epub doi: 10.1371/journal.pone.0150041.
- Evenson, K.R., et al., *Calibration of two objective measures of physical activity for children*. J Sports Sci, 2008. 26(14): p. 1557-1565.
- 285. Rice, P.L. and D. Ezzy, eds. *Qualitative research methods: a health focus*. 1999, Melbourne: Oxford University Press.
- 286. Peterson-Sweeney, K., *The use of focus groups in pediatric and adolescent research*. J
 Pediatr Health Care., 2004. **19**(2): p. 104-110.
- 287. Kitzinger, J., Focus group research: Using group dynamics to explore perceptions, experiences and understandings. Qualitative research in health care, I.I.H. (ed.). 2005, Maidenhead: Open University Press: p. 56-66.
- Alasuutari, P., et al., *Focus groups*. The Sage handbook of social research methods, first ed. 2008, London: Sage.
- 289. Hsieh, H.F. and S.E. Shannon, *Three approaches to qualitative content analysis*. Qual Health Res, 2005. 15(9): p. 1277-1288
- 290. Hyde, K.F., *Recognising deductive processes in qualitative research*. QMR: Int J.,
 2000. 3(2): p. 82-90.
- 291. Fereday, J. and E. Muir-Cochrane, *Demonstrating rigor using thematic analysis: a hybrid approach of inductive and deductive coding and theme development.* "Int J Qual Methods, 2006. 5(1): p. 80-92.
- Zhang, S., et al., *Physical activity classification using the GENEA wrist-worn accelerometer*. Med Sci Sports Exerc, 2012. 44(4): p. 742-748.

- 293. Pavey, T.G., et al., *The validity of the GENEActiv wrist-worn accelerometer for measuring adult sedentary time in free living*. J Sci Med Sport, 2015. 19(5): p. 395-399.
- 294. Scott, J.J., et al., *Comparability and feasibility of wrist- and hip-worn accelerometers in free-living adolescents*. J Sci Med Sport, 2017. [epub ahead of print] doi: 10.1016/j.jsams.2017.04.017.
- 295. Lee, P.H., D.J. Macfarlane, and T.H. Lam, *Factors associated with participant compliance in studies using accelerometers*. Gait Posture, 2013. **38**(4): p. 912-917.
- 296. Van Coevering, P., et al., *Feasibility of using accelerometers to measure physical activity in young adolescents*. Med Sci Sports Exerc, 2005. **37**(5): p. 867-871.
- 297. Belton, S., et al., *Patterns of noncompliance in adolescent field-based accelerometer research.* J Phys Act Health, 2013. **10**(8): p. 1181-1185.
- 298. Sutherland, R., et al., '*Physical activity 4 everyone' school-based intervention to prevent decline in adolescent physical activity levels: 12 month (mid-intervention) report on a cluster randomised trial.* Br J Sports Med, 2016. **50**(8): p. 488-495.
- 299. Phillips, L.R.S., G. Parfitt, and A.V. Rowlands, *Calibration of the GENEA* accelerometer for assessment of physical activity intensity in children. J Sci Med Sport, 2013. 16(2): p. 124-128.
- Freedson, P.S. and D. John, *Comment on "estimating activity and sedentary behavior from an accelerometer on the hip and wrist"*. Med Sci Sports Exerc, 2013. 45(5): p. 962-963.
- 301. Tudor-Locke, C., et al., Improving wear time compliance with a 24-hour waist-worn accelerometer protocol in the International Study of Childhood Obesity, Lifestyle and the Environment (ISCOLE). Int J Behav Nutr Phys Act, 2015. 12(11). epub doi: 10.1186/s12966-015-0172-x.
- 302. van Hees, V.T., et al., Separating movement and gravity components in an acceleration signal and implications for the assessment of human daily physical activity. PLoS ONE, 2013. 8(4): p. e61691, epub doi: 10.1371/journal.pone.0061691.

- 303. Rich, C., et al., *Quality control methods in accelerometer data processing: defining minimum wear time*. PloS One, 2013. 8(6): p. e67206, epub doi: 10.1371/journal.pone.006720
- Treuth, M.S., et al., *Defining accelerometer thresholds for activity intensities in adolescent girls*. Med Sci Sports Exerc, 2004. 36(7): p. 1259-1266.
- 305. Kamada, M., et al., *Comparison of physical activity assessed using hip- and wrist-worn accelerometers*. Gait Posture, 2016. **44**: p. 23-28.
- 306. Kerr, J., et al., *Comparison of accelerometry methods for estimating physical activity*.
 Med Sci Sports Exerc, 2016. 49(3) p. 617-624
- 307. Hislop, J., et al., Validity of wrist worn accelerometers and comparability between hip and wrist placement sites in estimating physical activity behaviour in preschool children. Physiol Meas, 2016. **37**(10): p. 1701-1714.
- 308. Rowlands, A.V. and V.H. Stiles, *Accelerometer counts and raw acceleration output in relation to mechanical loading*. J Biomech, 2012. **45**(3): p. 448-454.
- 309. Crouter, S.E., J.I. Flynn, and D.R. Bassett, Jr., *Estimating physical activity in youth using a wrist accelerometer*. Med Sci Sports Exerc, 2015. **47**(5): p. 944-951.
- 310. Brooke, H.L., et al., *A systematic literature review with meta-analyses of within- and between-day differences in objectively measured physical activity in school-aged children.* Sports Med, 2014. **44**(10): p. 1427-1438.
- 311. Davison, K.K., et al., Why are early maturing girls less active? Links between pubertal development, psychological well-being, and physical activity among girls at ages 11 and 13. Soc Sci Med, 2007. 64(12): p. 2391-2404.
- Wanner, M., et al., *What physical activity surveillance needs: validity of a single-item questionnaire*. Br J Sports Med, 2012. 48(21): p. 1570-1576
- 313. Maher, C., et al., *The reliability and validity of a research-grade pedometer for children and adolescents with cerebral palsy*. Dev Med Child Neurol, 2013. **55**(9): p. 827-833.

- Ainsworth, B.E., et al., *Recommendations to improve the accuracy of estimates of physical activity derived from self report.* J Phys Act Health, 2012. 9(Suppl 1): p. S76-84.
- 315. Scientific Advisory Committee of the Medical Outcomes Trust, *Assessing health status and quality-of-life instruments: attributes and review criteria.* Qual Life Res, 2002.
 11(3): p. 193-205.
- Ridley, K., et al., *Intra-individual variation in children's physical activity patterns: Implications for measurement.* J Sci Med Sport, 2009. 12(5): p. 568-572.
- 317. Ridgers, D.N., A.M. McNarry, and A.K. Mackintosh, *Feasibility and effectiveness of using wearable activity trackers in youth: A systematic review*. JMIR Mhealth Uhealth, 2016. 4(4): p. e129, epub doi: 10.2196/mhealth.6540.
- 318. Schaefer, S.E., et al., *Wearing, thinking, and moving: testing the feasibility of fitness tracking with urban youth.* J Health Educ, 2016. **47**(1): p. 8-16.
- 319. Schaefer, S.E., M. Van Loan, and J.B. German, A feasibility study of wearable activity monitors for pre-adolescent school-age children. Prev Chronic Dis., 2014. 11(e85): epub doi: 10.5888/pcd11.130262.
- 320. Freedson, P., et al., Assessment of physical activity using wearable monitors:
 recommendations for monitor calibration and use in the field. Med Sci Sports Exerc.,
 2012. 44(Supp 1): p. S1-4.
- 321. Goran, M.I. and M. Sun, *Applications of the doubly labeled water technique for studying total energy expenditure in young children: a review.* Pediatr Exerc Sci, 1994.
 6(1): p. 11-30.
- 322. Livingstone, M.B., et al., Daily energy expenditure in free-living children: comparison of heart-rate monitoring with the doubly labeled water (2H2(18)O) method. Am J Clin Nutr, 1992. 56(2): p. 343-352.
- 323. Arvidsson, D., F. Slinde, and L. Hulthen, *Physical activity questionnaire for adolescents validated against doubly labelled water*. Eur J Clin Nutr, 2005. **59**(3): p. 376-383.

- 324. Sjoberg, A., et al., *Energy intake in Swedish adolescents: validation of diet history with doubly labelled water*. Eur J Clin Nutr, 2003. **57**(12): p. 1643-1652.
- 325. Brown, H., C. Hume, and A.M. Chin, *Validity and reliability of instruments to assess potential mediators of children's physical activity: A systematic review.* J Sci Med Sport, 2009. **12**(5): p. 539-548.
- 326. Nichols, R., et al., *The use of heart rate monitors in physical education*. Strategies, 2009. 22(6): p. 19-23.
- 327. Janz, K.F., et al., Heart rate monitoring of physical activity in children and adolescents: the Muscatine Study. Pediatrics, 1992. 89(2): p. 256-261.
- 328. Kaikkonen, K.M., et al., *Physical activity and aerobic fitness are positively associated with heart rate variability in obese adults*. J Phys Act Health, 2014. **11**(8): p. 1614-1621.

9 Appendices

Appendix 1: Thesis publication guidelines
Appendix 2: University of Newcastle ethics approval letter
Appendix 3: Maitland-Newcastle Catholics schools' ethics approval letter176
Appendix 4: Funding acknowledgement: Priority research centre for physical activity and
nutrition (PRCPAN) seed funding grant approval letter
Appendix 5: Measuring activity with energy expenditure: The strengths and limitations of
doubly labelled water and heart-rate monitoring178
Appendix 6: Parent information sheet and consent form
Appendix 7: Student (participant) information sheet and consent form
Appendix 8: Accelerometer log sheet: This sheet was used by participants to log non-wear
time of measuring devices (i.e., swimming, contact sports etc.)190
Appendix 9: Pedometer log books: This was used by the participants to record their daily
step counts
Appendix 10: Oxford Physical Activity Questionnaire (OPAQ). Previous validated
questionnaire in the adolescent population (Lubans, 2008)
Appendix 11: Single-item physical activity questionnaire: A single-item questionnaire for
assessing physical activity (modified from Milton, et al., 2010)
Appendix 12: Pedometry Behaviour Questionnaire (PBQ)
Appendix 13: Accelerometry Behaviour Questionnaire (ABQ
Appendix 14: Focus-group questions

Appendix 1: The University of Newcastle thesis by publication guidelines

Office of Graduate Studies Information Sheet Thesis by Publication



A thesis may be submitted in the form of a series of published papers and the additional rules specific to this style of thesis are presented below. It is important to note that the general rules for a University of Newcastle thesis are also applicable. Please ensure you also refer to <u>The Rules</u> <u>Governing Research Higher Degrees</u> for the full scope of applicable rules.

Rule 39.1 A thesis by publication will include:

- a full explanatory overview that links the separate papers and places them in the context of an established body of knowledge;
- ii. a literature review;
- iii. if detailed data and descriptions of methods are not otherwise given within the separate papers, they must be included in the body of the thesis or as appendices to the thesis;

Rule 39.2 For a thesis by publication:

- i. the separate papers provided under sub-clause 39.1(i) must be published, in press or submitted to scholarly media only, i.e. refereed publications classified by current national standards and refereed conference papers, however at least 50% of these papers must have been published. Papers published up to three years prior to enrolment may be included provided they were published in scholarly media and do not represent more than 50% of the total papers;
- publications submitted by the candidate for another degree may only be referred to in the thesis literature review;
- iii. the number of papers submitted should demonstrate that the body of work meets the requirements of the degree as outlined in the relevant schedule;
- iv. the candidate must be the lead author in at least 50% of the papers written in the time of their formal Research Higher Degree candidature. Any published paper of which the candidate is a joint author may only be included in the thesis provided the work done by the candidate is clearly identified. The candidate must include in the thesis a written statement from each co-author attesting to the candidate's contribution to a joint publication included as part of the thesis. These statements must be endorsed by the Assistant Dean (Research Training).
- the Assistant Dean (Research Training) may seek the approval of the Dean of Graduate Studies to include a paper that is outside the scope of these rules.

Office of Graduate Studies, East Wing, The Chancellery Telephone: (02) 4921 6537 Fax: (02) 4921 6908 Email: research@newcastle.edu.au

Considerations

- Each discipline area will have different issues to consider in the decision to submit a thesis in the form of a series of published papers.
- It is essential that you discuss your options carefully with your supervisor(s). The thesis by
 publication must reflect a sustained and cohesive theme, an integrated whole that sits logically
 in the context of the available literature. Overall the material presented for examination needs
 to equate to that which would otherwise be presented in the traditional thesis format.
- The review process for some journals is significant resulting in lengthy waiting periods for
 papers to be accepted and this can delay thesis submission/completion. Time management
 and selection of journals/publishers is critical. Focusing on publication rather than research
 may lead to candidates being tempted to publish sections of their work prematurely and
 missing opportunities to fully capitalize on the significance of the work.
- Consider the thesis from the examiners' view point if the publications do not have a clear cohesion and the contribution to knowledge is not clearly demonstrated, then the thesis may attract criticism and be rejected by examiners. The content of the thesis remains a matter of professional judgment for the supervisor(s) and candidate.
- Any published paper of which the candidate is a joint author may only be included in the thesis
 provided the work done by the candidate is clearly identified. The candidate must include in the
 thesis a written statement from each co-author attesting to the candidate's contribution to a
 joint publication included as part of the thesis. The statement/s need to be signed by the
 Faculty Assistant Dean (Research Training). A sample statement is provided below.
- We strongly advise that you arrange for the signatures from co-authors to be collected as soon
 as the paper is prepared or submitted for publication rather than trying to collect them at the
 time of thesis submission.
- There is no minimum or maximum requirement on the number of papers. Of equal, or perhaps more importance than quantity, is the quality of the journals. Please refer to your school or faculty for more specific guidance on the number and length of papers that would normally be expected in your discipline.

Alternative option

As discussed above, you need to consider if your publications will form a sufficient body of cohesive work to meet the requirements of thesis by publication. You may like to consider the other option of including publications within a standard thesis format, either in the body or as an appendix as supported in the rule below.

Rule 38.5. A thesis may:

i. Include publications arising as a consequence of the research undertaken for a thesis. When the candidate includes a co-authored published paper or co-authored scholarly work, or a substantive component of a co-authored published paper or co-authored scholarly work in the body of the thesis, the candidate must include in the thesis a written statement attesting to their contribution to the joint publication. This statement must be signed by the supervisor. A statement is not required when publications are included as an appendix to the thesis.

Components and Layout

PLEASE NOTE: the layout and ordering of the contents is flexible and should be based on the judgement and experience of candidates and supervisors as well as discipline norms. Please use your own discretion and seek expert advice. The following is a <u>suggested</u> layout only.

1. Title Page

2. Declarations

Originality

I hereby certify that to the best of my knowledge and belief this thesis is my own work and contains no material previously published or written by another person except where due references and acknowledgements are made. It contains no material which has been previously submitted by me for the award of any other degree or diploma in any university or other tertiary institution.

Thesis by Publication

I hereby certify that this thesis is in the form of a series of *papers. I have included as part of the thesis a written statement from each co-author, endorsed in writing by the Faculty Assistant Dean (Research Training), attesting to my contribution to any jointly authored papers. (*Refer to clause 39.2 of the Rules Governing Research Higher Degrees for acceptable papers).

3. Acknowledgements

4. List of publications included as part of the thesis

4.1 List all of the included published work with the full bibliographic citations in the order they appear in the thesis.

4.2 Provide a statement to indicate that where necessary permission regarding copyright has been obtained from copyright owners. For example, the statement may say "I warrant that I have obtained, where necessary, permission from the copyright owners to use any third party copyright material reproduced in the thesis (e.g. questionnaires, artwork, unpublished letters), or to use any of my own published work (e.g. journal articles) in which the copyright is held by another party (e.g. publisher, co-author)."

5. Table of Contents

6. Abstract

An abstract of approximately 300 words is required to describe the content of the thesis.

7. Overview

A full explanatory overview is required to link the published papers to the research thesis. This may include sections for Literature Review (if not included separately), Research Design and Review/Discussion. Not all of these sections may be necessary. Choose the format that underpins the academic argument so that the contents of the thesis are established as a substantial and significant body of work, but without unnecessary repetition.

8. Literature Review

9. Statement of Contribution of Others

In the thesis, at the front of each paper, include a written statement from each co-author attesting to the candidate's contribution to a joint publication included as part of the thesis. The purpose of this statement is to summarise and clearly identify the nature and extent of the intellectual input by the candidate and any co-authors.

9.1 Sample co-author statement

By signing below I confirm that [Candidate Name] contributed [insert outline of contribution]) to the paper/publication entitled [insert reference details].

List:

Full Name of Co-Author/s, Date, Signature of Co-Authors

Full Name of Faculty Assistant Dean Research Training, Date, Signature

10. Papers/Chapters

Each paper/chapter should have an introduction to explain how it contributes to the overall body of knowledge. It is not necessary to reformat published papers in the thesis. Where appropriate publications can be included in full or in parts thereof.

- 11. Appendices
- 12. Bibliography

Appendix 2: University of Newcastle ethics approval letter

HUMAN RESEARCH ETHICS COMMITTEE



Notification of Expedited Approval

To Chief Investigator or Project Supervisor:	Associate Professor David Lubans
Cc Co-investigators / Research Students:	Professor Ronald Plotnikoff Professor Philip Morgan Mr Joseph Scott
Re Protocol:	The measurement of physical activity in adolescents: Testing objective monitoring protocols and validation of self-reporting measures
Date: Reference No: Date of Initial Approval:	28-Jun-2011 H-2011-0137 22-Jun-2011

Thank you for your **Response to Conditional Approval** submission to the Human Research Ethics Committee (HREC) seeking approval in relation to the above protocol.

Your submission was considered under Expedited review by the Chair/Deputy Chair.

I am pleased to advise that the decision on your submission is **Approved** effective **22-Jun-2011**.

For noting: Please provide a copy of the revised risk assessment form for our records, as referred to in your response letter.

In approving this protocol, the Human Research Ethics Committee (HREC) is of the opinion that the project complies with the provisions contained in the National Statement on Ethical Conduct in Human Research, 2007, and the requirements within this University relating to human research.

Approval will remain valid subject to the submission, and satisfactory assessment, of annual progress reports. *If the approval of an External HREC has been "noted" the approval period is as determined by that HREC*.

The full Committee will be asked to ratify this decision at its next scheduled meeting. A formal *Certificate of Approval* will be available upon request. Your approval number is **H-2011-0137**.

If the research requires the use of an Information Statement, ensure this number is inserted at the relevant point in the Complaints paragraph prior to distribution to potential participants You may then proceed with the research.

Conditions of Approval

This approval has been granted subject to you complying with the requirements for *Monitoring of Progress, Reporting of Adverse Events*, and *Variations to the Approved Protocol* as <u>detailed below</u>.

PLEASE NOTE:

In the case where the HREC has "noted" the approval of an External HREC, progress reports and reports of adverse events are to be submitted to the External HREC only. In the case of Variations to the approved protocol, or a Renewal of approval, you will apply to the External HREC for approval in the first instance and then Register that approval with the University's HREC.

• Monitoring of Progress

Other than above, the University is obliged to monitor the progress of research projects involving human participants to ensure that they are conducted according to the protocol as approved by the HREC. A progress report is required on an annual basis. Continuation of your HREC approval for this project is conditional upon receipt, and satisfactory assessment, of annual progress reports. You will be advised when a report is due.

• Reporting of Adverse Events

- 1. It is the responsibility of the person **first named on this Approval Advice** to report adverse events.
- 2. Adverse events, however minor, must be recorded by the investigator as observed by the investigator or as volunteered by a participant in the research. Full details are to be documented, whether or not the investigator, or his/her deputies, consider the event to be related to the research substance or procedure.
- 3. Serious or unforeseen adverse events that occur during the research or within six (6) months of completion of the research, must be reported by the person first named on the Approval Advice to the (HREC) by way of the Adverse Event Report form within 72 hours of the occurrence of the event or the investigator receiving advice of the event.
- 4. Serious adverse events are defined as:
 - Causing death, life threatening or serious disability.
 - Causing or prolonging hospitalisation.
 - Overdoses, cancers, congenital abnormalities, tissue damage, whether or not they are judged to be caused by the investigational agent or procedure.
 - Causing psycho-social and/or financial harm. This covers everything from perceived invasion of privacy, breach of confidentiality, or the diminution of social reputation, to the creation of psychological fears and trauma.
 - Any other event which might affect the continued ethical acceptability of the project.
- 5. Reports of adverse events must include:
 - Participant's study identification number;
 - \circ date of birth;
 - date of entry into the study;
 - treatment arm (if applicable);
 - o date of event;
 - details of event;
 - the investigator's opinion as to whether the event is related to the research procedures; and

- action taken in response to the event.
- 6. Adverse events which do not fall within the definition of serious or unexpected, including those reported from other sites involved in the research, are to be reported in detail at the time of the annual progress report to the HREC.
- Variations to approved protocol

If you wish to change, or deviate from, the approved protocol, you will need to submit an *Application for Variation to Approved Human Research*. Variations may include, but are not limited to, changes or additions to investigators, study design, study population, number of participants, methods of recruitment, or participant information/consent documentation. **Variations must be approved by the (HREC) before they are implemented** except when Registering an approval of a variation from an external HREC which has been designated the lead HREC, in which case you may proceed as soon as you receive an acknowledgement of your Registration.

Linkage of ethics approval to a new Grant

HREC approvals cannot be assigned to a new grant or award (i.e. those that were not identified on the application for ethics approval) without confirmation of the approval from the Human Research Ethics Officer on behalf of the HREC.

Best wishes for a successful project.

Professor Alison Ferguson Chair, Human Research Ethics Committee

For communications and enquiries: Human Research Ethics Administration

Research Services Research Integrity Unit HA148, Hunter Building The University of Newcastle Callaghan NSW 2308 T +61 2 492 18999 F +61 2 492 17164 Human-Ethics@newcastle.edu.au

Linked University of Newcastle administered funding:



Appendix 3: Maitland-Newcastle Catholics schools' ethics approval letter

CATHOLIC SCHOOLS OFFICE

9 August 2011

Mr Joseph Scott PO Box 338 Watatah NSW 2298

Dear Mr Scott,

Thank you for your Application to Conduct Research in the Diocese of Maitland-Newcastle which we received on 15 June 2011. Your research application has been approved. I am happy for you to approach the following high schools in our diocese in order to test objective monitoring protocols in the measurement of physical activity in adolescents:

- San Clemente, Mayfield
- St. Marys, Gateshead
- St. Pius X, Adamstown
- St. Pauls, Booragul
- All Saints College, St. Peters Campus, Maitland

Although the CSO has approved this project it is still up to the the school principal to give final permission for research to be carried out in their school. They are uniquely aware of school specific circumstances and timelines that may make the research proposal difficult to implement.

We also stress the following points in relation to research requests:

- Confidentiality needs to be observed in reporting and must comply with the requirements of the Commonwealth Privacy Amendment (Private Sector) Act 2000.
- There should be some feedback to schools and a copy of the findings of the research forwarded to this office.
- This letter of approval should accompany any approach to schools.

I look forward to the results of this study and wish you the best over the coming months. If you would like to discuss any aspect of this research in our diocese, please do not hesitate to contact me on 02 4979 1201 or vicki.sheriff@mn.catholic.edu.au.

Yours sincerely,

Vicki Sheriff Professional Assistant to the Director Catholic Schools Office Diocese of Maitland-Newcastle

Appendix 4: Funding acknowledgement: Priority research centre for physical activity and nutrition (PRCPAN) seed funding grant approval letter

PRIORITY RESEARCH CENTRE PHYSICAL ACTIVITY & NUTRITION



Wednesday 15th June, 2011

Mr Joseph Scott School of Education Faculty of Education & Arts The University of Newcastle Callaghan NSW 2308

PRC in Physical Activity & Nutrition -2011 Seed Funding Application

Dear Joseph

Thank you for submitting your application for Seed Funding in 2011 for consideration by the PRC in Physical Activity & Nutrition. The PRC Research Management committee have reviewed your application and have decided to support the proposal – Congratulations!

You have been awarded \$5,000 to be used towards your proposal titled - The measurement of physical activity in adolescents: testing objective monitoring protocols and validation of self-reporting measures.

Applications were assessed on the quality and approach/methodology of the research project, the potential for strategic growth of research within the PRC, including the potential of the project to attract external research funding, and the researcher/s track record/s relative to opportunity.

You are required to liaise with the Theme Leader Dr David Lubans to finalise your budget. The PRC does not accept responsibility for the provision of general equipment and facilities, including basic desk top computers. This is the responsibility of Chief Investigator & Theme Lead to organise within your School. You are also required to liaise with relevant School staff to arrange any staff appointments.

Staff on fixed-term contracts should clearly state their period of appointment, for example, 1 July 2011 to 31 December, 2011. Staff on a fractional appointment should indicate percentage, for example, 50%. Discuss any queries relating to the appropriate level of staff appointments with Human Resource Services staff http://www.newcastle.edu.au/service/recruitment/contacts.html. Ensure that salary on-costs are included in the budget as appropriate. Information on salary rates and on-costs can be found from the relevant links on the following website: http://www.newcastle.edu.au/service/grants-and-contracts/internal-grants/preparing-research-budgets.html

All staff appointments should quote the cost collector **10.21720**. No 'overspending' on staff salaries will be permitted. Any overspends will be the responsibility of the Chief Investigator. Copies of all staff appointments should be forwarded to the PRC Centre Coordinator, Airlie Johnson.

You have been allocated a unique code PRCSF11-22 to be used in all cost coding in Spend-vision. In the comments section please use the code PRCSF11-22 followed by a brief description of the expenditure. This will be used for reporting purposes. All purchase orders are to be processed through your School. Copies of purchase orders should be forwarded to the PRC Centre Coordinator, Airlie Johnson.

Restrictions on Expenditure

Funding Exceptions – funding may not be spent on:

- Equipment over \$1,000 each; and
- any item that was not originally included in the budget proposal of the application form.

PRIORITY RESEARCH CENTRE PHYSICAL ACTIVITY & NUTRITION



Where a researcher wishes to vary approved expenditure and/or include any of the items listed under Funding Exceptions above, permission should be sought in writing from the PRC Director. Reasonable requests for variation will be accommodated wherever possible, but permission should be sought in advance of expenditure.

Forfeiture of Grant Funding

Grants not spent or committed within the nominated funding period of the grant will not be carried forward to the following year. All monies that you have been awarded must be spent by <u>9 December 2011</u>. No monies can be carried over to 2012.

If you are aware for any reason that the monies cannot be spent please advise the PRC Director as soon as possible so that any unspent balances can be re-allocated.

Ethics Clearances: The University is bound by legislation to ensure research does not commence without appropriate ethics and safety approvals. Research projects conducted within the University or in the name of the University must comply with the AVCC Guidelines for Responsible Practice in Research and must have the relevant Human/Animal ethics and/or Safety approvals. Separate application must be made, if appropriate, for ethics approval of the Animal Care and Ethics Committee and/or the Human Research Ethics Committee. Approval must be obtained before the project commences. Projects which raise safety issues cannot proceed until they have been cleared by the Safety Committee.

Safety Approvals: There are statutory requirements for the control of risks arising from workplace hazards which both the University and the Chief Investigator are obliged to meet. Accurate identification of the hazards associated with the research project is vital to meeting this obligation.

Reporting Requirements for Successful Applicants: Final reports are required by 30 March 2012. The report is to be submitted on the PRC Grants Scheme 2011 Report Form. Failure to submit a final report will prevent eligibility for future PRC seed funding rounds. The Final report must include details of external grant applications, manuscripts and presentations. The PRC must be acknowledged in all publications and conferences that result from this funding.

If you have any question please contact the Centre Co-ordinator, Airlie Johnson via email: <u>Airlie Johnson@newcastle.edu.au</u> or Telephone: 02-4921 7792. Again, congratulations on your application and good luck with your research!

Best regards,

Ronald C. Plotnikoff, PhD Professor and Chair in Physical Activity and Population Health Education Director, Priority Research Centre in Physical Activity and Nutrition HPE3.08, HPE Building University of Newcastle Callaghan, NSW 2308 Australia Tel: 61 2 49854465 E-mail: ron.plotnikoff@newcastle.edu.au

Appendix 5: Measuring activity with through energy expenditure: The strengths and weaknesses of doubly labelled water and heart-rate monitoring

Doubly labelled water

The doubly labelled water (DLW) method is based on the kinetics of two stable isotopes of water, ${}^{2}\text{H}_{2}\text{O}$ (deuterium-labelled water) and $\text{H}_{2}{}^{18}\text{O}$ (oxygen-18-labeled water). Deuterium-labelled water is lost from the body through the usual routes of water loss (urine, sweat, evaporative losses). Oxygen-18-labeled water is lost from the body at a slightly faster rate because this isotope is also lost via carbon dioxide production in addition to all routes of water loss. The difference in the rate of loss between the two isotopes is a function of the rate of carbon dioxide production. This is therefore reflection of the rate of energy production over time [321]. By combining this with the measurement of resting energy expenditure, DLW can be used to estimate energy expenditure (EE) as a result of physical activity.

Strengths and weaknesses of Doubly labelled water

The DLW method has been regarded as the 'gold standard' and the most valid and reliable way to measure EE in free living conditions [52]. This is reflected in validation studies in adults and children that have shown a small measurement error of 5-10% [321, 322]. Due to the high level of reliability and validity, the DLW method has also been used to validate physical activity questionnaires in adolescents [323]. Further strengths of the DLW method is that it is unobtrusive and non-invasive to the participant [49].

A major limitation in the use of DLW is its very labour intensive for researchers and very expensive and burdensome for participants, making it only suitable for small studies with a large of amount of resources and funding. The use of the DLW method is therefore most seen in well-resourced research activities rather than health promotion and intervention studies [324, 325]. The greatest weakness of the DLW method is it does not provide frequency, intensity or type of activity [65, 269]. For these reasons, there are limited studies within the literature where DLW has been used to measure physical activity in free-living adolescents.

Appendices

Although the research shows the DLW method can accurately measure EE, the labour intensive nature of data collection and the costly expense does not make it feasible for use with large samples[65]. As the DLW is a biological method, and the participant is provided with a dose rather than a monitor (such as a pedometer or accelerometer), non-wear time and missing data is eliminated which is a common issue among free-living adolescents [114]. In summary, the DLW method provides an accurate, yet not particularly feasible method of assessing physical activity in free-living adolescents.

Heart rate monitoring

Heart rate monitoring has been used in various populations to provide an objective assessment of the physiological effect of physical activity [49]. The monitor is fixed to the participant via an elastic belt normally across the chest (more modern devices can be worn on the wrist or arm) under the clothing and provides minute-by-minute heart rate data that can be stored over multiple days [326]. It has been reported that heart monitors are more useful for classifying groups of individuals rather than estimating individuals' physical activity level [15]. Nonetheless, they have shown usefulness when estimating children's and adolescents' physical activity in small-to-moderate sample sizes [52].

Strengths and limitations of heart rate monitoring

Heart rate monitors have been used to provide an estimate of adolescents' physical activity levels as it now well-documented in the literature that heart rate has a strong linear association with energy expenditure [327, 328]. Heart rate monitors are reasonably inexpensive in comparison to pedometers and some accelerometers other objective measures, especially those that measure energy expenditure. They also have the ability to store data over consecutive days which is an added advantage for their use with adolescents [65]. Heart rate monitoring compares well to the DLW method, self-report physical activity questionnaires, pedometry and accelerometry [65, 163, 322].

Appendices

Heart rate monitoring does not measure physical activity, it is an indirect estimate of physical activity which is based on linear relationship between heart rate and oxygen uptake [49]. A further weakness of the measure is heart rate can be affected by temperature and emotional stress and other factors such as age, body size, proportion of muscle mass and cardiorespiratory fitness, therefore making physical activity assessment more complex [65]. In addition, heart rate tends to lag momentarily behind movement and also tends to remain high after the cessation of movement and therefore masks intermittent and sporadic movement patterns [269]. Heart rate monitoring also provides no contextual information of the physical activity that has been completed. Researchers have developed techniques such as heart rate indices that control for individual differences and individualised HR-VO₂ calibration curves [56, 269]. However, these in-depth approaches are time consuming and burdensome for researchers and hence are not feasible for studies with large sample sizes.

Due to the sporadic and intermittent movement patterns adolescents, heart rate monitoring alone may not capture the true activity pattern of the individuals in free-living conditions. However, as stable exercise has been shown to strongly correlate with heart rate and respiration, there continues to be clinical studies validating the use of heart rate monitoring [65]. For heart rate monitors to accurately measure an individual's physiological change in heart rate a result of physical activity, the monitor needs to be calibrated to the wearer. This involves collecting and entering an individual's information which may not be feasible for larger studies [50]. In summary, heart rate monitoring will remain a useful and unobtrusive tool for measuring freeliving adolescents, although it has been recommended that it be used in conjunction with other measures such as accelerometers which have the ability to capture duration and intensity of activity [49, 56].

Appendix 6: Parent information letter and consent forms

Research Project: The Measurement of Physical Activity in Adolescents: Testing Objective Monitoring Protocols and Validation of Self-Reporting Measures

PARENT INFORMATION SHEET

Dear Parents/Guardians,

Your child's school has been invited to participate in a study being conducted by A/Prof David Lubans, Prof Ron Plotnikoff, Prof Philip Morgan and Research Higher Degree student Joseph Scott from the University of Newcastle.

Why is this research being done?

The measurement of physical activity is an important and challenging venture. Accurate measurement of physical activity is necessary to estimate how active young people are and to evaluate programs to promote activity. With more and more physical activity studies now using objective measurement devices, there is an urgent need to identify optimal assessment protocols for various populations. By comparing existing physical measurement protocols and validating a series of physical activity questionnaires we are attempting to refine and improve physical activity measurement.

Who can participate in this research?

Two Personal Development, Health and Physical Education (PDHPE) classes in years 9 or 10 will be invited to participate in the study. Students with pre-existing health issues and/or special needs can participate in the study if it does not impede their ability to wear a functional pedometer.

What is involved in this study?

If your child agrees to participate, s/he will take part in a study which will be conducted at school by a member of the research team. The study will involve your child wearing a pedometer and an accelerometer (devises for measuring physical activity) for 7 days and completing a physical activity questionnaires three times and a pedometer questionnaire once. Students will also be asked to self-report their height and weight (no physical measures will be taken). These questionnaires will focus solely on your child's physical activity levels and behaviours that they exhibited whilst wearing a pedometer. A small number of students (four from each class) may also be selected to be involved in a focus group. As part of a focus group, your child will then be asked questions by a member of the research team regarding their perceptions and attitudes of the measuring process. Discussion during the focus group will be recorded using a digital recorder. During the recording students, will not be identified by name and are not required to answer a question if they do not wish to. Should your child like to withdraw any comments personally made during the recording, this will be noted by the researcher conducting the focus group and written transcripts of the recording will be amended. Focus group questioning will occur only once in your child's normal Physical Education lesson on school grounds in a room allocated by the school and will run for no more the 30 minutes.

How much time will it take?

On day 1, the researchers will explain the study, provide each student with a pedometer and an accelerometer and then students will complete two brief physical activity questionnaires. (20 minutes). Students will then wear a pedometer and accelerometer for the next 7 days. On Day 8; students will their return pedometers and accelerometers complete the same physical activity questionnaires and answer questions regarding their perceptions of wearing the devices and (25 minutes). The following week students will be asked to complete the same physical activity questionnaire for a third time (10mins) and a selection of students will be invited to participate in focus groups to explore students' perceptions and attitudes of the measuring process.

What choice do participants have?

The school principal has agreed to your child's school being involved in the study. However, participation in the study is entirely you and your child's choice. If consent to participate is provided, your child can still choose to withdraw from the study at any time and will be free to discontinue participation in the assessments. A decision not to participate or discontinuation of involvement in the study will not jeopardise you or your child's relationships with the University of Newcastle or the school. Withdrawal from this task will not result in any disciplinary action, nor will it affect your child's academic grades, given that this is a purely voluntary research task. Students' participation is entirely voluntary. Students who choose not to participate in the study will be with their normal teacher and will continue lessons as per usual.

What are the benefits and risks of participating?

Taking part in this research project will allow students to benefit from an increased awareness of the importance of physical activity and leading a healthy lifestyle. Students will gain experience in physical activity measurement and an understanding of how research is conducted. The topic of 'physical activity measurement' fits into the Personal Development, Health and Physical Education (PDHPE) syllabus. Teachers can easily link the content of this study to the students' curriculum and hence increase their knowledge in certain areas of the syllabus. Mr Joseph Scott will provide a presentation to the schools on the results along with an overview of physical activity assessment methods. There are no foreseeable risks and students can choose to withdraw from the study at any time.

How will the information collected be used?

Steps counts, questionnaire responses and audio recorded data obtained during the focus groups will be used to further refine the physical activity measuring protocols and validate existing physical activity questionnaires. This information may be used for journal publications and conference presentations and will contribute to Mr Joseph Scott research higher degree.

How will privacy be protected?

Any personal information provided by you or your child will be confidential to the researchers. Any data collected from the focus group will be published in general terms and will not allow the identification of individuals or schools. Audio recorded data will be transcribed by a member of the research team and stored. All data will be securely retained for 5 years. No participant will be identifiable in the data files.

What do you need to do to participate?

You and your child will need to complete both accompanying *Student* <u>and</u> *Parent Consent Forms*, and have them returned to the school's office or your child's roll-class teacher as soon as possible.

Further information

Following the completion of the study, Joseph Scott will return to the school and present the study's findings to the principal, teachers and staff. A summary of the findings will be

given to the principal and involved staff. It is suggested that the findings will be disseminated to students and their parents via a school newsletter or similar method. If you would like further information you can contact A/Prof David Lubans using the details below. Thank you for considering this invitation.

A/Prof David Lubans F	Prof Ron Plotnikoff	Prof Philip Morgan	Joseph Scott
A/Prof David	Prof Ron Plotnikoff	Prof Philip Morgan	Joseph Scott
Lubans	Faculty of Education	Faculty of Education	Faculty of Education &
Faculty of Education	& Arts	& Arts	Arts
& Arts	School of Education	School of Education	School of Education
School of Education	Phone: (02) 4985	Phone: (02) 4921	Phone: 0431595272
Phone: (02) 4921	4465	7265	Joseph.scott@newcastl
2049	Ron.Plotnikoff@new	Philip.Morgan@newc	e.edu.au
David.Lubans@newc	castle.edu.au	astle.edu.au	
astle.edu.au			

This project has been approved by the University's Human Ethics Committee, Approval No. H-2011-0137. Should you have concerns about your rights as a participant in this research, or you have a complaint about the manner in which the research is conducted, it may be given to the researcher, or, if an independent person is preferred, to the Human Research Ethics Officer, Research Office, The Chancellery, The University of Newcastle, University Drive, Callaghan NSW 2308, Australia (02) 4921 6333, or email <u>Human-Ethics@newcastle.edu.au</u>

Research Project: The Measurement of Physical Activity in Adolescents: Testing Objective Monitoring Protocols and Validation of Self-Reporting Measures

PARENT CONSENT FORM

Chief Investigator: A/Prof	David Lubans
Associate Investigator:	Prof Ron Plotnikoff
Associate Investigator:	Prof Philip Morgan
Associate Investigator:	Research Higher Degree student Joseph Scott

I have been given information about the research project identified above and have discussed it with my child. I understand that if I consent to my child participating in this project, he/she will participate in this study. I understand that my child's participation is entirely voluntary.

This will involve my child wearing a pedometer and an accelerometer (devises for measuring physical activity) for 7 days and completing the same physical activity questionnaire at three different times (the Oxford Physical Activity Questionnaire) and a Pedometry Behaviour Questionnaire once. Students will also be asked to self-report their height and weight (no physical measures will be taken). I understand that *if* my child is selected to be involved in a focus group (which is a small group discussion) they will only answer questions based on their perception/attitudes of the measuring process. I understand that an audio recording will be taken during the focus group and data obtained from this recording may subsequently be used in making changes to the questionnaires. I am also aware that my child will not be identified by name during the recording, this will be noted by the researcher conducting the focus group and written transcripts of the recording will be amended.

I have had an opportunity to ask A/Prof Lubans questions about the research and my child's participation. I understand that my child's participation is voluntary and I am free to withdraw him/her from the research at any time. My refusal to his/her participation or withdrawal of consent will not affect my relationship with the University of Newcastle or my child's school. Withdrawal from this task will not result in any disciplinary action against my child, nor will it affect his/her academic grades, given that this is a purely voluntary research task.

By signing below, I am indicating consent for my child to participate in this research project conducted by A/Prof David Lubans, Prof Ron Plotnikoff, Prof Philip Morgan and Research Higher Degree student Joseph Scott as it has been described to us in the Information Statement, a copy of which I have retained.

Student name:	
Parent/guardian name:	

Contact	nhone number: /	(H)	
Contact		(' ')	

Signature: _____ Date: _____

Please sign the completed consent form and return with your CHILD'S CONSENT FORM to their roll-call teacher or the school's office

Appendix 7: Student (participant) information and consent forms

Research Project: The Measurement of Physical Activity in Adolescents: Testing Objective Monitoring Protocols and Validation of Self-Reporting Measures

STUDENT INFORMATION SHEET

Dear Student,

Your school is invited to participate in a study being conducted by A/Prof David Lubans, Prof Ron Plotnikoff, Prof Philip Morgan and Research Higher Degree student Joseph Scott from the University of Newcastle.

Why is this research being done?

The measurement of physical activity is an important and challenging venture. Accurate measurement of physical activity is necessary to estimate how active young people are and to evaluate interventions to promote activity and identify the predictors of physical activity behaviour. By comparing existing measuring protocols and validating a series of physical activity questionnaires we are attempting to refine physical activity measurement.

Who can participate in this research?

Two Personal Development, Health and Physical Education (PDHPE) classes in years 9 or 10 will be invited to participate in the study. Students with pre-existing health issues and/or special needs can participate in the study if it does not impede their ability to wear a functional pedometer.

What is involved in this study?

If you agree to participate, you will take part in a study which will be conducted at school by a member of the research team. The study will involve you wearing a pedometer and an accelerometer (devises for measuring physical activity) for 7 days and completing a physical activity questionnaires three times and a pedometer questionnaire once. Students will also be asked to self-report their height and weight (no physical measures will be taken). These questionnaires will focus solely on your physical activity levels and behaviours that you exhibited whilst wearing a pedometer. A small number of students may also be selected to be involved in a focus group. As part of a focus group, you will then be asked questions by a member of the research team regarding your perceptions and attitudes of the measuring process. Discussion during the focus group will be recorded using a digital recorder. During the recording you will not be identified by name and are not required to answer a question if you do not wish to. Should you like to withdraw any comments personally made during the recording, this will be noted by the researcher conducting the focus group and written transcripts of the recording will be amended. Focus group questioning will occur only once in your normal Physical Education lesson on school grounds in a room allocated by the school and will run for no more the 30 minutes.

How much time will it take?

On day 1, the researchers will explain the study, provide each student with a pedometer and an accelerometer and then students will complete two brief physical activity questionnaires. (20 minutes). Students will then wear a pedometer and accelerometer for the next 7 days. On Day 8; students will their return pedometers and accelerometers complete the same physical activity questionnaires and answer questions regarding their perceptions of wearing the devices and (25 minutes). The following week students will be asked to complete the same physical activity questionnaires for a third time (10mins)
and a selection of students will be invited to participate in focus groups to explore students' perceptions and attitudes of the measuring process.

What choice do participants have?

The school principal has agreed to your school being involved in the study. However, participation in the study is entirely your choice. If consent to participate is provided, you can still choose to withdraw from the study at any time and will be free to discontinue participation in the assessments. A decision not to participate or discontinuation of involvement in the study will not jeopardise your relationships with the University of Newcastle or the school. Withdrawal from this task will not result in any disciplinary action, nor will it affect your academic grades, given that this is a purely voluntary research task.

What are the benefits and risks of participating?

Taking part in this research project will allow you to benefit from an increased awareness of the importance of physical activity and leading a healthy lifestyle. You will gain experience in physical activity measurement and an understanding of how research is conducted. There are no foreseeable risks and you can choose to withdraw from the study at any time.

How will the information collected be used?

Steps counts, questionnaire responses and audio recorded data obtained during the focus groups will be used to further refine the physical activity measuring protocols and validate existing physical activity questionnaires. This information may be used for journal publications and conference presentations and will contribute to Mr Joseph Scott research higher degree.

How will privacy be protected?

Any personal information provided by you will be confidential to the researchers. Any data collected from the focus group will be published in general terms and will not allow the identification of individuals or schools. Audio recorded data will be transcribed by a member of the research team and stored. All data will be securely retained for 5 years. No participant will be identifiable in the data files.

What do you need to do to participate?

You and your parent(s) will need to complete both accompanying *Student* <u>and</u> *Parent Consent Forms*, and return them to the School's office or your roll-class teacher as soon as possible. Your participation is entirely voluntary.

Further information

Following the completion of the study, Joseph Scott will return to the school and present the study's findings to the Principal, Teachers and staff. A summary of the findings will be given to the principal and involved staff. It is suggested that the findings will be disseminated to students and their parents via a school newsletter or similar method. If you would like further information you can contact A/Prof David Lubans using the details below. Thank you for considering this invitation.

A/Prof David Lubans F	Prof Ron Plotnikoff	Prof Philip Morgan	Joseph Scott		
A/Prof David Lubans Faculty of Education & Arts School of Education Phone: (02) 4921 2049 David.Lubans@newc astle.edu.au	Prof Ron Plotnikoff Faculty of Education & Arts School of Education Phone: (02) 4985 4465 Ron.Plotnikoff@new castle.edu.au	Prof Philip Morgan Faculty of Education & Arts School of Education Phone: (02) 4921 7265 Philip.Morgan@newc astle.edu.au	Joseph Scott Faculty of Education & Arts School of Education Phone: 0431595272 Joseph.scott@newcastl e.edu.au		

This project has been approved by the University's Human Ethics Committee, Approval No. H-2011-0137. Should you have concerns about your rights as a participant in this research, or you have a complaint about the manner in which the research is conducted, it may be given to the researcher, or, if an independent person is preferred, to the Human Research Ethics Officer, Research Office, The Chancellery, The University of Newcastle, University Drive, Callaghan NSW 2308, Australia (02) 4921 6333, or email <u>Human-Ethics@newcastle.edu.au</u>

Research Project: The Measurement of Physical Activity in Adolescents: Testing Objective Monitoring Protocols and Validation of Self-Reporting Measures

STUDENT CONSENT FORM

Chief Investigator: A/Prof	David Lubans
Associate Investigator:	Prof Ron Plotnikoff
Associate Investigator:	Prof Philip Morgan
Associate Investigator:	Research Higher Degree student Joseph Scott

I have been given information about the research project identified above and have discussed it with my parent(s). I understand that if I consent to participating in this project, I will participate in the study that is called '*The measurement of physical activity in adolescents: testing objective monitoring protocols and validation of self-reporting measures*' I understand that my participation is entirely voluntary.

This will involve me wearing a pedometer and an accelerometer for 7 days and completing the same physical activity questionnaire at three different times (the Oxford Physical Activity Questionnaire) and a Pedometry Behaviour Questionnaire once. Students will also be asked to self-report their height and weight (no physical measures will be taken). I understand that *if* I am selected to be involved in a 'focus group' (which is a small group discussion) I will only answer questions based on my perceptions/attitudes of the measuring process. I understand that my answers will be recorded and maybe used in making changes to the questionnaires. I am also aware that I will not be identified by name during the recording and if I would like to withdraw any comments made during the recording this will be noted by the researcher conducting the focus group and written transcripts of the recording will be amended.

I have had an opportunity to ask A/Prof Lubans questions about the research and my participation. I understand that my participation is voluntary and I am free to withdraw from the research at any time. My refusal to participation or withdrawal of consent will not affect my relationship with the University of Newcastle or my school. Withdrawal from this task will not result in any disciplinary action against me, nor will it affect my academic grades, given that this is a purely voluntary research task.

By signing below I am indicating my consent to participate in this research project conducted by A/Prof David Lubans, Prof Ron Plotnikoff, Prof Philip Morgan and Research Higher Degree student Joseph Scott as it has been described to us in the Information Statement, a copy of which I have retained.

Student name:	
Parent/guardian name	e:
Student contact num	ber: (m)
Email:	(Only required if you don't have a mobile)
Signature:	Date:

Please sign the completed consent form and return with your CHILD'S CONSENT FORM to their roll-call teacher or the school's office

Appendix 8: Accelerometer log sheet: This sheet was used by participants to log non-wear time of measuring devices (i.e., swimming, contact sports etc.)





Appendices

	Day/Date	5.	6.	7.	8.
	12-1				
	1-2				
	2-3				
	3-4				
	4-5				
AM	5-6				
	6-7				
	7-8				
	8-9				
	9-10				
	10-11				
	11-12				
	12-1				
	1-2				
	2-3				
	3-4				
	4-5				
РМ	5-6				
	6-7				
	7-8				
	8-9				
	9-10				
	10-11				
	11-12				
	Total time swimming for the day:				
т	Total time riding a bike for the day:				
PED	#	##REMEN METERS H/	IBER### Ave to be f	RETURNED O	N DAY 8!!!

Appendix 9: Pedometer log books: This was used by the participants to record their daily step counts.



PEDOMETER LOG SHEET School: Name: INSTRUCTIONS: - When you take off your pedometer at night to go to sleep write down your step count - Also, write in the day and date next to your step count. EXAMPLE: DAY DAY OF WEEK DATE STEP COUNT 1 15.9.11 6,540 Tuesday 2 Wednesday 16.9.11 8,971 12,564 3 Thursday 17.9.11 4 Friday 18.9.11 6,826 5 19.9.11 14,879 Saturday Sunday 20.9.11 5,768 6 7 Monday 21.9.11 9,932 8 Tuesday 22.9.11 8,734

YOUR DAILY PEDOMETER LOG:

DAY	DAY OF WEEK	DATE	STEP COUNT
1			
2			
3			
4			
5			
6			
7			
8			

REMEMBER: PEDOMETERS AND ACCELEROMETERS HAVE TO BE RETURNED ON DAY 8!!!

Appendix 10: Oxford Physical Activity Questionnaire (OPAQ). Previous validated questionnaire in the adolescent population (Lubans, 2008)





Student Name: _____

School: _____

To protect your privacy this cover sheet will be removed and destroyed once

You've been allocated a study number.

A/Prof David	Prof Ron Plotnikoff	Prof Philip Morgan	Joseph Scott			
Lubans	Faculty of Education	Faculty of Education	Faculty of Education &			
Faculty of Education	& Arts	& Arts	Arts			
& Arts	School of Education	School of Education	School of Education			
School of Education	Phone: (02) 4985	Phone: (02) 4921	Phone: 0431595272			
Phone: (02) 4921	4465	7265	Joseph.scott@newcastl			
2049	Ron.Plotnikoff@new	Philip.Morgan@newc	e.edu.au			
David.Lubans@newc	castle.edu.au	astle.edu.au				
astle.edu.au						

Appendices

Oxford Physical Activity Questionnaire (OPAQ)
Age:
Gender: please tick (<) 🛛 🗠 Male 🔅 Female
Height:cm Weight:kg
In what country were you born? <i>Please tick</i> (✓) □ Australia □ Another country (Please specify):
What is your cultural background?
□ Australian □ Aboriginal □ Torres Strait Islander □ Asian □ Europear □ Middle Eastern □ African □ Other: (please specify)
What language do you speak most at home? Please tick (🗸)

The purpose of this questionnaire is to estimate the amount of time you spent participating in physical activity over the past 7 days. This includes physical education, school sport and other moderate to vigorous physical activity you completed during the week and on the weekend.

Moderate to vigorous physical activity makes you breathe heavily and increases your heart rate. It includes all sports, exercise activities, games, swimming, running, skateboarding etc.

Please indicate the **ACTIVITY** and the **TIME** you spent doing the activity. There are a number of questions that will help you to remember, along with a list of common activities at the top of the next page. If you did not do any physical activity, please leave the table blank.

1) How do you usually travel to school? (Please circle)

Walk Cycle Bus Car Other

2) How long does each journey take? (Write) _____ mins.

3) Over the past 7 days, on how many days did you walk or ride, scoot or skate to school? (*Please circle*).

0 1 2 3 4 5 6

Now, please record the following information on the timetable on the NEXT PAGE:

- 4) List all the practical activities that you did in PE lessons over the past 7 days (e.g. football, hockey, dance, and gymnastics).
- 5) List all the activities that you did in school sport over the past 7 days (e.g. basketball, weight training, and golf).
- 6) List any sports or activities that you did before/after school or on weekends over the past 7 days (e.g. surfing, cricket and dance).
- 7) List any other physical activities that you did over the past 7 days that lasted 20 minutes or longer (e.g. you might have played soccer at recess or lunchtime).

Oxford Physical Activity Questionnaire (EXAMPLE)

Aerobics Athletics Austag Baseball Basketball	Cricket Cycling Dance (ba Dance (otl Dance (jaz	illet) ner) zz)	Golf Gymnastics Hockey Indoor socc Inline hocke	er ey	Inline skati Lifesaving Martial arts Netball Rowing	ng	Running (jog Rugby leagu Rugby unior Skateboardi Soccer	gging) Je N ng	Surfing Swimming Tennis Touch footb Volleyball	all													
				Partici	pation in pl	hysical a	activity during	g the la	st 7 days														
Component	Mond	ay	Tuesda	ay	Wednes	sday	Thursd	ay	Frida	у	Saturo	lay	Sunda	ıy									
	Activity	Mins	Activity	Mins	Activity	Mins	Activity	Mins	Activity	Mins	Activity	Mins	Activity	Mins									
Before school																							
Morning																							
(including breaks)							Dance(PE)	40															
Lunch time			Basketball	45																	Tauah	Touch	
Afternoon			2 40110 12 411								Surfing) 120	football	60									
(including breaks)																							
After school									Soccer	90													
Evening																							
	No	ote: Ren	nember to inc	clude th	e number c	of minute	es, examples	have b	een have bee	en provie	ded for you.												

Oxford Physical Activity Questionnaire

Aerobics Athletics Austag Baseball Basketball	Cricket Cycling Dance (ba Dance (otł Dance (jaz	llet) ner) zz)	Golf Gymnastics Hockey Indoor socc Inline hocke	er ey Partici	Inline skati Lifesaving Martial arts Netball Rowing pation in ph	ng s nysical a	Running (jo Rugby leag Rugby unio Skateboard Soccer ctivity durin	gging) ue n ling g the la	Surfing Swimming Tennis Touch footb Volleyball st 7 days	all				
Commonweat	Mond	ay	Tuesd	ay	Wednes	sday	Thursd	ay	Frida	y	Saturd	lay	Sunda	y
Component	Activity	Mins	Activity	Mins	Activity	Mins	Activity	Mins	Activity	Mins	Activity	Mins	Activity	Mins
Before school														
Morning														
(including breaks)														
Lunch time														
Afternoon														
(including breaks)														
After school														
Evening														
			Not	e: Reme	ember to inc	lude the	e activity and	d numb	er of minutes	i.				

I

Appendix 11: Single-item physical activity questionnaire: A single-item questionnaire for assessing physical activity.



THE SINGLE ITEM QUESTIONNAIRE

Student Name: _____

School: _____

To protect your privacy this cover sheet will be removed and destroyed once

You've been allocated a study number.

A/Prof David 🗔	0006	Prof Ron Blatolkeff	Prof Philip Morgan	Joseph Soott
Faculty of Educa	tion & Arts	Faculty of Education & Arts	Faculty of Education & Arts	Faculty of Education & Arts
School of Educa	tion	School of Education	School of Education	School of Education
Phone: (02) 492	1 2049	Phone: (02) 4985 4465	Phone: (02) 4921 7265	Phone: 0431595272
David Lubans@	reveastie.edu.au	Ron.Plotnikoff@newcastle.edu.au	Philip Morgan@newcastle.edu.au	Joseph.scott@newcastle.edu.au

INSTUCTIONS:

To complete this questionnaire, answer the following question, by ticking the appropriate box.

QUESTION:

In the past week, on how many days have you done a total of 60 minutes or more of physical activity, which was enough to raise your breathing rate? This may include sport, exercise and brisk walking or cycling for recreation or to get to and from places? *Please tick only one box.*

No. of days	0	1	2	3	4	5	6	7
Tick one								

Appendix 12: Pedometry Behaviour Questionnaire (PBQ)



PEDOMETRY BEHAVIOUR QUESTIONNAIRE (PBQ)

STUDENT NAME:

SCHOOL:

To protect your privacy this cover sheet will be removed and destroyed once you have been allocated a study number.

A/Prof David Lubans	Prof Ron Plotnikoff	Prof Philip Morgan	Mr Joseph Scott
Faculty of Education &	Faculty of Education &	Faculty of Education &	Faculty of Education & Arts
Arts	Arts	Arts	School of Education
School of Education	School of Education	School of Education	Phone: 0431595272
Phone: (02) 4921 2049	Phone: (02) 4985 4465	Phone: (02) 4921 7265	Joseph.scott@newcastle.edu
David.Lubans@newcastle	Ron.Plotnikoff@newcastl	Philip.Morgan@newcastle	.au
.edu.au	e.edu.au	.edu.au	

INSTRUCTIONS:

- Please make sure your name is on the front page of the questionnaire. Your name will not be recorded and no-one except the researchers will see your answers.
- · This is not a test. There are no right or wrong answers.
- When you complete the questionnaire please read back over your answers to ensure you have not missed any questions.
- · There are three sections, please make sure you complete all three.

SECTION 1: Pedometer Behaviour

The following statements focus on how you wore a pedometer.

INSTRUCTIONS: Circle ONE option for each statement						
1.	When I forgot to write down my step count, I estimated my step count.	Never	Rarely	Sometimes	Often	Everyday
2.	Other than swimming, showering and sleeping, I wore my pedometer throughout the day.	Never	Rarely	Sometimes	Often	Everyday
3.	When wearing my pedometer I was more active than normal.	Never	Rarely	Sometimes	Often	Everyday
4.	I shook my pedometer to increase my step count.	Never	Rarely	Sometimes	Often	Everyday
5.	I purposely wrote down more steps than I actually did.	Never	Rarely	Sometimes	Often	Everyday

SECTION 2: Attitudes toward wearing pedometers

The following statements focus on how wearing a pedometer made you feel.

INCEDUCEIONS, Oracle ONE action for each statement						
INSTRUCTIONS: Circle ONE option for each statement						
1.	I wanted to get a high step count to impress the researchers.	Strongly Disagree	Disagree	Neutral	Agree	Strongly Agree
2.	I wanted to get a high step count to impress my friends and peers.	Strongly Disagree	Disagree	Neutral	Agree	Strongly Agree
3.	I found that wearing the pedometer was uncomfortable.	Strongly Disagree	Disagree	Neutral	Agree	Strongly Agree
4.	I found that wearing the pedometer was embarrassing.	Strongly Disagree	Disagree	Neutral	Agree	Strongly Agree
5.	I found that wearing the pedometer was a waste of time.	Strongly Disagree	Disagree	Neutral	Agree	Strongly Agree



SECTION 3: Wearing Pedometers				
INSTRUCTIONS: Please read the following questions and tick the appropriate box (You can tick more than one box for each answer).				
Questio	n	Answer		
1.	If you forgot to wear your pedometer, what did you do?	 Didn't wear it that day Put it on later in the day Estimated my step count for the day Other: 		
2a.	Did you do more physical activity than normal when you were wearing your pedometer?	 [] No [] A little more (less than 30min/day more) [] a fair bit more (30mins more) [] A lot more (greater than 30mins/day more) 		
2b.	In what ways did you increase your step count?	 [] Did more exercise [] Shook my pedometer [] Gave my pedometer to someone else [] Wrote down more steps than I actually did [] I didn't [] Other 		
3.	Who else wore your pedometer?	[] No-one [] Friend [] Boyfriend/girlfriend [] Brother or sister [] Parent [] A Pet		
4.	Did you like wearing a pedometer?	 Yes (If you answered 'yes' you are finished the questionnaire) No (If you answered 'no' go onto question 4b) 		
4b.	If you did you not like wearing a pedometer, what were your reasons?	[] I didn't want people to know how much activity I do [] It was uncomfortable [] It was embarrassing [] It was a hassle [] Other		

THE END THANK YOU FOR PARTICIPATING IN THIS QUESTIONNAIRE

Appendix 13: Accelerometry Behaviour Questionnaire (ABQ)



SCHOOL:

To protect your privacy this cover sheet will be removed and destroyed once you have been allocated a study number.

A/Prof David Lubans	Prof Ron Plotnikoff	Prof Philip Morgan	Mr. Joseph Scott
Finite of Films	From the of Education	Front milip worgan	Frank (Films
Faculty of Education &	Faculty of Education &	Faculty of Education &	Faculty of Education & Arts
Arts	Arts	Arts	School of Education
School of Education	School of Education	School of Education	Phone: 0431595272
Phone: (02) 4921 2049	Phone: (02) 4985 4465	Phone: (02) 4921 7265	Joseph.scott@newcastle.edu
David.Lubans@newcastle	Ron.Plotnikoff@newcastl	Philip.Morgan@newcastle	.au
.edu.au	e.edu.au	.edu.au	

INSTRUCTIONS:

- Please make sure your name is on the front page of the questionnaire. Your name will not be recorded and no-one except the researchers will see your answers.
- This is not a test. There are no right or wrong answers.
- When you complete the questionnaire please read back over your answers to ensure you have not missed any questions.
- · There are three sections, please make sure you complete all three.

SECTION 1: Accelerometer Behaviour

The following statements focus on how you wore the accelerometers.

INSTRUCTIONS: Circle ONE option for each statement						
1.	Other than swimming, showering and sleeping, I wore my accelerometers throughout the day.	Never	Rarely	Sometimes	Often	Everyday
2.	I wore wrist accelerometer more than my hip accelerometer.	Never	Rarely	Sometimes	Often	Everyday
3.	I wore hip accelerometer more than my wrist accelerometer.	Never	Rarely	Sometimes	Often	Everyday
4.	When wearing the accelerometers, I was more active than normal.	Never	Rarely	Sometimes	Often	Everyday
5.	I shook my accelerometers to increase my physical activity level	Never	Rarely	Sometimes	Often	Everyday

SECTION 2: Attitudes toward wearing Accelerometers

The following statements focus on how wearing the accelerometers made you feel.

INSTRUCTIONS: Circle ONE option for each statement						
The next 3 questions are in relation to the WRIST accelerometer.						
1.	I liked wearing the wrist accelerometer.	Strongly Disagree	Disagree	Neutral	Agree	Strongly Agree
2.	I found that wearing the accelerometer on my wrist was uncomfortable.	Strongly Disagree	Disagree	Neutral	Agree	Strongly Agree
3.	I found that wearing the wrist accelerometer was embarrassing.	Strongly Disagree	Disagree	Neutral	Agree	Strongly Agree
The next 3 questions are in relation to the HIP accelerometer.						
4.	I liked wearing the hip accelerometer.	Strongly Disagree	Disagree	Neutral	Agree	Strongly Agree
5.	I found that wearing the accelerometer on my hip was uncomfortable.	Strongly Disagree	Disagree	Neutral	Agree	Strongly Agree
6.	I found that wearing the hip accelerometer was embarrassing.	Strongly Disagree	Disagree	Neutral	Agree	Strongly Agree

SECTION 3: Wearing Accelerometers				
<u>INSTRUCTIONS:</u> Please read the following questions and tick the appropriate box (You can tick more than one box for each answer).				
Question	n	Answer		
1.	If you forgot to wear the accelerometers, what did you do?	[] Didn't wear it that day [] Put it on later in the day [] Other:		
2a.	Did you do more physical activity than normal when you were wearing the accelerometers?	 [] No [] A little more (less than 30min/day more) [] a fair bit more (30mins more) [] A lot more (greater than 30mins/day more) 		
2b.	In what ways did you increase your physical activity level?	[] Did more exercise [] Shook my accelerometer [] Gave my accelerometer to someone else [] I didn't [] Other		
3.	Who else wore your accelerometers?	[] No-one [] Friend [] Boyfriend/girlfriend [] Brother or sister [] Parent [] A Pet		
4.	Did you like wearing accelerometers?	 Yes (If you answered 'yes' you are finished the questionnaire) No (If you answered 'no' go onto question 4b) 		
4b.	If you did you not like wearing the accelerometers, what were your reasons?	[] I didn't want people to know how much activity I do [] It was uncomfortable [] It was embarrassing [] It was a hassle [] Other		

THE END THANK YOU FOR PARTICIPATING IN THIS QUESTIONNAIRE

Appendix 14: Focus-group questions

Focus group questions

Focus groups participants will be stratified into groups depending on their 7-day physical activity levels as measured by pedometers and accelerometers These focus groups will be numbered from 1-6 (Each group will not be named so that students so that do not know why they have been selected and will assume that it is random selection. Interviews will be recorded and later transcribed. Participants will be categorised as follows:

- 1. *High-active boys' group*: Participants within this focus group will be boys who have achieved an average of 60 minutes/day in MVPA over their monitoring period.
- 2. *High-active girls' group:* Participants within this focus group will be girls who have achieved an average of 60 minutes/day in MVPA over their monitoring period.
- 3. *Middle-active boys' group:* Participants within this focus group will be boys who have achieved an average of 30-60 minutes/day in MVPA over their monitoring period.
- 4. Middle active girls group: Participants within this focus group will be girls who have achieved an average of 30-60 minutes/day in MVPA over their monitoring period.
- 5. Low-active boys group: Participants within this focus group will be boys who have achieved an average of less than 30 minutes/day in MVPA over their monitoring period.
- 6. Low-active girls' group: Participants within this focus group will be girls who have achieved an average of less than 30 minutes/day in MVPA over their monitoring period.

ATTITUDINAL

- 1. Tell me about wearing a pedometer?
- 2. Did you feel that you had to get a high step count to impress the researchers, teachers or your friends? Why or why not?
- 3. What were some of the reasons for attempting to increase your step counts?
- 4. Did you think that most people are honest or dishonest about their step count?

5. Do you think that trying to find physical activity levels with pedometers is a waste of time? Why or Why not?

BEHAVIOURAL

- 1. Can you give some reasons that some of the other students may have removed the pedometers over the 7 days of measurement?
- 2. Did you do more exercise than normal when wearing a pedometer? Why or why not?
- 3. Did you attempt to increase your step count in anyway? If so, how?
- 4. What were the other students doing? Do you think they changed their behaviours?
- 5. Did you let anyone else wear your pedometer? Why/why not